Minister for Planning and Public Spaces

Dahua Group Sydney Project 3 Pty Ltd

Planning Agreement

Section 7.4 of the Environmental Planning and Assessment Act, 1979 (NSW)

BROST TAKILLOOP



3442-3062-0428v9

Contents

1	Definitions	1
2	Operation and Application of this Deed	7
	2.1 Operation	7
	2.2 Planning agreement under the Act	7
	2.3 Application	7
3	Application of Sections 7.11, 7.12 and 7.24 of the Act	7
4	Development Contributions	7
	4.1 Developer to provide Development Contributions	7
	4.2 Acknowledgment	7
	4.3 Conditions Precedent under the TIC Deed	7
	4.4 No satisfactory arrangements for over 925 Dwellings	8
5	Interest for late provision of Monetary Contribution	8
6	Effect of future SIC Determination	8
7	Effect of termination of TIC Deed	9
8	Publicity	9
9	Enforcement	10
	9.1 Developer to provide security	10
	9.2 Restrictions on subdivision	10
	9.3 Restriction on Construction Certificates	11
10	Land ownership and registration of Deed	11
	10.1 Land ownership	11
	10.2 Registration of Deed	11
	10.3 Evidence of registration	12
	10.4 Right to lodge caveat	12
	10.5 Release and discharge of Deed	12
11	Dispute Resolution	14
	11.1 Not commence	14
	11.2 Written notice of dispute	14
	11.3 Attempt to resolve	14
	11.4 Mediation	15
	11.5 Court proceedings	15
	11.6 Use of information	15
	11.7 No prejudice11.8 Continued performance of obligations	15
		15
12	GST	15
	12.1 Definitions	15
	12.2 Intention of the parties	16
	12.3 Reimbursement	16
	12.4 Consideration GST exclusive	16

	12.5 Additional Amounts for GST	16
	12.6 Non monetary consideration	16
	12.7 Assumptions	16
40	12.8 No merger	16
13	Assignment and Dealings	17
	13.1 Assignment or novation by Developer	17
	13.2 Assignment by Minister	17
	13.3 Dealings with the Land	17
14	Capacity	18
	14.1 General warranties	18
	14.2 Power of attorney	18
15	Reporting Requirement	18
16	General Provisions	19
	16.1 Liability	19
	16.2 Entire agreement	19
	16.3 Inconsistency	20
	16.4 Variation	20
	16.5 Waiver	20
	16.6 Further assurances	20
	16.7 Time for doing acts	20
	16.8 Governing law and jurisdiction	20
	16.9 Severance	20
	16.10 Preservation of existing rights	21
	16.11 No merger	21
	16.12 Counterparts	21
	16.13 Relationship of parties	21
	16.14 Good faith	21
	16.15 No fetter	21
	16.16 Explanatory Note	21
	16.17 Costs, expenses and stamp duty 16.18 Notices	21
	16.19 Interpretation	22
Sche	edule 1 - Section 7.4 Requirements	22
	edule 2 – Address for Service	24
		27
	edule 3 – Land	28
	edule 4 - Development Contribution	29
	edule 5 - Security Terms	30
		33
	exure A – TIC Deed	34
	exure B – Novation Deed	35
MIIIE	exure C – Greater Macarthur Growth Area	36

Date

3 December 2019 Krett While

Parties

Minister for Planning and Public Spaces ABN 20 770 707 468 of Level 15, 52 Martin Place, Sydney, New South Wales, 2000 (Minister)

Dahua Group Sydney Project 3 Pty Ltd ACN 606 391 922 of Suite 02, Level 20, 201 Elizabeth Street, Sydney NSW 2000 (Developer)

Background

- А The Developer owns the Land.
- В In December 2016, Campbelltown City Council endorsed a planning proposal to amend the LEP to rezone land in Menangle Park, NSW, including the Land, to create the Menangle Park Urban Release Area.
- С The Developer proposes to carry out the Development on the Land.
- D Clause 6.1 of the LEP provides that the Consent Authority must not grant Development Consent for certain land subdivision in an Urban Release Area unless the Secretary has issued a Satisfactory Arrangements Certificate regarding the provision of Designated State Public Infrastructure.
- Е The Developer has offered to enter into this Deed with the Minister to provide the Development Contributions to enable a Satisfactory Arrangements Certificate to be issued for the Development under clause 6.1 of the LEP, subject to compliance with this Deed. For the avoidance of doubt, this Deed only constitutes satisfactory arrangements for a maximum of 925 Dwellings on the Land.

Operative provisions

1 Definitions

In this Deed these terms have the following meanings:

Act

the Environmental Planning and Assessment Act 1979 (NSW).

Address for Service	the address of each party appearing in Schedule 2 of this Deed or any new address notified in writing by any party to all other parties as its new address for service.		
Authority	any Federal, State or local government or semi- governmental, statutory, judicial or public person, instrumentality or department.		
Bank Guarantee	an irrevocable and unconditional undertaking without an expiry or end date in favour of the Minister to pay an amount or amounts of money to the Minister, on demand issued by:		
	(a)	one	of the following banks:
		(i)	Australia and New Zealand Banking Group Limited;
		(ii)	Commonwealth Bank of Australia;
		(iii)	Macquarie Bank Limited;
		(iv)	National Australia Bank Limited;
		(v)	St George Bank Limited;
		(vi)	Westpac Banking Corporation; or
	(b)		other financial institution approved by the ster in his absolute discretion.
Business Day	a day on which banks are open for business generally in Sydney, and which is not a Saturday, Sunday or bank or public holiday in Sydney and specifically excluding 27, 28, 29, 30 and 31 December.		
Carriageway	has the meaning given to that expression in the TIC Deed.		
Completion	has the meaning given to that expression in the TIC Deed.		
Complying Development Certificate	has the meaning given to that expression in the Act.		
Conditions Precedent	has the meaning given to that term in the TIC Deed.		
Consent Authority	has the meaning given to that expression in the Act.		
Construction Certificate	has the meaning given to that expression in the Act.		
Dahua 2	means Dahua Group Sydney Project 2 Pty Ltd ACN 606 391 235 of Suite 02, Level 20, 201 Elizabeth Street, Sydney NSW 2000.		

Dealing	in relation to the Land, means sell, transfer, assign, or dispose of the Land in whole or part, but excludes the entry by the Developer into a contract for the sale of a Final Lot (including a contract for an "off the plan" sale).
Dedicated Land	means the land to be transferred to RMS under the TIC Deed for the purposes of the Spring Farm Parkway.
Deed	means this document.
Designated State Public Infrastructure	has the meaning given to that expression in the LEP.
Development	the subdivision and other development of the Land for residential, commercial, industrial and related purposes and includes the subdivision of the Land into Final Lots to enable the erection of a maximum of 925 Dwellings.
Development Application	has the same meaning given to that expression in the Act.
Development Consent	has the same meaning given to that expression in the Act.
Development Contribution	the contributions to be provided by the Developer in accordance with clause 4 and Schedule 4 .
Dwelling	has the meaning given to that expression in the <i>Standard Instrument (Local Environmental Plans) Order 2006.</i>
Explanatory Note	means the note exhibited with a copy of this Deed when this Deed is made available for inspection by the public pursuant to the Act, as required by the Regulation.
Education Land	has the meaning given to that expression in the Landcom and Dahua 2 VPA.

Final Lot	sepa	ns any lot created in the Development for trate residential, retail, industrial or commercial pation and disposition and which is not:			
	(a)	intended to be further subdivided (including to create a strata or community lot); or			
	(b) a Service Lot or a lot of a kind or created for a purpose that is otherwise agreed by the Parties,				
	and Lanc	not being a lot created by the subdivision of the I:			
	(c)	that is to be dedicated or otherwise transferred to the Minister or RMS; or			
	(d)	on which is situated a dwelling-house that was in existence at the date of this Deed.			
Funding Amount	has Dee	the meaning given to that expression in the TIC d.			
Funding Amount Allocation	has Dee	the meaning given to that expression in the TIC d.			
Greater Macarthur Growth Area	means the area identified as the "Greater Macarthur Growth Area" in the <i>Proposed Special Infrastructure</i> <i>Contribution - Greater Macarthur</i> dated November 2018, as depicted in the plan at Annexure C , or if a SIC Determination is made in respect of the Land, then the area identified in that SIC Determination.				
GST	has the same meaning given to that expression in the GST Law.				
GST Law	has the meaning given to that expression in A New Tax System (Goods and Services Tax) Act 1999 (Cth) and any other Act or regulation relating to the imposition or administration of the GST.				
Land	the parcels of land identified in Schedule 3.				
Landcom	Landcom ABN 79 268 260 688 of Level 14, 60 Station Street, Parramatta NSW 2150.				
Landcom and Dahua 2 Development	means the development identified by the Landcom and Dahua 2 VPA as the 'Development'.				
Landcom and Dahua 2 VPA	Minis	ns the voluntary planning agreement between the ster, Landcom and Dahua 2 dated on or about late of this Deed.			
Law	ordir whet	statute, regulation, rule, proclamation, order, nance or by-law whether present or future and her Commonwealth, State, territorial or local, the common law.			

LEP	the Campbelltown Local Environmental Plan 2015 (NSW).
LRS	NSW Land Registry Services.
Menangle Park Development Area	means the Land and the parcels of land known as the "Land" in the Landcom and Dahua 2 VPA.
Menangle Park Urban Release Area	the Urban Release Area at Menangle Park, New South Wales.
Milestone 2 Payment	means the Funding Amount Allocation for Milestone 2 under the TIC Deed.
Monetary Contribution	a cash payment to the Minister (or RMS) in the amounts required by Schedule 4 .
Novation Deed	the deed attached at Annexure B.
Occupation Certificate	has the meaning given to that expression in the Act.
Overdraft Base Rate	the Commonwealth Bank of Australia's overdraft index rate as usually published by the Bank from time to time. Any change in that rate becomes effective as at the day specified in the relevant publication or, if no day is specified, the day on which the change is first published, or, if there is no such rate at any time, any substitute reference rate published by the Commonwealth Bank of Australia from time to time.
Per Dwelling SIC Amount	has the meaning given to that expression in clause 6(b)(ii).
Planning Agreement	a planning agreement under section 7.4 of the Act.
Real Property Act	the Real Property Act 1900 (NSW).
Register	the Torrens title register maintained under the Real Property Act.
Regulation	the Environmental Planning and Assessment Regulation 2000 (NSW).
Related Body Corporate	has the meaning given to that expression in section 50 of the <i>Corporations Act 2001</i> (Cth).
RMS	Roads and Maritime Services ABN 76 236 371 088, a NSW Government agency and corporation constituted under section 46 of the <i>Transport</i> <i>Administration Act 1988</i> (NSW).
Satisfactory Arrangements Certificate	a certificate issued by the Secretary that satisfactory arrangements have been made to contribute to the provision of Designated State Public Infrastructure in accordance with clause 6.1 of the LEP.

Secretary	the Secretary of the Department of Planning, Industry and Environment or any replacement position carrying out the same or a similar role.		
Security	has the meaning given to that term in the TIC Deed.		
Service Lot	means a registered lot that is created for one or more of the following purposes:		
	 (a) to be dedicated or otherwise transferred to an Authority; 		
	 (b) for any public utility undertaking (within the meaning of the Standard Instrument (Local Environmental Plans) Order 2006 as at the date of this Deed; 		
	 (c) for roads, open space, recreation, environmental conservation, water cycle management or riparian land management; or 		
	 (d) for avoidance of doubt – association property within the meaning of the <i>Community Land Development Act 1989</i> (NSW) that is to be used for any one or more of the purposes set out in (c) above. 		
SIC	a special infrastructure contribution required in accordance with a SIC Determination.		
SIC Credit	has the meaning given to that expression in clause 6 .		
SIC Determination	a development contribution (SIC) determined by the Minister in accordance with section 7.23 of the Act for a special contributions area that includes any part of the Land.		
Strata Certificate	has the same meaning given to that expression in the Strata Schemes Development Act 2015 (NSW).		
Subdivision Certificate	has the same meaning given to that expression in the Act.		
Sunset Date	has the meaning given to that term in the TIC Deed.		
Тах	a tax, duty (including stamp duty and any other transaction duty), levy, impost, charge, fee (including a registration fee) together with all interest, penalties, fines and costs concerning them.		
TIC Deed	the Transport Infrastructure Contributions Deed between RMS, Landcom, Dahua 2 and the Developer attached at Annexure A dated on or about the date of this Deed.		
Urban Release Area	has the meaning given to that expression in the LEP.		

2 Operation and Application of this Deed

2.1 Operation

This Deed will operate once it is signed by the parties.

2.2 Planning agreement under the Act

This Deed is a Planning Agreement governed by Part 7, Division 7.1, Subdivision 2 of the Act.

2.3 Application

This Deed applies to the Land and the Development.

3 Application of Sections 7.11, 7.12 and 7.24 of the Act

The application of sections 7.11, 7.12 and 7.24 of the Act are excluded to the extent stated in **Schedule 1**.

4 Development Contributions

4.1 Developer to provide Development Contributions

The Developer must provide, or must procure the provision of, the Development Contributions comprising the payment of the Monetary Contributions in accordance with **Schedule 4**.

4.2 Acknowledgment

The Developer acknowledges and agrees that the Minister:

- (a) has no obligation to use or expend, or to require RMS (where applicable) to use or expend, the Development Contribution for a particular purpose;
- (b) has no obligation to repay or provide any compensation or payment for the Development Contribution, except as provided for in the TIC Deed; and
- (c) has not made any representation or warranty that, if any part of the Development Contribution is transferred, dedicated or provided to another Authority, the Development Contribution will or must be used for any particular purpose by that or any other Authority.

4.3 Conditions Precedent under the TIC Deed

- (a) The parties acknowledge that the obligation for the Developer to pay all of the Monetary Contribution (Portion 1) is subject to the Conditions Precedent being satisfied (or waived in accordance with the TIC Deed) by the Sunset Date.
- (b) If the Conditions Precedent are not satisfied (or waived in accordance with clause 2.2 of the TIC Deed) by the Sunset Date and the TIC Deed is

terminated the parties agree to negotiate in good faith the implications under this Deed and until such time as this Deed is amended in accordance with **clause 16.4** to address the implications of termination of the TIC Deed, the Developer remains bound by all of the obligations under this Deed, including in relation to all Subdivision Certificate and Construction Certificate caps.

4.4 No satisfactory arrangements for over 925 Dwellings

For the avoidance of doubt, provision of the Development Contributions under this Deed does not enable a Satisfactory Arrangements Certificate to be issued for the subdivision of the Land to develop more than 925 Dwellings.

5 Interest for late provision of Monetary Contribution

- (a) If the Developer is required to pay a Monetary Contribution under this Deed and fails to do so on the due date for provision of that Monetary Contribution, the Developer must also pay to the Minister or RMS (as instructed by the Minister) interest on the overdue amount at a rate of 2% above the Overdraft Base Rate charged by the Commonwealth Bank of Australia from time to time.
- (b) Interest will be payable on the daily balance of amounts due from the due date for provision of those amounts until those amounts (including interest on those amounts) have been paid to the Minister.

6 Effect of future SIC Determination

- (a) This clause 6 applies if a SIC Determination is made before the Subdivision Certificate or Strata Certificate is issued for the subdivision of land into Final Lots on which the 925th Dwelling in the Development is to be developed and the Per Dwelling SIC Amount is less than \$25,000.
- (b) Within 20 Business Days of the date the last Development Contribution is made, the Developer is to notify the Minister of:
 - the number of **Post SIC Dwellings** in the Development, being 925 less the number of Dwellings for which a Subdivision Certificate or Strata Certificate had been issued as at the date of the SIC Determination;
 - (ii) the amount which is payable per Dwelling under the SIC Determination (or, if the SIC Determination requires contributions to be paid other than on a per Dwelling basis, the total amount of contributions that, but for this Deed, would have been payable for the Development under that SIC Determination, divided by 925) as at the date of the SIC Determination (**Per Dwelling SIC Amount**); and
 - (iii) the **SIC Credit**, being an amount calculated in accordance with the following formula:

SIC Credit = Post SIC Dwellings *multiplied by* (\$25,000 *less* the Per Dwelling SIC Amount)

- (c) Notwithstanding any other provision in this clause 6, the SIC Credit accrues to the Developer once all Development Contributions obligations have been satisfied and provided that the Developer is not otherwise in material default of any of its obligations under this Deed.
- (d) Once accrued to the Developer, the SIC Credit may be used by the Developer or its Related Body Corporate to satisfy an obligation to make a SIC relating to any land which is located within the Greater Macarthur Growth Area.
- (e) Nothing in this **clause 6** entitles the Developer to a cash refund from the Minister for any amount paid under this Deed by the Developer.

7 Effect of termination of TIC Deed

If the TIC Deed is terminated by RMS in accordance with clause 12 of the TIC Deed:

- (a) for the purposes of clause 12.2 of the TIC Deed, in addition to any amount which RMS is entitled to retain under the TIC Deed, the Minister may direct RMS to retain:
 - (i) all or any sums of money in the hands of RMS in respect of the TIC Deed; and
 - (ii) all Security provided to RMS under the TIC Deed,

so that the value of the Development Contributions provided by the Developer as at the date of termination of the TIC Deed are equivalent to the number of Dwellings which are developed or are expected to be delivered as part of the Development on the Land, and from which the Deed has been released and discharged under **clause 10.5** as at the date the TIC Deed was terminated, multiplied by \$25,000 being the notional contribution rate under this Deed; and

(b) the parties agree to negotiate in good faith the implications under this Deed and until such time as this Deed is amended in accordance with clause 16.4 to address the implications of termination of the TIC Deed, the Developer remains bound by all of the obligations under this Deed, including in relation to all Subdivision Certificate and Construction Certificate caps.

8 Publicity

The Developer will not refuse any reasonable request by the Minister to provide copies of photographic montages, plans or other materials in the Developer's possession which are necessary to assist the Minister to publish a media release or announcement about the Development Contributions required under this Deed, including the TIC Deed, provided that the Developer holds sufficient intellectual property rights in those materials to allow the Minister to use them for that purpose.

9 Enforcement

9.1 Developer to provide security

The Developer has agreed to provide security to the Minister for the performance of the Developer's obligation to provide the Development Contributions under this Deed by:

- (a) registering this Deed on the title of the Land in accordance with clause 10;
- (b) providing Security for the Funding Amount in accordance with the TIC Deed; and
- (c) providing one or more Bank Guarantees to the Minister in accordance with the terms and procedures set out in **clause 4** of **Schedule 5**.

9.2 Restrictions on subdivision

- (a) The requirement to pay or make a Development Contribution by a timeframe required in this Deed constitutes a restriction on the issue of a Subdivision Certificate within the meaning of section 6.15 of the Act, with the effect that, if the required timeframe for the provision of a Development Contribution has passed (or the Developer applies for a Subdivision Certificate which, if issued, would cause that timeframe to pass), a Subdivision Certificate may not be issued unless and until the requirement to pay or make that Development Contribution has been discharged.
- (b) The Completion and opening of the Carriageway must occur in accordance with the TIC Deed prior to the issue of a Subdivision Certificate in respect of the subdivision of the Land into Final Lots on which the 1800th Dwelling in the Menangle Park Development Area is to be developed as part of the Development or the Landcom and Dahua 2 Development. The parties agree that this **clause 9.2(b)** constitutes a restriction on the issue of a Subdivision Certificate within the meaning of section 6.15 of the Act.
- (c) The requirement to transfer the Education Land by the timeframe required by the Landcom and Dahua 2 VPA (as that timeframe may be extended under that deed) constitutes a restriction on the issue of a Subdivision Certificate within the meaning of section 6.15 of the Act, with the effect that, if the required timeframe for transfer of the Education Land has passed (or the Developer applies for a Subdivision Certificate which, if issued, would cause that timeframe to pass), a Subdivision Certificate may not be issued unless and until the requirement to transfer the Education Land is discharged.

(d) The requirement to transfer the Dedicated Land by the timeframe required by the TIC Deed (as that timeframe may be extended under that deed) constitutes a restriction on the issue of a Subdivision Certificate within the meaning of section 6.15 of the Act, with the effect that, if the required timeframe for transfer of the Dedicated Land has passed (or the Developer applies for a Subdivision Certificate which, if issued, would cause that timeframe to pass), a Subdivision Certificate may not be issued unless and until the requirement to transfer the Dedicated Land is discharged.

9.3 Restriction on Construction Certificates

- (a) If, for any reason, the TIC Deed is terminated before the Developer's obligation to provide the Funding Amount and the Dedicated Land has been fully satisfied under the TIC Deed, a Construction Certificate must not be issued for subdivision work and building work comprising the erection of a Dwelling on a Final Lot as part of the Development unless:
 - (i) the Developer has fulfilled its Development Contribution obligations in accordance with Schedule 4 in so far as they relate to the part of the Land on which the Dwelling is to be erected, including, if applicable, in respect of the timing for the transfer of the Education Land and Dedicated Land; and
 - (ii) Development Contributions of a proportionate value have been made under this Deed and by reference to the notional contribution rate under this Deed of \$25,000 per Dwelling.
- (b) The parties agree that this **clause 9.3** constitutes a restriction on the issue of a Construction Certificate within the meaning of clause 146A of the Regulation.

10 Land ownership and registration of Deed

10.1 Land ownership

The Developer represents and warrants that the Developer is the registered proprietor of the Land.

10.2 Registration of Deed

Within 10 Business Days of receiving a copy of this Deed executed by the Minister, the Developer must, at its expense, take all practical steps and otherwise do anything necessary to procure:

- (a) the consent of each person, as required by the Registrar-General, who:
 - (i) has an estate or interest in the Land registered under the Real Property Act; or
 - (ii) is seized or possessed of an estate or interest in the Land,

to the registration of this Deed on the title of the Land and to the terms of this Deed;

- (b) the execution of any documents required to enable registration of this Deed on the title of the Land;
- (c) the production of the certificates of title for the Land; and
- (d) the lodgement for registration of this Deed by the Registrar-General in the relevant folios of the Register for the Land.

10.3 Evidence of registration

The Developer must provide the Minister with:

- (a) evidence of the lodgement of this Deed pursuant to clause 10.2(d) within 10 Business Days of such lodgement at the LRS; and
- (b) a copy of the relevant folios of the Register and a copy of all registered dealings for the Land within 10 Business Days of registration of this Deed.

10.4 Right to lodge caveat

- (a) Subject to clause 10.4(b), until such time as this Deed is registered on the title of the Land in accordance with clause 10.2, the Developer acknowledges that this Deed confers on the Minister an interest in the Land and entitles the Minister to lodge and maintain a caveat on the title to the Land to prevent any Dealing in respect of the Land.
- (b) If the Minister lodges a caveat in accordance with **clause 10.4(a)**, then the Minister will do all things reasonably necessary to:
 - (i) ensure that the caveat does not prevent or delay the registration of this Deed; and
 - (ii) remove the caveat from the title to the Land promptly, following registration of this Deed in accordance with **clause 10.2**.
- (c) If the Developer has failed or has been unable to comply with clause 10.2 within the time frames specified in that clause, the Developer must pay the Minister's reasonable costs and expenses, including legal costs, of exercising the Minister's rights under clause 10.4(a) to lodge and withdraw a caveat(s) (as applicable).

10.5 Release and discharge of Deed

- (a) The Developer will be released from its obligations under this Deed (other than the obligation to grant an easement for batter and drainage under the TIC Deed and the reporting obligations in clause 15) once the Developer has satisfied its obligations to provide the Development Contributions in accordance with this Deed.
- (b) After the Developer has satisfied its obligations to provide the Development Contributions in accordance with this Deed, the Minister will do all things reasonably required by the Developer to have the Registrar-General remove this Deed from the relevant folio(s) of the Register.

- (c) From time to time, the Developer may request that the Minister facilitate the progressive release and removal of this Deed from the relevant folio(s) of the Register for any part of the Land for which a Subdivision Certificate or Strata Certificate has been issued. For the avoidance of doubt, a request may be made under this **clause 10.5(c)** in advance of the issue of the relevant Subdivision Certificate or Strata Certificate.
- (d) The request under clause 10.5(c) must be accompanied by:
 - (i) a table identifying the number of:
 - (A) Dwellings for which a Subdivision Certificate or a Strata Certificate has been issued for each stage of the Development;
 - (B) Dwellings proposed to be developed as part of the Development on the part of the Land to which the request under clause 10.5(c) relates; and
 - (C) Occupation Certificates which have been issued for the Development; and
 - (ii) an aerial photograph of the Development with sufficient detail to show the number of Dwellings on the Land.
- (e) Within 30 Business Days of receipt of a request from the Developer under **clause 10.5(b)** and:
 - (i) any additional period of time during which the Minister, acting reasonably, has requested further information from the Developer which is necessary for the Minister to consider the Developer's request and the Developer has not provided that information; and
 - (ii) for a request made in advance of the issue of a Subdivision Certificate or Strata Certificate, within a further 5 Business Days after the Minister is provided with a copy of the relevant Subdivision Certificate or Strata Certificate,

the Minister must facilitate the release and removal of this Deed from the relevant folio(s) of the Register for the part(s) of the Land forming the subject of the request, subject to the Developer satisfying the Minister:

- (iii) that if the required timeframe for provision of a Development Contribution under this Deed has passed, that the requirement to pay or make that Development Contribution has been discharged;; and
- (iv) that:

$$\frac{\$DCV}{\$23,125,000} \ge \frac{N}{925}$$

Where:

DCV is the value of the Development Contributions which have been provided as at the date of the request; and

N is the number of Dwellings which are developed, or will be developed, on the Land as part of the Development as enabled by the issue of Subdivision Certificate(s) and/ or Strata Certificate(s) as at the date of the request; and

- (v) that the Developer is not otherwise in material default of any of its obligations under this Deed.
- (f) Notwithstanding any other provision in this Deed, the Developer must not request a release and discharge of this Deed under this **clause 10.5**:
 - (i) if the required timeframe for transfer of the Education Land under the Landcom and Dahua 2 VPA has passed (as that timeframe may be extended under that deed), and the requirement to transfer the Education Land is not discharged;
 - (ii) if the required timeframe for transfer of the Dedicated Land under the TIC Deed has passed (as that timeframe may be extended under that deed), and the requirement to transfer the Dedicated Land is not discharged; and
 - (iii) for the part of the Land on which the 1801st Dwelling will be developed in the Menangle Park Development Area as part of the Development or the Landcom and Dahua 2 Development unless the Carriageway is completed and open to be public.
- (g) The Minister may, but has no obligation to, agree to the release and discharge of this Deed from the whole or any part of the Land under this clause 10.5, on the basis of Security provided, and contributions paid, for the Milestone 2 Payment under the TIC Deed, in circumstances where a construction contract for the Carriageway has not been awarded.

11 Dispute Resolution

11.1 Not commence

A party must not commence any court proceedings relating to a dispute unless it complies with this **clause 11**.

11.2 Written notice of dispute

A party claiming that a dispute has arisen under or in relation to this Deed must give written notice to the other party specifying the nature of the dispute and requiring the dispute be addressed in accordance with this **clause 11**.

11.3 Attempt to resolve

On receipt of a notice under **clause 11.2**, the parties must endeavour in good faith to resolve the dispute expeditiously using informal dispute resolution techniques such as mediation, expert evaluation or other techniques agreed by them.

11.4 Mediation

If the parties do not agree within 21 Business Days of receipt of a notice under **clause 11.2** (or any further period agreed in writing by them) as to:

- (a) the dispute resolution technique and procedures to be adopted;
- (b) the timetable for all material steps in those procedures; or
- (c) the selection and compensation of the independent person required for such technique,

then, subject to **clause 11.5**, the parties must mediate the dispute in accordance with the Law Society of NSW's Mediation Program. The parties must, as soon as possible, request the president of the Law Society of NSW or the president's nominee to select the mediator and determine the mediator's remuneration.

11.5 Court proceedings

If the dispute is not resolved within 60 Business Days after notice is given under **clause 11.2** (or any other period agreed in writing by the parties) then any party which has complied with the provisions of this **clause 11** may, by written notice to the other parties, terminate any dispute resolution process undertaken under this clause and either party may then commence court proceedings in relation to the dispute.

11.6 Use of information

The parties acknowledge the purpose of any exchange of information or documents or the making of any offer of settlement under this **clause 11** is to attempt to resolve the dispute. No party may use any information or documents obtained through any dispute resolution process undertaken under this **clause 11** for any purpose other than in an attempt to resolve the dispute.

11.7 No prejudice

This **clause 11** does not prejudice the right of a party to institute court proceedings for urgent injunctive or declaratory relief in relation to any matter arising out of or relating to this Deed.

11.8 Continued performance of obligations

Despite the existence of a dispute under this **clause 11**, but subject to any order of a court or the agreement of the parties, the parties must continue to perform their obligations under this Deed.

12 GST

12.1 Definitions

Words used in this clause that are defined in the GST Law have the meaning given in that legislation.

12.2 Intention of the parties

The parties intend that:

- (a) Divisions 81 and 82 of the GST Law apply to the supplies made under and in respect of this Deed; and
- (b) no additional amounts will be payable on account of GST and no tax invoices will be exchanged between the parties.

12.3 Reimbursement

Subject to **clause 16.17(a)**, any payment or reimbursement required to be made under this Deed that is calculated by reference to an amount paid or incurred will be limited to the total amount less the amount of any input tax credit to which an entity is entitled for the acquisition to which the amount relates.

12.4 Consideration GST exclusive

Unless otherwise expressly stated, all prices or other sums payable or consideration to be provided under this Deed are exclusive of GST . Any consideration that is specified to be inclusive of GST must not be taken into account in calculating the GST payable in relation to a supply for the purposes of **clause 12.5**.

12.5 Additional Amounts for GST

To the extent an amount of GST is payable on a supply made by a party (**Supplier**) under or in connection with this Deed (the **GST Amount**), the Recipient will pay to the Supplier the GST Amount. However, where a GST Amount is payable by the Minister as recipient of the supply, the Developer will ensure that:

- the Developer makes payment of the GST Amount on behalf of the Minister, including any gross up that may be required; and
- (b) the Developer provides a Tax Invoice to the Minister.

12.6 Non monetary consideration

Clause 12.5 applies to non-monetary consideration.

12.7 Assumptions

The Developer acknowledges and agrees that in calculating any amounts payable under **clause 12.5** the Developer will assume the Minister is not entitled to any input tax credit.

12.8 No merger

This clause will not merge on completion or termination of this Deed.

13 Assignment and Dealings

13.1 Assignment or novation by Developer

- (a) Prior to a proposed assignment or novation of its rights or obligations under this Deed, the Developer must seek the written consent of the Minister and:
 - satisfy the Minister, acting reasonably, that the person to whom the Developer's rights or obligations are to be assigned or novated (Incoming Party) has sufficient assets, resources and expertise required to perform the Developer's obligations under this Deed insofar as those obligations are to be novated to the Incoming Party;
 - (ii) satisfy the Minister, acting reasonably, that the Developer is not in material breach of its obligations under this Deed; and
 - (iii) procure the execution of an agreement with the Minister, the Developer and the Incoming Party substantially in the form of the Novation Deed.
- (b) The Minister may impose reasonable conditions on the Minister's entry into the Novation Deed under clause 13.1(a)(iii) including requirements that the Incoming Party provide additional security to that identified in clause 9.1 and that the Incoming Party indemnify the Minister in respect of any breach of this Deed by the Incoming Party.
- (c) The Developer must pay the Minister's reasonable costs incurred under this **clause 13.1** including legal costs and disbursements.

13.2 Assignment by Minister

The Minister may assign the rights and benefits of this Deed in the Minister's absolute discretion, without the need for consent from the Developer.

13.3 Dealings with the Land

- (a) Subject to clause 13.3(b), the Developer must not enter into any Dealing with a third party (Transferee) relating to its interest in the Land unless before the Dealing is entered into with the Transferee:
 - the Developer satisfies the Minister (acting reasonably) that the proposed Transferee is financially capable of complying with the Developer's obligations under this Deed;
 - the Transferee enters into a deed in favour of the Minister substantially in the form of the Novation Deed, under which the Transferee agrees to comply with the Developer's obligations under this Deed as if the Transferee was the Developer;
 - (iii) any default by the Developer under this Deed has been remedied by the Developer, unless that default has been waived expressly by the Minister (or RMS as the case may be) in writing; and

- (iv) the Developer and the Transferee pay all costs reasonably incurred by the Minister in relation to that Dealing, including legal costs on a GST inclusive basis.
- (b) Clause 13.3(a) does not apply to any portion(s) of the Land in respect of which the Minister has granted a release pursuant to clauses 10.5(c) to (f).

14 Capacity

14.1 General warranties

- (a) Each party warrants to the other party that:
 - (i) this Deed creates legal, valid and binding obligations, enforceable against the relevant party in accordance with its terms; and
 - (ii) unless otherwise stated, it has not entered into this Deed in the capacity of trustee of any trust.
- (b) The Developer represents and warrants to the Minister that:
 - (i) it is duly registered and remains in existence;
 - the execution, delivery and performance of this Deed does not violate its constitution or any Law applying to it;
 - (iii) the execution, delivery and performance of this Deed does not violate any Law, or any document or agreement to which it is a party or which is binding on it or the Land, including any matters or interests not registered on the title of the Land;
 - (iv) any document or agreement to which it is a party or which is binding on it or any of its assets, including any matters or interests not registered on the title of the Land, does not in any way limit its ability to perform its obligations under this Deed; and
 - (v) no litigation, arbitration, mediation, conciliation, criminal or administrative proceedings are current, pending or, to its knowledge, threatened, which, if adversely determined, would or could have a material adverse effect on it or the Land, which has not been advised in writing to the Minister.

14.2 Power of attorney

If an attorney executes this Deed on behalf of any party, the attorney declares that it has no notice of the revocation of that power of attorney.

15 Reporting Requirement

(a) The Developer must provide at least 20 Business Days' written notice to the Minister of any application for a Construction Certificate, Occupation Certificate, Subdivision Certificate or Strata Certificate in relation to the Development.

- (b) If any part of the Development is to be carried out without the need for a Subdivision Certificate, Strata Certificate or a Construction Certificate, and which triggers any obligation under this Deed, then the Developer must provide at least 20 Business Days' written notice to the Minister of the date of:
 - (i) the proposed commencement of that part of the Development; and
 - (ii) the date of any application for a Complying Development Certificate in respect of that part of the Development.
- (c) On 1 September of each year, until the Development is complete, the Developer must deliver to the Secretary a report which must include those matters set out below:
 - (i) a description of the status of the Development and the provision of the Development Contributions;
 - (ii) a forecast in relation to the anticipated progression and completion of the Development;
 - (iii) an estimated date for when the Developer expects any application for a Subdivision Certificate to be submitted for the Development in respect of any part of the Land;
 - (iv) the estimated and actual number of Dwellings created and Occupation Certificates granted to date as part of the Development;
 - (v) an indicative forecast of the number of Dwellings estimated for completion as part of the Development over the next 5 years; and
 - (vi) the actual or expected (as applicable) land uses for the Land as part of the Development.
- (d) Within five Business Days following receipt of a written request from the Secretary, the Developer must deliver to the Secretary all documents and other information which, in the reasonable opinion of the Secretary, are necessary for the Secretary to assess the status of the Development.

16 General Provisions

16.1 Liability

An obligation of two or more persons binds them separately and together.

16.2 Entire agreement

This Deed constitutes the entire agreement between the parties regarding the matters set out in it and supersedes any prior representations, understandings or arrangements made between the parties, whether orally or in writing.

16.3 Inconsistency

If there is any inconsistency between this document and the TIC Deed, this Deed prevails to the extent of that inconsistency.

16.4 Variation

This Deed must not be varied except by a later written document executed by all parties.

16.5 Waiver

A right created by this Deed cannot be waived except in writing signed by the party entitled to that right. Delay by a party in exercising a right does not constitute a waiver of that right, nor will a waiver (either wholly or in part) by a party of a right operate as a subsequent waiver of the same right or of any other right of that party.

16.6 Further assurances

Each party must promptly execute all documents and do everything necessary or desirable to give full effect to the arrangements contained in this Deed.

16.7 Time for doing acts

(a) If:

- (i) the time for doing any act or thing required to be done; or
- (ii) a notice period specified in this Deed,

expires on a day other than a Business Day, the time for doing that act or thing or the expiration of that notice period is extended until the following Business Day.

(b) If any act or thing required to be done is done after 5.00pm on the specified day, it is taken to have been done on the following Business Day.

16.8 Governing law and jurisdiction

- (a) The laws applicable in New South Wales govern this Deed.
- (b) The parties submit to the non-exclusive jurisdiction of the courts of New South Wales and any courts competent to hear appeals from those courts.

16.9 Severance

If any clause or part of any clause is in any way unenforceable, invalid or illegal, it is to be read down so as to be enforceable, valid and legal. If this is not possible, the clause (or where possible, the offending part) is to be severed from this Deed without affecting the enforceability, validity or legality of the remaining clauses (or parts of those clauses) which will continue in full force and effect.

16.10 Preservation of existing rights

The expiration or termination of this Deed does not affect any right that has accrued to a party before the expiration or termination date.

16.11 No merger

Any right or obligation of any party that is expressed to operate or have effect on or after the completion, expiration or termination of this Deed for any reason, will not merge on the occurrence of that event but will remain in full force and effect.

16.12 Counterparts

This Deed may be executed in any number of counterparts. All counterparts taken together constitute one instrument.

16.13 Relationship of parties

Unless otherwise stated:

- (a) nothing in this Deed creates a joint venture, partnership, or the relationship of principal and agent, or employee and employer between the parties; and
- (b) no party has the authority to bind any other party by any representation, declaration or admission, or to make any contract or commitment on behalf of any other party or to pledge any other party's credit.

16.14 Good faith

Each party must act in good faith towards all other parties and use its best endeavours to comply with the spirit and intention of this Deed.

16.15 No fetter

Nothing in this Deed shall be construed as requiring the Minister to do anything that would cause the Minister to breach any of the Minister's obligations at law and, without limitation, nothing in this Deed shall be construed as limiting or fettering in any way the discretion of the Minister in exercising any of the Minister's statutory functions, powers, authorities or duties.

16.16 Explanatory Note

The Explanatory Note must not be used to assist in construing this Deed.

16.17 Costs, expenses and stamp duty

- (a) The Developer must pay its own and the Minister's reasonable GST inclusive legal and valuation costs and disbursements in connection with the negotiation, preparation, execution and carrying into effect of this Deed.
- (b) The Developer must pay for all reasonable costs and expenses associated with the giving of public notice of this Deed and the Explanatory Note in accordance with the Regulation.

- (c) The Developer must pay all Taxes assessed on or in respect of this Deed and any instrument or transaction required or contemplated by or necessary to give effect to this Deed.
- (d) The Developer must provide the Minister with bank cheques in respect of the Minister's costs pursuant to **clauses 16.17(a)** and **(b)**:
 - where the Minister has provided the Developer with written notice of the sum of such costs prior to execution of this Deed, on the date of execution of this Deed; or
 - (ii) where the Minister has not provided the Developer with prior written notice of the sum of such costs prior to execution, within 10 Business Days of demand by the Minister for payment.

16.18 Notices

- (a) Any notice, demand, consent, approval, request or other communication (Notice) to be given under this Deed must be in writing and must be given to the recipient at its Address for Service by being:
 - (i) hand delivered;
 - (ii) sent by email; or
 - (iii) sent by prepaid ordinary mail within Australia.
- (b) A Notice is given if:
 - (i) hand delivered, on the date of delivery;
 - (ii) sent by email during any Business Day, on the date that the sending party's email records indicate the email was sent, unless the sender receives a delivery failure notification indicating that the email has not been delivered to the addressee; or
 - sent by prepaid ordinary mail within Australia, on the date that is 5 Business Days after the date of posting.

16.19 Interpretation

In this Deed unless the context clearly indicates otherwise:

- (a) A reference to the Minister, includes a reference to the Secretary and any person specifically nominated by the Minister for the purposes of this Deed or for Planning Agreements generally to which the Minister is a party;
- (b) a reference to **this Deed** or another document means this Deed or that other document and any document which varies, supplements, replaces, assigns or novates this Deed or that other document;
- (c) a reference to legislation, a legislative or other instrument (such as a local environmental plan) or a legislative provision includes any modification, amendment or substitution of that legislation, instrument or legislative provision and any subordinate legislation issued under that legislation or legislative provision;

- (d) a reference to a **body** or **authority** which ceases to exist is a reference to either a body or authority that the parties agree to substitute for the named body or authority or, failing agreement, to a body or authority having substantially the same objects as the named body or authority;
- (e) a reference to the **introduction**, a **clause**, **schedule** or **annexure** is a reference to the introduction, a clause, a schedule or an annexure to or of this Deed;
- (f) **clause headings**, **the introduction** and the **table of contents** are inserted for convenience only and do not form part of this Deed;
- (g) the schedules and annexures form part of this Deed;
- (h) a reference to a **person** includes a natural person, corporation, statutory corporation, partnership, the Crown or any other organisation or legal entity;
- (i) a reference to a **natural person** includes their personal representatives, successors and permitted assigns;
- (j) a reference to a **corporation** includes its successors and permitted assigns;
- (k) a reference to a right or obligation of a party is a reference to a right or obligation of that party under this Deed;
- an obligation or warranty on the part of 2 or more persons binds them jointly and severally and an obligation or warranty in favour of 2 or more persons benefits them jointly and severally;
- (m) a requirement to do any thing includes a requirement to cause that thing to be done and a requirement not to do any thing includes a requirement to prevent that thing being done;
- (n) including and includes are not words of limitation;
- (o) a word that is derived from a defined word has a corresponding meaning;
- (p) monetary amounts are expressed in Australian dollars;
- (q) the singular includes the plural and vice-versa;
- (r) words importing one gender include all other genders;
- (s) a reference to a thing includes each part of that thing; and
- (t) neither this Deed nor any part of it is to be construed against a party on the basis that the party or its lawyers were responsible for its drafting.

Requirements under the Act

The parties acknowledge and agree that the table set out below provides for certain terms, conditions and procedures for the purpose of this Deed complying with the Act.

Table 1 – Requirements under section 7.4 of the Act

Req	uirement under the Act	This	Deed	
	<pre>ining instrument and/or elopment application - (section 1))</pre>			
The	Developer has:			
(a)	sought a change to an environmental planning instrument.	(a)	Yes	
(b)	made, or proposes to make, a Development Application.	(b)	Yes	
(c)	entered into an agreement with, or is otherwise associated with, a person, to whom paragraph (a) or (b) applies.	(c)	N/A	
	cription of land to which this Deed lies – (section 7.4(3)(a))	See Schedule 3		
Description of development to which this Deed applies – (section 7.4(3)(b))			Development as defined in se 1 .	
The scope, timing and manner of delivery of contribution required by this Deed – (section 7.4(3)(c))		See	Schedule 4.	
Applicability of sections 7.11 and 7.12 of the Act – (section 7.4(3)(d))		The application of sections 7.11 and 7.12 of the Act are not excluded in respect of the Development.		
Applicability of section 7.24 of the Act – (section 7.4(3)(d))		The application of section 7.24 of the Act is excluded in respect of the Development.		
Con	sideration of benefits under this	The	Development Contributions to	

Requirement under the Act	This Deed
Deed if section 7.11 applies – (section 7.4(3)(e))	be provided by the Developer under this Deed must not be taken into consideration in determining a development contribution in respect of the Development under section 7.11 of the Act.
Mechanism for Dispute Resolution – (section 7.4(3)(f))	See clause 11.
Enforcement of this Deed – (sections 7.4(3)(g) and 7.6)	See clauses 5, 9.1, 9.3 and 10.
No obligation to grant consent or exercise functions – (section 7.4(10))	See clause 16.15.

Table 2 – Other Matters

Requirement under the Act or Regulation	This Deed
Registration of the Planning Agreement – (section 7.6 of the Act)	Yes (see clause 9.3)
Whether the Planning Agreement specifies that certain requirements of the agreement must be complied with before a construction certificate is issued – (clause 25E(2)(g) of the Regulation)	Yes
Whether the Planning Agreement specifies that certain requirements of the agreement must be complied with before an occupation certificate is issued – (clause 25E(2)(g) of the Regulation)	No
Whether the Planning Agreement specifies that certain requirements of the agreement must be complied with before a subdivision certificate is issued – (clause 25E(2)(g) of the	Yes
1442,3062,0428,0	200

Requirement under the Act or Regulation This Deed

Regulation)

Address for Service (clause 1)

<u>Minister</u>	
Contact:	The Secretary
Address:	Department of Planning, Industry and Environment 320 Pitt Street SYDNEY NSW 2000
Email:	PlanningAgreements@planning.nsw.gov.au
<u>Developer</u>	
Contact:	Yuxing (Eric) Li
	General Manager
	Dahua Australia
Address:	Suite 2, Level 20, 201-217 Elizabeth Street, Sydney NSW 2000
Email:	eric@dahuaaustralia.com.au

Land (clause 1)

	Registered Proprietor	Loji	Deposited Plan	Folio Identifier
1.	Dahua Group Sydney Project 3 Pty Ltd	3	DP236059	3/236059
2.	Dahua Group Sydney Project 3 Pty Ltd	124	DP1097090	124/1097090
3.	Dahua Group Sydney Project 3 Pty Ltd	7	DP787284	7/787284
4.	Dahua Group Sydney Project 3 Pty Ltd	1	DP727098	1/727098
5.	Dahua Group Sydney Project 3 Pty Ltd	1	DP708770	1/708770
6.	Dahua Group Sydney Project 3 Pty Ltd	1	DP707225	1/707225
7.	Dahua Group Sydney Project 3 Pty Ltd	1	DP249393	1/249393
8.	Dahua Group Sydney Project 3 Pty Ltd	33	DP1101983	33/1101983
9.	Dahua Group Sydney Project 3 Pty Ltd	31	DP1101983	31/1101983
10.	Dahua Group Sydney Project 3 Pty Ltd	125	DP1097138	125/1097138
11.	Dahua Group Sydney Project 3 Pty Ltd	1	DP1247661	1/1247661

Development Contribution (clause 4)

1 Development Contribution

The Developer must provide the following Development Contributions in accordance with this **Schedule 4**:

Column A:	Column B:	Column C;
Development Contribution	Value	Timing
Monetary Contribution (Portion 1)	The Funding Amount payable by the Developer in respect of the Land, in accordance with the TIC Deed, being \$23,000,000 or, if another amount is specified as a result of the adjustment to the Funding Amount Allocation pursuant to clause 5.4 of the TIC Deed, then that amount.	In accordance with clause 5 and Column 3 of Annexure B to the TIC Deed.
Monetary Contribution (Portion 2)	\$125,000	Upon execution of this Deed.

2 Monetary Contributions

The Developer must pay:

- (a) the Monetary Contribution (Portion 1) by way of payment of the Funding Amount to RMS in accordance with the TIC Deed; and
- (b) the Monetary Contribution (Portion 2) by way of payment to the Minister upon execution of this Deed.

Security Terms

1 Provision of Security

In order to secure the provision of the Development Contributions in accordance with **Schedule 4**, the Developer will provide security in accordance with this **Schedule 5**.

2 Security for the Development Contributions

The parties agree that the obligation of the Developer to provide the Development Contributions in accordance with **Schedule 4** is secured by:

- (a) clause 9.1(c) and clause 9.2 of this Deed;
- (b) clause 9.3 of this Deed;
- (c) clause 10 of this Deed; and
- (d) clause 4 of this Schedule 5.

3 Security under the TIC Deed

- (a) At the date of this Deed the Developer (in addition to Landcom and Dahua 2) has entered into a TIC Deed with RMS.
- (b) The Minister accepts the Security provided under the TIC Deed and the Security provided under clause 4 of this Schedule 5 by the Developer as security for the performance of the Developer's obligation under this Deed to provide the Monetary Contribution (Portion 1).

4 Bank Guarantee

- (a) Upon execution of this Deed, the Developer must provide to the Minister one or more Bank Guarantees with a total face value of \$200,000, as security for the provision of the Development Contributions.
- (b) Each Bank Guarantee provided by the Developer to the Minister under this Deed must:
 - (i) name the "Minister for Planning and Public Spaces ABN 20 770 707 468" as the relevant beneficiary; and

- (ii) not have an expiry date.
- (c) The Minister may call upon any Bank Guarantees or may request RMS to call on any Security under the TIC Deed where the Developer has failed to comply with this Deed and the Minister may retain monies obtained from that Bank Guarantee or Security and apply those monies towards the costs and expenses incurred by the Minister in rectifying such failures by the Developer. Any expended costs and expenses incurred by the Minister in rectifying such failures by the Developer becomes a debt due and payable promptly on demand.
- (d) Prior to calling upon the Bank Guarantees or Security provided by the Developer under this Schedule 5, the Minister must give the Developer not less than 10 Business Days' written notice.
- (e) If the Minister:
 - (i) calls upon the Bank Guarantees or Security in accordance with clause 4 of this Schedule 5; and
 - (ii) applies all or part of the moneys obtained from that call in the manner authorised in **clause 4** of this **Schedule 5**,

then

- (iii) the Minister must notify the Developer in writing of the amount of the call; and
- (iv) promptly after receiving that notice, the Developer must provide to the Minister or RMS (as the case may be) a replacement Bank Guarantee or Security to ensure that the Developer complies with clause 4(a) of this Schedule 5.
- (f) If:
 - (i) the Developer has made the Development Contributions as required by this Deed;
 - (ii) the Developer provides the Minister with written notice containing:
 - (A) evidence that the Monetary Contributions have been paid; and
 - (B) a request for return of the Bank Guarantees or Security held by the Minister; and
 - (iii) the Minister is satisfied that those circumstances have occurred,
- (g) then the Minister must promptly return those Bank Guarantees or Security held by the Minister (less any costs, charges, duties and Taxes payable), or if there has been a call on those Bank Guarantees in accordance with this Deed, the remainder of the monies secured by those Bank Guarantees or Security, to the Developer The release of a Bank Guarantee or Security under this **Schedule 5** may involve the substitution of a Bank Guarantee or Security with another Bank Guarantee or Security having a different face value.

5 Right to Call for Additional Security

- (a) If the Developer breaches any of its obligations under this Deed including the TIC Deed or fails to provide, increase or maintain the Bank Guarantee or Security under this Deed, including the TIC Deed:
 - (i) any such breach will constitute a material breach of this Deed; and
 - (ii) the Minister may provide written notice to the Developer requesting that the Developer provide a Bank Guarantee to the Minister for an amount determined by the Minister, acting reasonably, that will rectify the Developer's breach of this Deed or the TIC Deed and ensure that sufficient Security is available for the Monetary Contributions.
- (b) Within 10 Business Days following receipt of a written notice pursuant to clause 5(a) of this Schedule 5, the Developer must provide the Bank Guarantee to the Minister.

6 Right to Call for Security

If, for any reason:

- (a) RMS does not call on any Security under the TIC Deed;
- (b) that Security has been relied on by the Secretary to enable the issue of a Subdivision Certificate or otherwise to allow the Minister to provide a release and discharge of the Deed under clause 10.5; and
- (c) RMS advises the Minister that it does not intend to call on that Security,

then the Minister may call on that Security for such an amount so that sufficient Development Contributions have been made under this Deed for the number of Dwellings on the Land from which this Deed has been released and discharged.

Execution

Executed as a deed.

Executed by the **Minister for Planning** and **Public Spaces** in the presence of:

)

))

Signature of Witness

AMANDA BEAUMONT Name of Witness (print)

Signature of Minister ETWHITW Name of Minister the His

EXECUTED by Dahua Group Sydney)Project 3 Pty Ltd ACN 606 391 922 in)accordance with s127(1) of the)Corporations Act 2001 (Cth):)

Gompany-Secretary/Director

.....

.....

Name of Company_Secretary/Director (print)

....

Director

Name of Director (print)
Annexure A

TIC Deed

Roads and Maritime Services

Landcom

Dahua Group Sydney Project 2 Pty Ltd

Dahua Group Sydney Project 3 Pty Ltd

Menangle Park -Spring Farm Parkway Transport Infrastructure Contributions Deed

.

Contents

1	Interpretation		
	1.1 Definitions	2	
	1.2 Interpretation	8	
	1.3 A party and its Personnel	9	
	1.4 Order of precedence	9	
2	Operation of this Deed		
	2.1 Conditions Precedent	9	
	2.2 Waiver of Conditions Precedent	10	
3	Intention of the parties		
	3.1 General intent	10	
	3.2 Specific intent	10	
4	Required Works		
	4.1 Carrying out of the Required Works	10	
	4.2 Acknowledgement	12	
	4.3 Indicative Program	13	
	4.4 Formation of Spring Farm SRG	13	
	4.5 Intellectual Property Rights in the Deliverables	13 13	
5	Payment and administration of Funding Amount		
	5.1 Milestone 1	13	
	5.2 Milestone 2	13	
	5.3 Security in lieu of payment for Milestone 2	14	
	5.4 Adjustment to Funding Amount Allocation	15 16	
6	Grant of access		
7	Dedicated Land		
	7.1 Identification of Additional Dedicated Land	16	
	7.2 Subdivision of or boundary adjustment for Additional Dedicated I		
	7.3 Transfer	17	
	7.4 Liability	20	
	7.5 Sale to Dahua 2	20 20	
8	Easement for Batter or Drainage		
	8.1 Grant of Works Easement	20	
	8.2 Support Works	21	
	8.3 Drainage Works	23 25	
9	Reporting by RMS		
	9.1 Bi-monthly status reporting for the SRG	25	
10	Authorised Representatives and Notices	25	
	10.1 Representatives	25	
	10.2 Notices	25	

11	Dispute Resolution		26
	11.1 Resolution of dis	putes	26
	11.2 Continue to perfo	orm obligations	27
12	Termination		27
	12.1 Default by Lando	om or the Dahua Group	27
	12.2 Termination of D	eed	27
	12.3 No release		28
13	GST		28
	13.1 Construction		28
	13.2 Consideration G	ST exclusive	28
	13.3 Payment of GST		29
	13.4 Timing of GST p		29
	13.5 Tax invoice	•	29
	13.6 Adjustment even	t	29
	13.7 Reimbursements	3	29
	13.8 No merger		29
14	General		30
	14.1 No fetter on state	utory functions and no limits on Liability	30
	14.2 Assignment and	novation	30
	14.3 Entire agreemen	t	30
	14.4 Costs		30
	14.5 Waiver		30
	14.6 Governing Law		30
	14.7 Modification of th	nis Deed	30
	14.8 Representations	and warranties	31
	14.9 Replacement of	Government Agency	31
	14.10 No Delay		31

Corrs Chambers Westgarth

15	Caveat	31
Execution Annexure A - Required Works Annexure B - Milestones and Contributions		33
		35
		36
Anr	nexure C – Property A Land, Property B Land and Dahua Land	39
Ann	nexure D - Dedicated Land	44
Anr	nexure E - Lease Land	45
Annexure F - SRG Terms of Reference		47
Anr	nexure G - Lease	53
Anr	Annexure H - Plan of Available Land and Easement Area	
Annexure I - Terms of Works Easement		66
Anr	nexure J - Catchment Plan	71

Date

Parties

Roads and Maritime Services ABN 76 236 371 088 of 20-44 Ennis Road Milsons Point NSW 2061 (RMS)

Landcom ABN 79 268 260 688 of Level 14, 60 Station Street, Parramatta NSW 2150 (Landcom)

Dahua Group Sydney Project 2 Pty Ltd (ABN 80 606 391 235) of Suite 20.02, Level 20, 201-217 Elizabeth Street, Sydney NSW 2000 (Dahua 2)

Dahua Group Sydney Project 3 Pty Ltd (ABN 49 606 391 922) of Suite 20.02, Level 20, 201-217 Elizabeth Street, Sydney NSW 2000 (Dahua 3)

Background

- A Landcom owns the Property B Land.
- B Dahua 2 and Dahua 3 respectively own the Property A Land and the Dahua Land.
- C Landcom and Dahua 2 have entered into a put and call option agreement for Landcom to sell, and for Dahua 2 to purchase, the Property B Land.
- D Landcom and the Dahua Group have agreed to contribute to the funding, procuring and carrying out of the Required Works.
- E To facilitate the delivery of the Required Works, Landcom and Dahua 3 will:
 - (a) pay, or pay and provide Security for, the Funding Amount to RMS upon satisfaction of the Milestones; and
 - (b) transfer the Dedicated Land to RMS,

in accordance with this Deed.

- F Initial funding for development of the Project to Final Business Case and delivery readiness will be made from the Housing Acceleration Fund Round 4 contribution of \$30 million committed to planning and construction of the Project.
- G The combined contributions from Landcom and Dahua 3 to the funding of the Required Works is capped at \$86 million and Housing Acceleration Fund Round 4 contributions to the funding of the Required Works is capped at \$30 million, unless otherwise agreed in writing between RMS, Landcom, Dahua 3 and the Department.
- H RMS will carry out the Required Works in accordance with this Deed.
- I The objective of the Required Works is to provide access to support the Development.

- J RMS will undertake project development and work within the INSW assurance gateway framework to confirm the specifications, project estimate and program for delivery of the Project.
- K While the Landcom design and cross-section is being used generally as a guide for the Required Works, both can be varied to facilitate delivery of the Project within the total amount of funding available.
- L The parties will agree an indicative program for the procurement and Final Completion of the Required Works.
- M RMS will use best endeavours to complete the Carriageway and as much of the other Required Works as possible by the Target End Date, up to the total amount of funding available for the Required Works and subject to the terms of this Deed.
- N Landcom's and the Dahua Group's contribution to the funding, procuring and carrying out of the Required Works and the value of the Dedicated Land to be transferred to RMS are provided as part of arrangements which are intended to enable the Secretary of the Department of Planning and Environment to issue Satisfactory Arrangements Certificates regarding the provision of Designated State Public Infrastructure under clause 6.1 of the Campbelltown Local Environmental Plan 2015 (NSW) for the Development.
- O The parties have otherwise agreed to the matters in this Deed.

Agreed terms

1 Interpretation

1.1 Definitions

In this Deed these terms have the following meanings:

Additional Dedicated Land means

- (a) those parts of the Available Land which is identified by RMS, acting reasonably; and
- (b) and any other land agreed between the parties in writing,

as set out in a Dedication Notice served in accordance with **clause 7.1(c)** as forming part of the Dedicated Land to be transferred to RMS under this Deed.

Approval means any consent, approval, authorisation, determination, licence, registration, order, permission or concurrence required by any Law required for the commencement, execution, Completion or use of the Required Works.

Authorised Representative means a person identified in clause 10.1(b) in respect of a party, or such replacement person as notified by a party in accordance with clause 10.1(c).

Available Land means land within the area bounded by a red line as shown on the map in **Annexure H**, as may be amended by agreement between the parties from time to time, including under **clause 7.1(b)**.

Business Day means a day on which banks are open for business generally in New South Wales, other than a Saturday, Sunday or public holiday in New South Wales and specifically excluding 27, 28, 29, 30 and 31 December.

Carriageway means a sealed road which delivers a minimum satisfactory level of public vehicular access through the length of the Spring Farm Parkway Stage One roadway to the Menangle Park Urban Release Area to enable development of the Menangle Park Urban Release Area.

Completion means that the Required Works have been completed to the relevant percentage as specified for each Milestone set out in Column 2 in **Annexure B**, as determined by RMS acting reasonably and evidenced by the status reports given by RMS under **clause 9.1**.

Conditions Precedent has the meaning set out in clause 2.1(b).

Contamination has the same meaning as in section 5 of the *Contaminated Land Management Act* 1997 (NSW), including hazardous materials such as asbestos in structures and soil.

Cost of the Required Works means the total cost incurred by RMS in carrying out the Required Works and the strategic planning for Stage 2 Spring Farm Parkway, including RMS' overhead costs for project management, and the cost of external consultants, in accordance with this Deed.

Council means Campbelltown City Council.

Dahua Group means Dahua 2 and Dahua 3.

Dahua Land means the land identified in section 3 of Annexure C.

Dedicated Land means the land identified in the table in Annexure D.

Dedication Notice has the meaning given to that expression in clause 7.1.

Deliverables means all design documentation, including final as-executed drawings, created for or in connection with the Required Works, and all drawings, sketches, specifications, calculations, digital records; data, studies, investigations, and other documents and information used in the design and construction of the Required Works.

Department means the Department of Planning & Environment.

Detailed Design Cost Estimate means an estimate of the Cost of the Required Works prepared on the basis of detailed design documentation for the Required Works and which incorporates a contingency which is reasonable and typical for detailed design estimates as determined by RMS having regard to projects of a similar nature and risk.

Developer means any party which is the registered proprietor of any part of the Property A Land, the Property B Land or the Dahua Land.

Development means the subdivision and development of the Land for residential, commercial and related purposes.

Development Consent has the same meaning given to that expression in the *Environmental Planning and Assessment Act* 1979.

Easement Area means any part of the Land burdened by one or more Works Easements indicatively as the land bounded by the green line on the plan at **Annexure H**.

Environmental Law means any law relating to the environment.

Environmental Notice means any direction, order, demand, notice, declaration or other requirement from any Government Agency (whether written or oral) under an Environmental Law.

Final Business Case means the description of the Required Works and anticipated cost estimate for the Required Works set out in the Project's final business case submitted to INSW for assurance review and NSW Government approval.

Final Completion means that the Required Works have been finally completed with no defects or omissions, except for defects not known, as notified to each of the Developers in the Final Completion Notice or, if no Final Completion Notice is given, as evidenced by the status reports given by RMS under **clause 9.1**.

Final Completion Notice means a notice issued under clause 4.1(e)(i).

Final Lot means any lot created in the Development for separate residential, retail or commercial occupation and disposition and which is not:

- (a) intended to be further subdivided (including to create a strata or community lot);
- (b) a Service Lot or a lot of a kind created for a purpose that is otherwise agreed by the Minister for Planning and the Developer;
- (c) to be dedicated or otherwise transferred to the Minister for Planning, or his or her nominated transferee;
- (d) a lot on which a dwelling house is situated that was in existence at the date of this Deed.

Funding Amount means the total amount of funding to be contributed by Landcom and the Dahua Group under this Deed, being the grand total set out in **Annexure B**, subject to any variation agreed between the parties in the course of Spring Farm SRG discussions and any notice from RMS pursuant to **clause 5.2(c)**.

Funding Amount Allocation means each allocation of the Funding Amount applicable to a Milestone, as set out in Column 3 of the table in Annexure B, subject to any variation agreed between the parties in the course of Spring Farm SRG discussions and any notice from RMS pursuant to clause 5.2(c).

Funding Amount End Notice means a notice issued under clause 4.1(e)(ii).

Funding Date means for each Funding Amount Allocation, the date set out in Column 2 of the table in Annexure B.

Funding Party means the party liable to pay the relevant Funding Amount Allocation as set out in Column 4 of the table in **Annexure B**.

Government Agency means a department of State, statutory or public authority, instrumentality, statutory corporation, body or person whether Commonwealth, State, territorial or local.

GST has the meaning as given to in clause 13.1.

HAF means the Housing Acceleration Fund.

HAF Funding means the contributions from the Housing Acceleration Fund Round 4 to the funding of the Required Works.

Indicative Program has the meaning given to it in clause 4.3.

INSW means Infrastructure NSW.

Intellectual Property Rights means all industrial and intellectual property rights, both in Australia and throughout the world, and includes any copyright, moral right, patent, registered or unregistered trade mark, registered or unregistered design, registered or unregistered plant breeder's right, trade secret, know-how, right in relation to semiconductors and circuit layouts, trade or business or company name, indication or source or appellation of origin or other proprietary right, or right of registration of those rights.

Just Terms Act means the Land Acquisition (Just Terms Compensation) Act 1991 (NSW).

Land means the Dahua Land, the Property A Land and the Property B Land.

Land Notification Date means the date that is 6 months after the date of execution of this Deed.

Law means:

- (a) any statute, regulation, rule, proclamation, order, ordinance or by-law whether present or future and whether Commonwealth, State, territorial or local;
- (b) the common law;
- (c) any Approval (including any condition or requirement under an Approval); and
- (d) any standards, criteria and guidelines made or approved by the NSW Environment Protection Authority in relation to Contamination or the Remediation of Contamination.

Lease means the lease of the Lease Land between the owner of the Lease Land (as lessor) and RMS (as lessee) for the purposes of carrying out the Required Works on the terms set out in **Annexure G**.

Lease Land means the land generally identified as bounded by the red line the plan in Annexure E, being land owned by Landcom, Dahua 2 or Dahua 3 and required by RMS to construct the Required Works, as varied from time to time pursuant to the Lease.

Liability means any damage, loss, loss of use, liability, cost, charge, expense (including legal costs), outgoing or payment suffered or incurred by the relevant party.

Menangle Park Urban Release Area means the Urban Release Area at Menangle Park, New South Wales.

Milestone means each milestone as set out in Column 1 of the table in Annexure B.

Personnel means the employees, agents, officers, contractors, subcontractors, suppliers, consultants and invitees of RMS, Landcom or the Dahua Group (as applicable).

Preliminary Assessment means the stage 1 preliminary site investigation assessment carried out on behalf of RMS by JBS&G Australia Pty Ltd dated 8 August 2017.

Preliminary Design Plans means indicative concept plans prepared by Jacobs copies of which are attached at Annexure A.

Project means the Stage One Spring Farm Parkway Project, which includes the strategic planning and delivery of the Required Works.

Property A Land means the land identified in section 1 of Annexure C.

Property B Land means the land identified in section 2 of Annexure C.

Relevant VPAs means one or more voluntary planning agreements between the Minister for Planning and Landcom or Dahua 2 or Dahua 3 or a combination of them, in relation to all (or substantially all) of the Land.

Remediation means any clean up, remediation or restoration measures relating to any Contamination with respect to the Dedicated Land, whether in existence prior to, on or after the date of this Deed, including any remediation required under any Environmental Notice or under any Law in relation to such Contamination.

Required Works means the activities and works set out in Annexure A – section 1 Description.

Security or Securities means:

- (a) for the Dahua Group an unconditional undertaking or certificate in a form acceptable to RMS and given by a bank or other financial institution regulated by the Australian Prudential Regulation Authority and acceptable to RMS;
- (b) for Landcom a letter of undertaking in a form acceptable to RMS and naming the Minister for Planning and the Department as beneficiaries.

Service Lot means a registered lot that is created for one or more of the following purposes:

- (a) to be dedicated or otherwise transferred to an Government Authority;
- (b) for any public utility undertaking (within the meaning of the Standard Instrument (Local Environmental Plans) Order 2006 as at the date of this Deed;
- (c) for roads, open space, recreation, environmental conservation, water cycle management or riparian land management;
- (d) for avoidance of doubt association property within the meaning of the *Community Land Development Act 1989* (NSW) that is to be used for any one or more of the purposes set out in (c) above.

Specifications means the specifications generally described in Section 2 of **Annexure A**.

Spring Farm SRG means the Spring Farm Stakeholder Reference Group set up pursuant to clause 4.4.

Sunset Date means 15 months after the date this Deed is executed.

Target End Date means:

- (a) if the Relevant VPAs are executed before 3 February 2020, the date that is:
 - (i) 33 months after the date on which Relevant VPAs have been executed by the parties to the Relevant VPAs; or
 - (ii) 48 months from the date this Deed is executed by the parties,

whichever is the latest; or

- (b) if the Relevant VPAs are executed on or after 3 February 2020, the date that is up to:
 - (iii) 36 months after the date on which Relevant VPAs have been executed by the parties to the Relevant VPAs; or
 - (iv) 51 months from the date this Deed is executed by the parties,

whichever is the latest.

Terms of Reference means the Spring Farm SRG terms of reference generally in the form set out in Annexure F.

Urban Release Area has the meaning given to that expression in the Campbelltown Local Environmental Plan 2015 (NSW).

Works Easement means an easement over Easement Area for the benefit of the Dedicated Land or an easement in gross for the benefit of one or more Government Agencies, which is required:

- to maintain the slope and stability of land adjoining the Required Works and the support provided to the Required Works (easement for batters); or
- (b) for the purposes of draining water (easement for drainage), or
- (c) for both (a) and (b) above,

on the terms set out in Annexure I subject to the drafting notes in Annexure I as applicable to the particular Works Easement in each case.

Works End Date means the earlier of the date on which:

- (a) Final Completion of the Required Works has occurred; or
- (b) all of the Funding Amount and HAF Funding have been spent on the Required Works.

1.2 Interpretation

In this Deed:

- (a) if any clause of this Deed is genuinely capable of different interpretations, the interpretation which reflects the intentions of the parties as set out in clause 3 is to be preferred;
- (b) clause headings are for convenience only and will be ignored in the interpretation of this Deed;
- (c) references to a party include the successors and permitted assigns of that party;
- (d) words importing the singular include the plural and words importing the plural include the singular;
- (e) words importing a person include a corporation, firm or body corporate;
- (f) nothing contained in this Deed will be deemed or construed as creating the relationship of partnership or joint venture of any kind;
- (g) references to a month mean a calendar month;
- (h) references to any document include any permitted amendment, supplement to or replacement or novation of this Deed;
- (i) references to any legislation or to any section or provision of any legislation includes any:
 - statutory modification or re-enactment of or any statutory provision substituted for that legislation, section or provision;
 - (ii) ordinances, by-laws, regulations and other statutory provision substituted for that legislation, section or provision;
- no waiver of any breach of this Deed or of any of its terms will be effective unless the waiver is in writing and signed by the party against whom the waiver is claimed, and no waiver of any breach will operate as a waiver of any other breach or subsequent breach;

- (k) other grammatical forms of defined words or expressions have corresponding meanings; and
- (I) 'including' and similar expressions are not words of limitation.

1.3 A party and its Personnel

If this Deed requires a party to do or not do something, that party must ensure that its Personnel do or not do that thing, as the case may be.

1.4 Order of precedence

In the event of any inconsistency between this Deed and a Relevant VPA, the Relevant VPA prevails to the extent of any inconsistency.

2 Operation of this Deed

2.1 Conditions Precedent

- (a) This Deed commences on the date of this Deed except for the following provisions, which will not commence unless and until the Conditions Precedent have been satisfied (or waived in accordance with clause 2.2):
 - (i) clause 4 (Required Works);
 - (ii) clause 5.2 (Milestone 2)
 - (iii) clause 5.3 (Security in lieu of payment for Milestone 2);
 - (iv) clause 5.4 (Reconciliation);
 - (v) clause 7.2 (Subdivision or Boundary Adjustment of Additional Dedicated Land);
 - (vi) clause 7.3(Transfer of Dedicated Land);
 - (vii) clause 8 (Easement for Batter or Drainage);
 - (viii) clause 9 (Reporting by RMS); and
 - (ix) Annexure B (Milestones and Contributions) (except in respect of Milestone 1).
- (b) The Conditions Precedent are:
 - the associated funding request from HAF for the HAF Funding for carrying out the Required Works be approved by the NSW Government; and
 - (ii) the Relevant VPAs are executed by the parties to the Relevant VPAs.
- (c) If the Conditions Precedent are not satisfied (or waived in accordance with clause 2.2) by the Sunset Date then:
 - (i) any party may terminate this Deed immediately on written notice to the other parties and the Department;

- (ii) within 20 Business Days of the termination of this Deed pursuant to clause 2.1(c)(i), RMS must repay to Landcom any part of the Milestone 1 Payment less the total of the following amounts:
 - (A) any amount which RMS has committed or spent; and
 - (B) any amount which is specified in the Relevant VPAs as an amount which may be retained for the purposes of this clause 2.1(c)(ii)(B).
- (iii) no party is entitled to make any claim against another arising out of or in connection with the termination of this Deed under clause 2.1(c)(i).

2.2 Waiver of Conditions Precedent

The parties agree that:

- (a) the Condition Precedent in clause 2.1(b)(i) is for the sole benefit of RMS and may only be waived in writing by RMS to Landcom and the Dahua Group (in its absolute and sole discretion); and
- (b) the Condition Precedent in **clause 2.1(b)(ii)** is for the benefit of Landcom and the Dahua Group jointly, and may only be waived in writing by Landcom and the Dahua Group jointly to RMS (in their absolute discretion).

3 Intention of the parties

3.1 General intent

The general intention of the parties in entering into this Deed is to formalise the agreement reached between the parties on the various matters referred to in the background section of this Deed.

3.2 Specific intent

The specific intent of the parties in entering into this Deed is (subject to the limitations and conditions set out in this Deed) to set out the process for:

- (a) the carrying out of the Required Works by RMS;
- (b) the payment of the Funding Amount by Landcom and Dahua 3 to RMS; and
- (c) the transfer of the Dedicated Land by Landcom and Dahua 2 to RMS,

in accordance with, and subject to, this Deed.

4 Required Works

4.1 Carrying out of the Required Works

(a) RMS must:

- use best endeavours to procure the carrying out of as much of the Required Works as possible (whilst prioritising delivery of the Carriageway) up to the total of the Funding Amount and the HAF Funding available for the Required Works in accordance with the Indicative Program; and
- (ii) in carrying out the Required Works, comply with all Approvals, Laws and requirements of all Government Agencies affecting the Required Works.
- (b) RMS is not required to expend more money on the Required Works than the total of the Funding Amount and the HAF Funding received by it for the Required Works and may elect, subject to Annexure F, to vary the specifications of the Required Works to ensure that the balance of the Required Works can be carried out for an amount equal to or less than the total of the Funding Amount and the HAF Funding received by RMS.
- (c) Without limiting the terms of Annexure F, RMS will consider any recommendations from Dahua 2 or Dahua 3 and Landcom on the concept design for the Required Works and any recommendations which are agreed by RMS prior to submission of the Final Business Case for NSW Government approval will be documented in the Final Business Case. Dahua 2, Dahua 3 and Landcom acknowledge that RMS may propose changes to the timing set out for the carrying out and Completion of the Required Works in the Indicative Program to take into account of delays, including delays beyond the control of RMS such as delays on the part of the Developers in paying the Funding Amount or in transferring the Dedicated Land to RMS, or to the commencement of the Relevant VPAs promptly after the commencement of this Deed.
- (d) At a minimum, and notwithstanding anything else in this Deed, the Carriageway will be delivered to ensure vehicular access via a public road or road to be dedicated as a public road to the Property A Land, the Property B Land and the Dahua Land for the Development.
- (e) RMS must issue to each of the Developers:
 - (i) a Final Completion Notice as soon as reasonably practicable after Final Completion of the Required Works has occurred; and
 - (ii) a Funding Amount End Notice as soon as reasonably practicable after all of the Funding Amount and the HAF Funding have been spent, if the Required Works have not been completed at that time and RMS is unable to complete the Required Works.
- (f) RMS must ensure that the land comprising the Carriageway, and as much of any other Required Works as is reasonably practicable, is dedicated as a public road under the *Roads Act 1993* as soon as practicable after the Target End Date or, if that is not practicable, then as soon as practicable after that date (having regard to the terms of this Deed).

4.2 Acknowledgement

The parties acknowledge and agree that:

- (a) RMS is solely responsible for, and takes all risks with respect to:
 - (i) obtaining all Approvals which are relevant in order to carry out the Required Works; and
 - (ii) the procurement, design, construction, security, safety and insurance of the Required Works;
- (b) RMS is not responsible for:
 - Landcom's and Dahua 3's obligations with respect to contributing to the funding of the Required Works in accordance with this Deed;
 - (ii) the cost of any additional works requested of RMS by Landcom or the Dahua Group; and
 - (iii) providing or obtaining any additional funding for the Required Works;
- the Required Works do not include the provision of noise walls and provision for noise walls has not been made in the Dedicated Land;
- (d) the Dahua Group and Landcom are solely responsible for, and take all risks with respect to:
 - (i) obtaining all Approvals which are relevant in order to construct any noise walls which may be required along the Carriageway; and
 - the procurement, design, construction, security, safety and insurance of any noise walls which may be required along the Carriageway;
- (e) in carrying out the Development, the Dahua Group and Landcom must prevent runoff (both piped and overland) from Catchment A and Catchment C as shown on the plan in Annexure J discharging onto the Carriageway via Spine Road for storm events up to 1% (1 in 100) annual exceedance probability in intensity;
- (f) the final concept design and the anticipated Cost of the Required Works (inclusive of a reasonable contingency) based on final concept design (not detailed design) will be set out in the Project's Final Business Case submitted to INSW for assurance review and NSW Government Approval; and
- (g) this Deed is not to be construed in any way as a contract for the provision of construction work in respect of the Required Works or otherwise. Notwithstanding that the parties agree that this Deed is not a construction contract, the parties agree that if this Deed is found to be a construction contract, it is a construction contract under which it is agreed that the consideration payable for construction work carried out under the contract is to be calculated otherwise than by reference to the

value of the work carried out or the value of the goods and services supplied.

4.3 Indicative Program

- (a) RMS agrees to prepare an indicative program:
 - (i) identifying the Target End Date for the opening of the Carriageway to traffic and completion of as much of the other Required Works as possible; and
 - (ii) which addresses the timing for the carrying out, completion and Final Completion of the Required Works (in particular the anticipated timing for each of the Milestones).

(the Indicative Program)

(b) If it appears unlikely that all of the Required Works will be Completed by the Target End Date, the parties must meet and negotiate in good faith to agree arrangements which will enable the Carriageway (as a priority) and as much of the other Required Works as possible to be completed before the Target End Date or, if that is not possible, as soon as possible after the Target End Date.

4.4 Formation of Spring Farm SRG

The parties agree to form a Spring Farm SRG to meet and discuss any issues arising out of this Deed, or during the course of the carrying out of the Required Works, in accordance with the Terms of Reference.

4.5 Intellectual Property Rights in the Deliverables

The parties acknowledge and agree that RMS owns all of the Intellectual Property Rights in the Deliverables.

5 Payment and administration of Funding Amount

5.1 Milestone 1

Landcom must pay the Funding Amount Allocation for Milestone 1 on the Funding Date.

5.2 Milestone 2

- (a) Landcom and Dahua 3 must pay the Funding Amount Allocations for Milestone 2 to RMS on or by the Funding Date in accordance with and subject to this clause 5.
- (b) Other than Milestone 1, Landcom and Dahua 3 have no obligation to pay any Funding Amount Allocation to RMS until RMS has complied with clause 5.2(c).
- (c) Each of Landcom and Dahua 3 will pay the relevant Funding Amount Allocation to RMS after each of the following has occurred:
 - (i) for each Milestone, RMS has served a written notice on Landcom and Dahua 3 requiring payment of the Funding Amount Allocation,

by a date that is no earlier than 8 weeks prior to the Funding Date and at least 4 weeks prior to the Funding Date unless otherwise agreed. RMS acknowledges that if it does not serve the required notice on Landcom and Dahua 3 at least 4 weeks prior to a Funding Date, an adjustment to the relevant Funding Date to provide a minimum of 4 weeks' notice will be made in order to facilitate Landcom and Dahua 3's compliance with the requirement for the payment of the relevant Funding Amount Allocation; and

- RMS has provided Landcom and Dahua 3 (as the case may be) with a valid tax invoice in respect of the relevant amount.
- (d) Landcom and Dahua 3 agree that the report provided by RMS under clause 9.1 is reasonable evidence of achievement, and anticipated achievement, of the relevant Milestone.

5.3 Security in lieu of payment for Milestone 2

- (a) Subject to clause 5.2, on or before the Funding Date in respect of Milestone 2, each of Landcom and Dahua 3 must either:
 - (i) pay to RMS the total Funding Amount Allocation applicable to Milestone 2; or
 - (ii) pay to RMS the Funding Amount Allocation applicable to Milestone 2a and provide Security in an amount equal to the total Funding Amount Allocation for Milestones 2b to 2d payable by the party in Column 5 of the table in Annexure B.
- (b) Each of Landcom and Dahua 3 may provide multiple Securities to RMS provided:
 - the amount of each Security reflects the Funding Amount Allocation payable for the relevant sub-milestone under Milestone 2;
 - (ii) the total amount of the Securities equals the total Funding Amount Allocation payable by that party for Milestones 2b to 2d; and
 - (iii) all Securities are provided by the Funding Date for Milestone 2.
- (c) Where clause 5.3(a)(ii) applies:
 - (i) Landcom and Dahua 3 must pay to RMS the Funding Amount Allocation for each of Milestones 2b to 2d by no later than the Funding Date for each of those Milestones; and
 - (ii) Within 20 Business Days of receipt of the whole of the Funding Amount Allocation for each of Milestone 2b to 2d, RMS must return the relevant Security for that Funding Amount Allocation to Landcom and Dahua 3, as the case may be. RMS must use all reasonable endeavours to return the relevant Security as soon as possible within the 20 Business Day period.

- (d) If Landcom or Dahua 3 breach any of their respective obligations to pay all or part of their Funding Amount Allocations under this Deed, RMS may immediately call on the Security or Securities provided by the relevant party without notice in order to obtain all or part of the Funding Amount Allocation which is due and payable at that time.
- (e) If Dahua 3 provides Security to RMS, Dahua 3 agrees that RMS may, at any time prior to RMS calling on the Security or the Security is returned to Dahua 3, require Dahua 3 to replace the Security if the credit rating of the bank or financial institution changes after the date the Security was issued.

5.4 Adjustment to Funding Amount Allocation

- (a) The parties acknowledge and agree that:
 - upon finalisation of the detailed design for the Required Works by RMS, RMS must prepare the Detailed Design Cost Estimate and provide it to Landcom, Dahua 3 and the Department; and
 - (ii) if the total amount identified in the Detailed Design Cost Estimate as the estimated cost of the Required Works is less than the grand total Funding Amount set out in Annexure B, then within 15 Business Days of finalisation of the detailed design for the Required Works, RMS must give notice to Landcom, Dahua 3 and the Department of the amount of any difference.
- (b) The notice in clause 5.4(a)(ii) must attach a table in the form of Annexure B, with an updated Column 4 setting out the updated Funding Amount Allocation amounts, which are adjusted, in the same proportions as those in Annexure B at the date of this Deed, so that the updated Funding Amount Allocation amounts are based on the Detailed Design Cost Estimate and, in terms of the Funding Amount Allocations where Landcom is the Funding Party, have regard to any payment which has been made by Landcom in respect of Milestone 1. For the avoidance of doubt, RMS may not make any other changes to the table without the prior written agreement of the Developers.
- (c) On service of the notice under clause 5.4(a), the table in Annexure B is deemed to be replaced by the updated table provided under clause 5.4(b), and Landcom and Dahua 3 must pay the Funding Amount Allocation in accordance with the updated table.
- (d) If there is any disagreement about the Funding Amount Allocation in the updated table, that dispute must be resolved in accordance with **clause** 11.

6 Grant of access

The owner of the Lease Land as at the date of this Deed grants to RMS a lease to access, use and occupy the Lease Land, for the purpose of carrying out the Required Works, on the terms set out **Annexure G**.

7 Dedicated Land

7.1 Identification of Additional Dedicated Land

- (a) The parties acknowledge and agree that:
 - (i) the Available Land shown in the plan at **Annexure H** of this Deed on the date of this Deed:
 - (A) does not include an allowance for noise walls; and
 - (B) assumes that the configuration and levels of the intersection of Spring Farm Parkway and Spine Road at the western end of the Carriageway will be in accordance with Preliminary Design Plans;
 - the Dedicated Land areas remain subject to detailed design of the Required Works;
 - (iii) as at the date of this Deed:
 - (A) Lots 2011 and 2012 in DP1234643 have already been created and specifically identified as Dedicated Land for the purpose of this Deed; and
 - (B) RMS has identified that some Additional Dedicated Land will likely be required to be transferred to RMS as part of the Dedicated Land under this Deed, to enable construction and ongoing operation and maintenance of the Required Works;
 - (iv) to avoid doubt, this Deed does not allow for the provision of Additional Dedicated Land outside the area of the Available Land, unless the parties agree otherwise in writing; and
 - (v) despite any other provision in this Deed, the parties will negotiate in good faith to minimise the area of land required for the Additional Dedicated Land.
- (b) RMS, Dahua 2 or Dahua 3 may, at any time prior to the Land Notification Date, ask the other parties in writing to consider moving the boundaries of the Available Land. If such a request is made, the parties will negotiate prior to the Land Notification Date in good faith in an endeavour to agree on moving the boundaries. If the parties agree, then RMS will prepare a new plan of Available Land to reflect the agreement and the Available Land will be taken to be as shown in that plan.
- (c) As soon as possible after the Land Notification Date and in accordance with any agreement between the parties under clause 7.1(b) as to what

land should be included as the Available Land, RMS must issue a notice (Dedication Notice):

- (i) identifying, acting reasonably, any area of Land required to be transferred to RMS as part of the Dedicated Land under this Deed, but such a notice may include land outside the Available Land only if agreed by the parties in writing;
- (ii) attaching a plan (including a plan in DWG format) delineating the boundaries of Lots 2011 and 2012 in DP1234643 adjusted to incorporate the Additional Dedicated Land; and
- (iii) attaching a plan (including a plan in DWG format) delineating the boundaries of the Additional Dedicated Land as one or more separate allotments,

with such notice to be issued to the:

- (iv) owner(s) of the Additional Dedicated Land identified in the notice (Relevant Landowner); and
- (v) Minister for Planning.
- (d) The parties acknowledge and agree that any delay to the transfer of the Dedicated Land to RMS pursuant to clause 7.3 may affect RMS' ability to open the Carriageway to traffic and complete as much of the Required Works as possible by the Target End Date.
- (e) A Dedication Notice is of no effect to the extent that it relates to Land that is not owned by the Relevant Landowner.

7.2 Subdivision of or boundary adjustment for Additional Dedicated Land

Before transferring the Additional Dedicated Land to RMS in accordance with clause 7.3, the Relevant Landowner must, at its cost:

- (a) obtain all Approvals necessary to create a separate lot or lots for the Additional Dedicated Land or effect a boundary adjustment so that the boundaries of the Dedicated Land are extended to incorporate the Additional Dedicated Land; and
- (b) in accordance with those Approvals, prepare and register a plan of subdivision to create a separate lot or lots for the Additional Dedicated Land, or effect a boundary adjustment so that the boundaries of the Dedicated Land are extended to incorporate the Additional Dedicated Land.

7.3 Transfer

- (a) The parties acknowledge and agree that:
 - (i) Landcom and Dahua 2 are the current owners of the Dedicated Land as set out in Annexure D;

- Landcom and Dahua 2 will, by the time set out in column B of Annexure D, effect the transfer of ownership of the Dedicated Land to RMS;
- (iii) the Dedicated Land will be transferred to RMS on an 'as is' basis, but, subject to clauses 7.3(a)(iv) and 7.3(a)(vii), any identification, assessment, management and Remediation carried out by or on behalf of RMS forms part of the Required Works and any costs incurred by RMS of such identification, assessment, management and Remediation will be paid for out of the total amount of funding available for the Required Works, including the Funding Amount and the HAF Funding;
- subject to clause 10 of the Lease, RMS will only accept responsibility for Contamination on the Dedicated Land, and Remediation of that Contamination, to the extent it is identified in the Preliminary Assessment;
- (v) RMS accepts responsibility for any Liability that arises in respect of the movement or handling of Contamination regardless of whether the Contamination is identified in the Preliminary Assessment, provided any Liability is limited to any claims that may be made for personal injuries, workplace health and safety or death;
- (vi) subject to compliance with Environmental Law, where Contamination is identified in the Preliminary Assessment, RMS will, where RMS determines, acting reasonably that it is appropriate to do so, utilise any Contaminated materials (including soils and building and demolition waste) on the land to the best extent possible and not dispose of any Contaminated materials offsite;
- (vii) subject to clause 7.3(a)(v), if any Contamination on the Dedicated Land is discovered that is not identified in the Preliminary Assessment, that additional Contamination and its Remediation, remains the responsibility of Landcom or Dahua 2 (as the case may be) and the following process will apply:
 - (A) upon the discovery of the relevant Contamination, RMS will promptly provide written notification to Landcom or Dahua 2 (as the case may be) of the Contamination, including provision of copies of any relevant contamination assessment reports / information;
 - (B) Landcom and Dahua 2 (as the case may be) may elect to undertake an independent assessment of the relevant Contamination (including for the purpose of seeking an independent expert opinion as to whether any relevant Contamination was or was not identified in the Preliminary Assessment), at its cost, provided the independent assessment is undertaken and the results are received by

Landcom and Dahua 2 (as the case may be) and provided to RMS within the earlier of 20 Business Days after the date on which RMS provided the written notification in **clause 7.3(a)(vii)(A)** or other timeframe imposed by a Government Agency or by Law for responding to the discovery of the relevant Contamination;

- (C) RMS will, acting reasonably, and in consultation with Council (if Council will be the roads authority for the Project after the Required Works are completed), determine the appropriate Remediation and management of the Contamination to be undertaken having regard to the requirements of the Law and any future liability which RMS may incur arising from the existence of the Contamination; and
- (D) subject to clause 7.3(a)(vii)(C), the standard of any Remediation of such Contamination should be only to make the affected Dedicated Land suitable for use as a road; and
- (viii) Landcom and Dahua 2 (as the case may be) will provide RMS with all necessary documents which are relevant to the transfer of ownership of the Dedicated Land to RMS.
- (b) If Landcom and Dahua 2 (as the case may be) fails to transfer the Dedicated Land, or any part of the Dedicated Land, in accordance with clause 7.3(a), the parties acknowledge and agree that RMS may determine, in its absolute discretion, to compulsorily acquire the Dedicated Land at any time.
- (c) If RMS determines to compulsorily acquire the Dedicated Land, or any part of the Dedicated Land, under clause 7.3(b), the parties agree:
 - (i) for the purposes of section 10A(3) of the Just Terms Act, to a shorter negotiation period of 5 Business Days; and
 - (ii) for the purposes of section 13(2) of the Just Terms Act, to a shorter proposed acquisition notice period of 5 Business Days.
- (d) If the Dedicated Land, or any part of the Dedicated Land is compulsorily acquired by RMS in accordance with **clause 7.3(b)** and **(c)**, then:
 - (i) the parties agree for the purpose of section 30 of the Just Terms Act that:
 - (A) Part 2 Division 1 and Part 3 of the Just Terms Act do not apply; and
 - (B) the total compensation payable by RMS for the compulsory acquisition is \$1 (excluding GST); and
 - (ii) if required by RMS, Landcom and Dahua 2 (as the case may be) will enter into a deed of release and indemnity in a form reasonably required by RMS; and

- (iii) Landcom and Dahua 2 (as the case may be) agrees to pay RMS' reasonable costs associated with the acquisition of the relevant land.
- (e) If, at any time after entry into this Deed and prior to the transfer of the Dedicated Land to RMS, Landcom proposes to transfer any part of the Property B Land to a third party (other than a transfer to Dahua 2), Landcom must:
 - (i) ensure that no part of the Dedicated Land is included in that transfer; and
 - (ii) no less than 20 Business Days prior to entry into any agreement to transfer the land, notify RMS of the proposed transfer.
- (f) As soon as reasonably practicable but no later than 10 Business Days following the transfer or compulsory acquisition of any part of the Dedicated Land (including the Additional Dedicated Land) to RMS in accordance with this Deed, RMS must issue a notice to Landcom or Dahua 2 (as the case may be) confirming compliance with this obligation.

7.4 Liability

On and from the date when the transfer of ownership of the Dedicated Land to RMS has been effected in accordance with **clause 7.3**, RMS acknowledges and agrees that:

- (a) subject to clause 7.4(b), RMS assumes all responsibility, risk and Liability for the Dedicated Land (including, without limitation, assuming all responsibility for compliance with any Laws or the requirement of any Government Agency (including any Approval), irrespective of whom the liabilities and requirements may be imposed on); and
- (b) subject to clause 10 of the Lease, and otherwise in accordance with clauses 7.3(a)(iii) to 7.3(a)(vii), RMS assumes all responsibility, risk and Liability for any:
 - (i) Contamination with respect to the Dedicated Land only to the extent disclosed in the Preliminary Assessment; and
 - (ii) Remediation.

7.5 Sale to Dahua 2

If the sale of the Dedicated Land to Dahua 2 occurs prior to the transfer of the Dedicated Land to RMS, Dahua 2 must comply with **clause 7.3** as though all references to Landcom were references to Dahua 2.

8 Easement for Batter or Drainage

8.1 Grant of Works Easement

(a) RMS must use its best endeavours to ensure that the Required Works are designed and constructed so that any batters and drainage infrastructure are wholly contained within the Dedicated Land.

- (b) The parties acknowledge that the grant of one or more Works Easements is required as a result of Landcom and the Dahua Group's request that RMS reduce the area of land required for the Additional Dedicated Land as much as practicable.
- (c) If RMS (acting reasonably) determines that it is not reasonably practicable to design and construct the Required Works so that any batters or drainage infrastructure are wholly contained within the Dedicated Land, RMS may, within a reasonable time prior to the issue of the Final Completion Notice, issue a notice to Dahua 2, Dahua 3 or Landcom requiring Dahua 2, Dahua 3 or Landcom (as the case may be) (Relevant Landowner) to grant RMS one or more Works Easements over only those parts of the Land which RMS determines are necessary to support the Required Works (in the case of an easement for batters) or provide drainage (in the case of an easement for drainage) and which are delineated in a survey plan attached to the notice (Easement Notice).
- (d) Subject to clause 8.1(e), the Relevant Landowner must, as soon as reasonably practicable after receiving an Easement Notice but prior to the issue of the Final Completion Notice, procure the grant of one or more Works Easements over that part of the Land which is the subject of the Easement Notice.
- (e) An Easement Notice is of no effect to the extent that it relates to land that is not owned by the recipient of the Easement Notice.

8.2 Support Works

- (a) If the Relevant Landowner wishes to carry out works within an Easement Area which the Relevant Landowner believes would enable the Works Easement for batters and any related Works Easement for drainage in that Easement Area to be extinguished (Support Works), the Relevant Landowner must provide to RMS:
 - (i) plans describing the Support Works and showing the finished surface levels proposed for the completion of the Support Works (Support Works Plans);
 - (ii) calculations for the Support Works;
 - (iii) plans which demonstrate how the batter supported by the Works Easement for batter and any related Works Easement for drainage will be maintained during the construction period for the Support Works;
 - (iv) documents which address any criteria which the Relevant Landowner and RMS have agreed under clause 8.2(i); and
 - (v) any other documents or information which RMS, acting reasonably and expeditiously requires for the purpose of considering whether it should provide its consent to the Support Works.

- (b) After receiving all documents and information described in clause 8.2(a), RMS must determine whether to grant consent to the Support Works Plans. RMS must not unreasonably withhold or delay its consent. If RMS does not grant consent to the Support Works Plans, RMS must give reasons to the Relevant Landowner.
- (c) Once RMS has granted its consent to the Support Works Plans:
 - (i) the Relevant Landowner must provide RMS with 10 Business Days' notice prior to commencing the Support Works;
 - (ii) if the Relevant Landowner commences the Support Works, the Relevant Landowner:
 - (A) must use reasonable endeavours to carry out and complete the Support Works as expeditiously as possible; and
 - (B) must carry out any Support Works in accordance with any Approval and the Support Works Plans (as may be amended from time to time in accordance with this clause 8.2(c));
 - (iii) the Relevant Landowner must not make any material change to the Support Works Plans without the prior consent of RMS, which must not be unreasonably withheld or delayed.
- (d) If the Relevant Landowner believes it has completed the Support Works, the Relevant Landowner must provide to RMS:
 - a certificate from a registered surveyor certifying that the Support Works have been completed and the finished levels of the Easement Area are as specified in the Support Works Plans;
 - (ii) a copy of the "as built" drawings for the Support Works, showing the finished levels of the Easement Area; and
 - (iii) any other documents or information which RMS, acting reasonably and expeditiously, requests in relation to the Support Works for the purpose of responding in accordance with **clause 8.2(e)**.
- (e) Within 30 Business Days of the receipt of all the documents and information referred to in clause 8.2(d), or such longer period as may be agreed between the parties, RMS must give written notice to the Relevant Landowner stating that either:
 - (i) the Works Easements described in **clause 8.2(a)** are no longer required and that the Works Easements are released; or
 - (ii) the Support Works have not been completed to the satisfaction of RMS.
- (f) RMS must not give a notice under clause 8.2(e)(ii) unless, acting reasonably, RMS is of the opinion that the Support Works have not been completed to the satisfaction of RMS.
- (g) If RMS gives a notice under clause 8.2(e)(ii), RMS must give reasons to the Relevant Landowner.

- (h) If RMS gives a notice under clause 8.2(e)(i), RMS must promptly do all things and execute all necessary documents, dealings or instruments and provide any further consents reasonably necessary in order to effect the release of the Works Easements described in clause 8.2(a) and their removal from title to the Easement Area.
- (i) The Relevant Landowner and RMS agree to discuss in good faith the development of criteria which, if satisfied, would be sufficient for RMS to grant consent under clause 8.2(b) or give a notice under clause 8.2(e)(i).
- (j) The parties agree that Support Works which satisfy RMS' Specification R44 Earthworks (as amended from time to time, or any other standard which replaces or supersedes that standard) would be sufficient for RMS to grant consent under clause 8.2(b) and give a notice under clause 8.2(e)(i).

8.3 Drainage Works

- (a) If the Relevant Landowner wishes to carry out works within an Easement Area which the Relevant Landowner believes would enable the Works Easement for drainage in that Easement Area to be extinguished (Drainage Works), the Relevant Landowner must provide to RMS:
 - (i) plans describing the Drainage Works and showing details of the Drainage Works (Drainage Works Plans); and
 - (ii) plans which demonstrate how the drainage supported by the Works Easement for drainage will be maintained during the construction period for the Drainage Works;
 - (iii) a report, including hydraulic/drainage calculations and modelling (using a DRAINS model) from a hydraulic engineer;
 - (iv) documents which address any criteria which the Relevant Landowner and RMS have agreed under clause 8.3(i); and
 - (v) any other documents or information which RMS, acting reasonably and expeditiously requires for the purpose of considering whether it should provide its consent to the Drainage Works.
- (b) After receiving all documents and information described in clause 8.3(a), RMS must determine whether to grant consent to the Drainage Works Plans. RMS must not unreasonably withhold or delay its consent. If RMS does not grant consent to the Drainage Works Plans, RMS must give reasons to the Relevant Landowner.
- (c) Once RMS has granted consent to Drainage Works Plans:
 - (i) the Relevant Landowner must provide RMS with 10 Business Days' notice prior to commencing the Drainage Works;
 - (ii) if the Relevant Landowner commences the Drainage Works, the Relevant Landowner:

- (A) must use reasonable endeavours to carry out and complete the Drainage Works as expeditiously as possible; and
- (B) must carry out any Drainage Works in accordance with the Drainage Works Plans (as may be amended from time to time in accordance with this clause 8.3(c));
- (iii) the Relevant Landowner must not make any material change to the Drainage Works Plans without the prior consent of RMS which must not be unreasonably withheld or delayed.
- (d) If the Relevant Landowner believes it has completed the Drainage Works, the Relevant Landowner must provide to RMS:
 - (i) a certificate from a hydraulic engineer certifying that the Drainage Works have been completed;
 - a certificate from a registered surveyor certifying that the finished levels of the Easement Area are as specified in the Support Works Plans;
 - (iii) a copy of the "as built" drawings for the Drainage Works, showing the finished levels of the Easement Area; and
 - (iv) any other documents or information which RMS, acting reasonably and expeditiously, requests in relation to the Drainage Works for the purpose of responding in accordance with **clause 8.3(e)**.
- (e) Within 30 Business Days of the receipt of all the documents and information referred to in clause 8.3(d), or such longer period as may be agreed between the parties, RMS must give written notice to the Relevant Landowner stating that either:
 - (i) the Works Easement described in **clause 8.3(a)** is no longer required and that the Works Easement is released; or
 - (ii) the Drainage Works have not been completed to the satisfaction of RMS.
- (f) RMS must not give a notice under clause 8.3(e)(ii)unless, acting reasonably, RMS is of the opinion that the Support Works have not been completed to the satisfaction of RMS.
- (g) If RMS gives a notice under clause 8.3(e)(ii), RMS must give reasons to the Relevant Landowner.
- (h) If RMS gives a notice under clause 8.3(e)(i), RMS must promptly do all things and execute all necessary documents, dealings or instruments and provide any further consents reasonably necessary in order to effect the release of the Works Easement described in clause 8.3(a) and its removal from title to the Easement Area.
- (i) The Relevant Landowner and RMS agree to discuss in good faith the development of criteria which, if satisfied, would be sufficient for RMS to grant consent under clause 8.3(c)or give a notice under clause 8.3(e)(i).

9 Reporting by RMS

9.1 Bi-monthly status reporting for the SRG

RMS must, within 15 Business Days of the end of each two month period (being a two month period ending on the last day of each even numbered month each calendar year), provide to the other parties a report updating the other parties on the following matters in respect of the preceding two month period:

- the progress of the Required Works and Milestones achieved in the prior period;
- (b) major changes to the Required Works with the potential to substantially impact the cost and specifications of the Required Works;
- (c) the proportion of the Funding Amount and the HAF Funding expended in the prior period; and
- (d) the anticipated date for:
 - (i) achieving Final Completion;
 - (ii) expending the remaining funds from the total of the Funding Amount and the HAF Funding; and
 - (iii) achieving any remaining Milestones,

together with supporting information relating to these matters.

10 Authorised Representatives and Notices

10.1 Representatives

- (a) The Authorised Representative of a party may perform any function of that party under this Deed.
- (b) The Authorised Representatives of the parties are, as at the date of this Deed, the following:

RMS:	Colin Langford, Director, North West Precinct
Landcom:	Matthew Beggs, Executive General Manager, Projects

The Dahua Group: Kith Clark, Development Director

(c) A party may change its Authorised Representative by notice in writing to the other party at any time.

10.2 Notices

(a) A party notifying or giving notice under this Deed must do so in writing sent by prepaid registered post and the original by post to the Authorised Representative at the address specified below:

RMS

Address: Level 7, 27 Argyle Street, Parramatta NSW 2150

Telephone: 02 8849 2339

Email: Colin.Langford@rms.nsw.gov.au

Attention: Colin Langford

Landcom

Address: Level 14, 60 Station Street Parramatta NSW 2150

Telephone: 02 9841 8759

Email: mbeggs@landcom.nsw.gov.au

Attention: Matthew Beggs

The Dahua Group

Address: Suite 2, Level 20, 201 Elizabeth Street, Sydney NSW 2000

Telephone: 02 9267 7788

Email: kith.clark@dahuaaustralia.com.au

Attention: Kith Clark

- (b) A notice given in accordance with clause 10.2(a) will be deemed to have been given and received:
 - (i) if delivered, on receipt;
 - (ii) if posted, 3 Business Days after posting; and
 - (iii) in the case of email, if the sender of an email can provide evidence of the dispatch of the email and did not receive notice of the rejection or failure to deliver the email, it shall be deemed to have been given:
 - (A) if transmission was successfully completed before 4.00 pm on a Business Day, on the day it was sent; and
 - (B) otherwise on the next Business Day.
- (c) Any notice received after 5.00 pm or on a day not a Business Day shall be deemed to have been received at 9.00 am on the next Business Day.

11 Dispute Resolution

11.1 Resolution of disputes

- (a) If a party claims that a dispute has arisen under this Deed, it must give written notice to the other parties stating the matters in dispute.
- (b) As soon as practicable after receiving notice of the dispute, the parties' Authorised Representatives, or a suitable nominated representative, must meet and attempt to resolve the dispute within a timeframe agreed

by the parties, or, in the absence of agreement, within 15 Business Days from the delivery of the notice of the dispute.

- (c) If the dispute is not resolved, or an appropriate path for resolution is not agreed, by the Authorised Representatives of the parties within the timeframe set out in **clause 11.1(b)**, the dispute may be referred to the parties' senior management who must meet and attempt to resolve the dispute within a timeframe agreed by the parties, or, in the absence of agreement, within 20 Business Days from expiry of the relevant period in **clause 11.1(a)**.
- (d) If the dispute is not resolved within the timeframes set out in clauses 11.1(b) and 11.1(c) above, then the chief executive officers of the parties, or a suitably nominated representative, must meet and seek to resolve the dispute within a timeframe agreed by the parties, or, in the absence of agreement within 30 Business Days of the expiry of the 20 Business Day period referred to in clause 11.1(c).

11.2 Continue to perform obligations

Each party must continue to perform its obligations under this Deed, notwithstanding the existence of a dispute.

12 Termination

12.1 Default by Landcom or the Dahua Group

If any or all of Landcom, Dahua 2 and Dahua 3:

- (a) fails to pay or provide Security in accordance with clause 5;
- (b) is presumed to be insolvent under an applicable law, including any presumption under section 459C(2) of the *Corporations Act 2001* (Cth), or has proposed or made an arrangement or composition with any or all of its creditors, or application is made to a court for an or order, or a resolution is made, for its winding up or which may have the object of or result in its winding up;
- (c) has a judgment debt against it outstanding for more than 7 days;
- (d) commits any material breach of the Deed, or
- has a receiver or liquidator (including provisional) or any official manager, controller or administrator of any of its assets or business appointed;

and fails to remedy such event within 10 Business Days of a written request by RMS to do so, then RMS may, in its absolute discretion and without prejudice to its other rights, by notice in writing to the other parties, terminate the Deed as from the date of the notice without prejudice to its accrued rights under this Deed.

12.2 Termination of Deed

On termination of this Deed under clause 12.1,

- (a) all or any sums of money which may be in the hands of RMS in respect of this Deed; and
- (b) all Security which has been provided to RMS under clause 5 of this Deed,

which

- (c) RMS has committed or spent for the purposes of the Project; or
- (d) is specified in the Relevant VPAs as an amount which is forfeited for the purposes of this clause 12.2,

may be declared by RMS to be forfeited and all sums that are so declared to be forfeited shall be forfeited and shall be retained by or become payable to or vested in RMS, and in respect of the sums retained under **clause 12.2(c)** or **clause 12.2(d)**, RMS shall use such money and Security, and take appropriate steps to administer relevant contracts, for the completion of the Project.

12.3 No release

Termination by RMS will not release any party from Liability in respect of any breach of, or non-performance of any obligation pursuant to this Deed.

13 GST

13.1 Construction

In this clause 13:

- (a) unless there is a contrary indication, words and expressions which are not defined in this document but which have a defined meaning in the GST Law have the same meaning as in the GST Law;
- (b) GST Law has the same meaning given to that expression in the A New Tax System (Goods and Services Tax) Act 1999 (Cth) or, if that Act does not exist for any reason, means any Act imposing or relating to the imposition or administration of a goods and services tax in Australia and any regulation made under that Act; and
- (c) references to GST payable and input tax credit entitlements include:
 - notional GST payable by, and notional input tax credit entitlements of the Commonwealth, a State or a Territory (including a government, government body, authority, agency or instrumentality of the Commonwealth, a State or a Territory); and
 - (ii) GST payable by, and the input tax credit entitlements of, the representative member of a GST group of which the entity is a member.

13.2 Consideration GST exclusive

Unless otherwise expressly stated, all consideration, whether monetary or nonmonetary, payable or to be provided under or in connection with this Deed is exclusive of GST (**GST-exclusive consideration**).

13.3 Payment of GST

If GST is payable on any supply made by:

- (a) a party; or
- (b) an entity that is taken under the GST Law to make the supply by reason of the capacity in which a party acts,

(Supplier) under or in connection with this document, the recipient of the supply, or the party providing the consideration for the supply, must pay to the Supplier an amount equal to the GST payable on the supply.

13.4 Timing of GST payment

The amount referred to in **clause 13.3** must be paid in addition to and at the same time and in the same manner (without any set-off or deduction) that the GST-exclusive consideration for the supply is payable or to be provided.

13.5 Tax invoice

The Supplier must deliver a tax invoice or an adjustment note to the recipient of a taxable supply before the Supplier is entitled to payment of an amount under clause 13.3.

13.6 Adjustment event

If an adjustment event arises in respect of a supply made by a Supplier under or in connection with this document, any amount that is payable under **clause 13.3** will be calculated or recalculated to reflect the adjustment event and a payment will be made by the recipient to the Supplier or by the Supplier to the recipient as the case requires.

13.7 Reimbursements

- (a) Where a party is required under or in connection with this document to pay for, reimburse or contribute to any expense, loss, liability or outgoing suffered or incurred by another party or indemnify another party in relation to such an expense, loss, liability or outgoing (Reimbursable Expense), the amount required to be paid, reimbursed or contributed by the first party will be reduced by the amount of any input tax credits to which the other party is entitled in respect of the Reimbursable Expense.
- (b) This clause 13.7 does not limit the application of clause 13.3, if appropriate, to the Reimbursable Expense as reduced in accordance with clause 13.7(a).

13.8 No merger

This clause **13** does not merge on the completion, rescission or other termination of this document or on the transfer of any property supplied under this document.

14 General

14.1 No fetter on statutory functions and no limits on Liability

- (a) To the extent applicable, nothing in this Deed is deemed to in any way to restrict or limit the powers of Landcom or RMS or to comprise a fetter of the exercise by Landcom or RMS of their statutory functions.
- (b) Nothing in this Deed limits any Liability which Landcom, the Dahua Group or RMS (as applicable) may have to Landcom, the Dahua Group or RMS (as applicable) under this Deed as a result of a breach by Landcom, the Dahua Group or RMS (as applicable) of a term of this Deed.

14.2 Assignment and novation

Other than as provided for under **clause 14.9**, the parties must not assign, novate or encumber any right, obligation or interest under this Deed, without the prior written consent of the other parties.

14.3 Entire agreement

- (a) This Deed and any subsequent voluntary planning agreement(s) under s 7.4 of the Act supersede all previous agreements about its subject matter and any agreements collateral to those agreements. This Deed embodies the entire agreement between the parties in relation to its subject matter.
- (b) To the extent permitted by law, any statement, representation or promise made in any negotiation or discussion, has no effect except to the extent expressly set out or incorporated by reference in this Deed.
- (c) Each party acknowledges and agrees that it does not rely on any prior conduct or representation by the other party in entering into this Deed.

14.4 Costs

Each party bears its own costs in relation to the preparation, negotiation and finalisation of this Deed (including any transactions or documents contemplated under this Deed).

14.5 Waiver

Failure by a party to compel performance of any term of this Deed does not constitute a waiver of that term or condition and does not impair the right of the party to enforce it at a later time or to pursue remedies it may have for any subsequent breach of that term or condition.

14.6 Governing Law

This Deed is governed by New South Wales Law and each party irrevocably submits to the non-exclusive jurisdiction of the New South Wales courts and courts competent to hear appeals from those courts.

14.7 Modification of this Deed

No modification or alteration of any provision of this Deed will be valid unless it is in writing and signed by the parties to this Deed.

14.8 Representations and warranties

Each party represents and warrants as at the date of this Deed that:

- (a) it has the power to enter into this Deed and to comply with its obligations under this Deed;
- (b) it has in full force and effect the authorisations necessary for it to enter into this Deed and to comply with its obligations under this Deed and to allow them to be enforced;
- (c) its obligations under this Deed are valid and binding and are enforceable against it in accordance with the terms of this Deed; and
- (d) this Deed and the transactions under this Deed which involve it do not contravene any Law or obligation by which it is bound.

14.9 Replacement of Government Agency

If a Government Agency referred to in this Deed:

- (a) is reconstituted, renamed or replaced or if its powers or functions are transferred to another Government Agency, this Deed is deemed to refer to that Government Agency; or
- (b) ceases to exist, this Deed is deemed to refer to that Government Agency which serves substantially the same purpose or object as the former Government Agency.

14.10 No Delay

Each party must not unreasonably delay any document, action, approval, direction, determination or decision which is required under this Deed.

15 Caveat

- (a) The parties agree that upon execution of this Deed by the parties, RMS has a caveatable interest in:
 - (i) Lots 2011 and 2012 in DP1234643;
 - (ii) the Available Land; and
 - (iii) land which is up to 15 metres from the boundary of the Available Land (measured in all directions from the boundary of the Available Land),

(together, the Caveat Land)

until the earlier of:

- (iv) the date the Final Completion Notice is issued; or
- (v) the date the Funding Amount End Notice is issued to each of the Developers under clause 4.1(e); or
- (vi) the termination of this Deed,
and that Landcom and the Dahua Group will not do anything or omit to do anything that will impede those rights.

- (b) The Dahua Group and Landcom agree that:
 - the Dahua Group and Landcom will not object to RMS lodging with NSW Land Registry Services a caveat for registration on the relevant folios of the Register (or parts thereof) of the Caveat Land, nor will the Dahua Group and Landcom seek to remove that caveat lodged; and
 - (ii) the Dahua Group and Landcom will promptly and use all reasonable endeavours to obtain consent to the lodgement of the caveat from each person who has an estate or interest in the Caveat Land registered under the *Real Property Act 1990* (NSW) to facilitate registration of the caveat.
- (c) RMS agrees that any caveat registered on the folios of the Register (or parts thereof) of the Caveat Land under this clause 15 must not prohibit the transfer of the Property B Land from Landcom to Dahua 2.
- (d) If a caveat is registered on the relevant folios of the Register for the Caveat Land under this **clause 15**, RMS agrees to:
 - (i) promptly consent to any dealing or plan which relates to the Caveat Land which does not adversely affect RMS' interest in the Caveat Land, including any proposed subdivision of all or part of the Caveat Land to create a separate lot or lots for the Additional Dedicated Land and the creation of any security interest which might be proposed in connection with the lending of money to a member of the Dahua Group; and
 - (ii) upon the occurrence of the relevant event in clause 15(a)(iv), (v) or (vi) and the lapsing of RMS' caveatable interest, use reasonable endeavours to withdraw the caveat from the title to the Caveat Land within 10 Business Days of the relevant event occurring.

Corrs Chambers Westgarth

Execution

Executed as a deed.

Executed by Roads and Maritime Services by its duly appointed officer in the presence of:

)

)

)

)

.

Witness

Officer

Name of Witness (print)

Signed, sealed and delivered by) Landcom trading as UrbanGrowth NSW) by the party's attorney pursuant to power) of attorney registered Book 4716 No 510) who states that no notice of revocation of) the power of attorney has been received in) the presence of:) Name of Officer (print)



Attorney

Name of Attorney (print)

Executed by Dahua Group Sydney Project 2 Ltd

-Gompany Secretary/Director

Name of Witness (print)

Witness

Name of Gompany Secretary/Director (print)



3437-2025-8564v54 L\329906431.1 ME_152649462_1

page 33

Corrs Chambers Westgarth

Executed by Dahua Group Sydney Project 3 Pty Ltd Gempany Secretary/Director L For Name of Company Secretary/Director (print)

))

\$3,''S

Director



.

Annexure A

Required Works

1 Description

Construction of works as determined by RMS in its sole and absolute discretion in accordance with this Deed to provide safe and efficient access to the Menangle Park development, as generally indicated in the plans by Jacobs attached in this **Annexure A** of this Deed and attached, subject to refinement of the specifications by RMS in accordance with this Deed, including:

- (a) the Carriageway;
- (b) Hume Motorway Bridge;
- (c) Hume Motorway northern on and off ramps;
- (d) Menangle Rd tie in; and
- (e) any ancillary works.

The specifications and scope of the Required Works to be confirmed by RMS in accordance with this Deed in order to deliver the Required Works within the amount that is the total of the Funding Amount and the HAF Funding.

Further refinement to the design and construction staging as required once the tender price for construction is received and during construction.

2 Specifications

The specifications applicable to the Required Works are as follows, subject to refinement and confirmation in the Final Business Case:

- (a) all applicable Austroads, RMS/TfNSW and other relevant standards and specifications, including the proposed cross-section being used for the design of the Required Works; and
- (b) any other requirements RMS considers acting reasonably are relevant to the delivery of the Required Works.

3 Preliminary Design Plans

See attached.



0 3 10 13 24		1 12 14	/ 19	1.000	10165				-																																		
	CHANNES FILE LOCATION HANKS	RC-0107	11	1.1	5.1		RD-0271	RD-0261			RD-0251 RD-0252		RD-0244	RD-0242	RD-0241	RD-0223	RD-0231					RD-0214	RD-0212	RC-0211	RD-0204	RD 1202	RD-0201	RD-0112	1	RD-0109	RD-0108		i I	RD-0103	RD-0101	ROAD ALIGNMENT A	GE-0034	GE-0032	GE-021	GE-0011	GE-0002	GENERAL	SHEET No.
	PLCT UNIT	GROSS SECTIONS- SPRING FARM PARKWAY WAR (MC10) - SHEET 1 OF 17 GROSS SECTIONS- SPRING FARM PARKWAY WAR (MC10) - SHEET 8 OF 17	ORDSS SECTIONS - SP	CROSS SECTIONS - SP	CROSS SECTIONS - SPI	CROSS SECTIONS - SPI	LONGITUDINAL SECTION - SUBDIVISION ACCESS ROAD (MCF0)	LONGITUDINAL SECTION - SUBDIVISION ACCESS ROAD (MCEO)	LONGITUDINAL SECTIO	LONGITUDINAL SECTIO	LONGITUDINAL SECTION - MENANGLE ROAD NORTHBOUND (MCB0) - SHEET 1 OF 5 LONGITUDINAL SECTION - MENANGLE ROAD NORTHBOUND (MCB0) - SHEET 2 OF 5	LOSAN) UDBAN, SECTION - GENARACE KOAD SOUTHBOUND (MCAN) - SHEET 3 OF 3	LONGITUDINAL SECTIO	LONGTUDINAL SECTION - MEMARGLE ROAD SOUTHEOUND (MCAD) - SHEET 2 OF 5 LONGTUDINAL SECTION - MEMARGLE ROAD SOUTHEOUND (MCAD) - SHEET 3 DE 5	LONGITUDINAL SECTIO	LONGITUDINAL SECTION - SOUTHBOUND EXIT RAMP (MC30) - SHEET 2 OF 3 LONGITUDINAL SECTION - SOUTHBOUND EXIT RAMP (MC30) - SHEET 3 OF 3	LONGITUDINAL SECTIO	LONGITIJDINAL SECTIO	LONGITUDINAL SECTIO	LONGTUDINAL SECTION - NORTHBOUND ENTRY RAMP (AIC40) - SHEET 2 OF 5 LONGTUDINAL SECTION - NORTHBOUND ENTRY RAMP (AIC40) - SHEET 3 OF 5	LONGITUDINAL SECTIO	LONCTUDINAL SECTIO	LONGTINDINAL SECTION - SPRING FARM PARKWAY EB (MC20) - SHEET 2 OF A LONGTINDINAL SECTION - SPRING FARM PARKWAY EB (MC20) - SHEET 3 OF 4	LONGITUDINAL SECTIO	LONGTUDINAL SECTION - SPRING FARM PARKWAY WIS (MCTIV) - SHEET 3 OF 4	LONGITUDINAL SECTION	LONGITUDINAL SECTIO	GENERAL ARRANGEMENT PLAN - SHEET 13 OF 13 GENERAL ARRANGEMENT PLAN - SHEET 13 OF 13	GENERAL ARRANGEME	GENERAL ARRANGENE	GENERAL ARRANGEME	GENERAL ARRANGEME	GENERAL ARRANGEMENT PLAN - SHEET 4 OF 13 GENERAL ARRANGEMENT PLAN - SHEET 5 OF 13	GENERAL ARRANGEME	GENERAL ARRANCEME	AND EARTHMORKS	TYPICAL CROSS SECTION - SHEET 4 OF 4	TYPICAL CROSS SECTION - SHEET 2 OF 4	KEY PLAN	GENERAL NOTES	INDEX SHEET - SHEET 1 OF 2		SHEET TITLE
01 01 02 02 02 02 02 02 02 02 02 02 02 02 02		RING FARM PARKWAY	UNG FARM PARKWAY	UNG FARM PARKWAY	UNG FARM PARKWAY	UNG FARM PARKWAY	N - SUBDIVISION ACCE	N - SUBDIVISION ACCE	N- MENANGLE ROAD	N - MENANGLE ROAD	N - MENANGLE ROAD	N - MERVANGLE KUAD	N - MENANCLE ROAD	N - MENANGLE ROAD	N - MENANGLE ROAD	N - SOUTHBOUND EXT	N - SOUTHBOUND EXT	N - NORTHBOUND ENT	N - NORTHBOUND ENT	N - NORTHBOUND ENT	N - NORTHBOUND ENT	N - SPRING FARM PAR	N - SPRING FARM PAR	N- SPRING FARM PAR	N - SPRING FARM PAR	N - SPRING FARM PAR	N - SPRING FARM PAR	NT PLAN - SHEET 13 C	NT PLAN - SHEET 11 C	NT PLAN - SHEET 9 OF	NT PLAN - SHEET 7 OF	NT PLAN - SHEET 6 OF	NT PLAN - SHEET 4 OF	NT FLAN - SHEET 3 OF	NT PUAN - SHEET 1 OF		JN-SHEET 4 OF 4	DN-SHEET 2 OF 4	DN-SHEET OF 4	UF 2	IOF 2	ran a maka ta sa sa dan mar "A A mangapat ng mangapat ng	
ISSUE RA SIN CONCEPT LEOD ISSUE RA SIN CONCEPT LEOD ISSUE RA SIN CONCEPT LEOD	\$10701825457512 -	WIB (MC10) - SHEET	W/B (MC10) - SHEET	W/B (MC10) - SHEET	WARACTO - SHEET	WIR IMCIOL - SHEET	ESS ROAD (MCF0)	ESS ROAD (MCEO)	NORTHBOUND (MCBC	NORTHBOUND (MCBC	NORTHBOUND (MCBI	SOUTHBOOND (NICA	SOUTHBOUND (MICAL	SOUTHBOUND (MCAU	SCUTHBOUND (MCAC	T RAMP (MC50) - SHE	T RAMP (MC50) - SHE	TRY RAMP (MC40) - SI	TRY RAMP (MC40) - SI	TRY RAMP (MC40) - SI	TRY RAMP (MC40) - SI	URWAY EVE (MC20) - S	KWAY EB (MC20) - S	NOWAY EIB (MC20) - S	KWAY WB (MC10) - S	KWAY W/B (MC10) - S	KWAY WB (MC10) - S	4 4 3		3 3	13	រដ 	a a	ವ	5								
	รรษกมด	7 OF 17 B OF 17	6 0F 17		2 OF 17	105 17)) - SHSET 5 OF 5)- SHEET 3 OF 5)) - SHEET 1 OF 5)) - SHEET 2 OF 5	I) - SHEEL S UP S)- SHEET 4 OF 5)- SHEET 2 OF 5	I)- SHEET 1 OF 5	ET 3 OF 3	E10F3	HEET 5 OF 5	HEET 4 OF 5	HEET 2 OF 5	HET 1 OF 5	HEET & OF 4	HEET 2 OF 4	HEET T OF 4	HEET 4 OF 4	HEET 2 OF 4	HEET 1 OF 4		and the second s					Construction of the second sec	and an and a set of the set of th								
0 VINTACIAL VINTACIAL																															-												
ICOORDINATE STOTEN	ROJECT BREAKLOWN STRU		RC-0195	RC-0101			RC-0184	RC-0182		- i - i	- A - G	RC-0175	4 C.	1.1	RC-0171	RCOTE2				- i i		- E	RC-2151	RC-0141	RC-0140	- C.		RC-0135			RC-0129	÷ +		RC-0124	4	RC-0121		1.1				RC-0109	SHEET No.
HEart Datus	STORE		CROSS SECTIONS - SUBDIVISION ACCESS ROAD (MCF0)	CROSS SECTIONS - SUBDIVISION ACCESS ROAD (MCEO)	CROSS SECTIONS - MENANGLE ROAD NORTHBOUND	CROSS SECTIONS - I	CROSS SECTIONS - N CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - N	GROSS SECTIONS - MENANGLE ROAD NORTHBOUND (MC80) - SHEET 7 OF 18 GROSS SECTIONS - MENANGLE ROAD NORTHBOUND (MC80) - SHEET 8 OF 18	CROSS SECTIONS - N	CROSS SECTIONS - I	CROSS SECTIONS - N	CROSS SECTIONS - 1	CROSS SECTIONS - SCOTHBOUND EXIT RAMP (MCSO) - SHEET 12 OF 13 CROSS SECTIONS - SOUTHBOUND EXIT RAMP (MCSO) - SHEET 13 OF 13	CROSS SECTIONS - S	CROSS SECTIONS - 9	CROSS SECTIONS - S	CROSS SECTIONS - S	CROSS SECTIONS - S	CROSS SECTIONS - 9	GROSS SECTIONS - SOUTHBOUND EXIT RAMP (MCS0) - SHEET T OF 13 GROSS SECTIONS - SOUTHBOUND EXIT RAMP (MCS0) - SHEET 2 OF 13	CRCSS SECTIONS - N	CROSS SECTIONS - P	CROSS SECTIONS - N	CROSS SECTIONS - NORTHOUND ENTRY RAMP (MC40) - SHEET 16 OF 21 CROSS SECTIONS - NORTHOUND ENTRY RAMP (MC40) - SHEET 17 OF 21	CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - M	CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - N	CROSS SECTIONS - N	CRUSS SECTIONS - 2	CROSS SECTIONS - S	CROSS SECTIONS - S	CROSS SECTIONS - S	CRUSS SECTIONS - SPRING FARM FRAMMER WE (M. U) - SPEET 11 OF 11 CRUSS SECTIONS - SPRING FARM FRAMMER (M. (U) - SPEET 11 OF 17	CROSS SECTIONS - S	SHEET TITLE
Hard Contraction of the second	JACORS DOCUMENT NUMEOR 1A185400-DC-GE-C002		UBDIVISION ACCESS	USDIVISION ACCESS	JENANGLE ROAD NO	RENANCI E ROAD NO	AENANGLE ROAD NO	MENANGLE ROAD NO	AEMANGLE ROAD NO	TENANGLE ROAD NO	RENANGLE ROAD NO	AENANGLE ROAD NO	JENANGLE ROAD NO	VENANCI E DOAD NO	TENANGLE ROAD NO	SOUTHBOUND EXIT R	OUTHBOUND SKT 9	SOUTHBOUND EXIT R	SOUTHBOUND EXIT R	SOUTHBOUND EXIT R	SOUTHBOUND EXIT R	OUTHBOUND EXT R	OUTHBOUND EXIT R	IORTHBOUND ENTRY	IORTHBOUND ENTRY	ORTHBOUND ENTRY	ORTHBOUND ENTRY	IORTHBOUND ENTRY	IORTHBOUND ENTRY	IORTHBOUND ENTRY	IORTHBOUND ENTRY	IORTHBOUND ENTR	ORTHBOUND ENTRY	TORTHBOUND ENTRY	ORTHBOUND ENTRY	ORTHBOUND ENTRY	PRING FARM PARKY	PRING FARM PARKV	SPRING FARM PARKY	SPRING FARM PARKY	PRING FARM PARKY	PRING FARM PARKY	
ATTRONUUM TO Darfe Highway NTT Darfe Highway NTTR BORCH, INFINIS INFINISTICS, INFINIST INFINISTICS, INFINIST INFINISTICS, INFINIST INFINISTICS, INFINIST INFINISTICS, INFINISTICS INFINISTICS, INFINISTICS INFINIS	62 02		ROAD (MCFO)	ROAD (MCEO)	RTHBOUND (MCBD) - SHEET	RTHBOUND (MOBO) -	RTHBOUND (MCB0) -	RTHBOUND (MCB0) -	RTHBOUND (MCB0) -	RTHBOUND (MCB0) -	RTHBOUND (MCB0) - RTHBOUND (MCB0) -	RTHBOUND (MCBU) -	RTHBOUND (MCB0) -	RTHEOUND (MCEO) -	RTHBOUND (MCBO) -	AMP (MCS0) - SHEET	AMP (MC50) - SHEET	AMP (MC50) - SHEET	AMP (MCS0) - SHEET	AMP (MC50) - SHEET	AMP (MCS0) - SHEET	AMP (MCS0) - SHEET	AMP (MCS0) - SHEET	' RAMP (MC40) - SHEI	RAMP (MCAU) - SHE	RAMP (MC40) - SHE	RAMP (MC/0) - SHE	RAMP (MCAU) - SHE	RAMP (MC40) - SHE	RAMP (MC40) - SHE	" RAMP (MC40) - SHE	RAMP (MC40) - SHE	RAMP (MC40) - SHE	RAMP (MCAU) - SHE	RAMP (MC40) - SHEI	SECTIONS - NORTHBOUND ENTRY RAMP (MC40) - SHEET 1 OF 21 SECTIONS - NORTHBOUND ENTRY RAMP (MC40) - SHEET 2 OF 21	HAY WIN (MICTU) - SHE	IAY W/B (MC10) - SHE	IAY W/B (MC10) - SHE	IAY WIB (MC10) - SHE	IAY WB (MC10) - SHE	AY WIB (MCTO) - SHE	in a second difference of the second s
becrec Discond	PLOTONIE/1 3/10/2018			- man - mar -	SHEET 18 OF 18	SHEET 15 OF 18	SHEET 14 OF 18 SHEET 15 OF 18	SHEET 13 OF 18	SHEET 11 OF 18	SHEET 9 OF 18	SHEET 7 OF 18 SHEET 8 OF 18	SHEET 6 OF 18	SHEET A OF 18	SHEET 2 OF 18	SHEET I OF 18	12 UF 13 13 OF 13	9 9 10	90F13	BOF13	50F 13	4 0F 13 5 0F 13	30F 13	70F13	ET 21 OF 21	ET 20 OF 21	ET 18 OF 21	51160F21 51170F21	5T 15 0F 21	FT 13 OF 21	5110F21	ET 9 0F 21	578.0F 21	ET 6 0F 27 ET 7 0F 21	14 CF 21	-T30F21	= 1 0F 21 - 1 2 0F 21	EI IV UF IV	ET 16 OF 17	ET 14 OF 17	ET 13 OF 17		ET 9 OF 17	
EX SAMAN EX SALDAV ATREAVED DIGOT DHORMOO DHORMOO DHORMOO DHORMOO DHORMOO DHORMOO DHORMOO	1					and the second s			A REAL POINT AND A REAL		and the second se	-			a succession in the second second						A COLUMN TAXABLE AND A COLUMN								An anna Anna an Anna ann an Anna Anna A	and a second of the second sec			A CONTRACTOR AND			and a second s			and a second				
0.102711 0.102711 0.102711 0.102711 0.102711 0.102711 0.102711 0.102711 0.00211			1	UT-0112	1 1	1 1	UT-0107	01-0106		UT-0102	UTILITY SERVICES	SW-0423	SM-0422	1000	SMJM11	SM-0404	SM 0402	SILLINI	SM-0351	1 :		÷	COEUTIS	SM-0113		1 1	1.1	SM-0105	1 1	1 1	SM-0102	STORMWATER MANAGEMENT	RF-0201	RF-0113	1		RF-0108		RF-0105	RF-0104	RF-0102	DELINEATION, SH	SHEET No.
NSW Cards & Marthr		UTILITY COORDI		UTILITY COORDIN	UTILITY COORDINATION PLAN - SHEET 11 OF 13	UTILITY COORDI	UTILITY COORDI	UTILITY COORDI	UTILITY COORDI			TYPICAL SEDIME	TYPICAL SEDIME		TYPICAL EXTEN	TYPICAL CHANNE	TYPICAL CHANN	TYDICAL CHANNE	CULVERT SCHEDULE	PAVEMENT DRAU	PAVEMENT DRAI	PAVEMENT DRAI	PAVEMENT DRAMAGE PIT SCHEDULE - SHEET 1 OF 6	DRAINAGE AND V	DRAINAGE AND V	DRAINAGE AND V	DRAMAGE AND WATER CUALITY PLAN - SHEET 8 OF 13 DRAMAGE AND WATER CUALITY PLAN - SHEET 9 OF 13	DRAINAGE AND V	DRAINAGE AND V	DRAWAGE AND A	DRAINAGE AND V	ANAGEMENT	DELINEATION, SI	DELINEATION, SP	DELINEATION, SH	DELINEATION, SI	DELINEATION, SP	DELINEATION, SI	DELINEATION, ST	DELINEATION, SI	DELINEATION, SIGNPOSTING AND ROAD FURNITURE PLAN-SHEET 2 OF 12 DELINEATION, SIGNPOSTING AND ROAD FURNITURE PLAN-SHEET 2 OF 12	DELINEATION, SIGNPOSTING AND ROAD FURNITURE	SHEET TITLE
	142 4	ATION - UTILITY IMP	VATION - UTILITY INP	VATION PLAN - SHEE	ATION PLAN - SHEE	WITON PLAN - SHEE	WITTON PLAN - SHEE	VATION PLAN - SHEE	UTILITY COORDINATION FLAN - SHEET 4 OF 13	UTILITY COORDINATION PLAN - SHEET 2 OF 13	UTILITY COORDINATION PLAN - SHEET 1 OF	NT AND PERMANENT	NT AND PERMANENT		TYPICAL EXTENDED APRON DETAIL	LS AND DRAIN DETA	LS AND DRAIN DET/	I S AND DOAN DETA	ULE .	WIGE PIT SCHEDULE	NAGE FIT SCHEDULE	WAGE PIT SCHEDULE	VACE PIT SCHEDULE	VATER QUALITY PLA	WATER QUALITY PLA	VATER QUALITY PLA	VATER QUALITY PLA	WATER QUALITY PLA	VATER OUAUTY PLA	WATER OUALITY PLA	DRAINAGE AND WATER QUALITY PLAN - SHEET 1 OF 13 DRAINAGE AND WATER QUALITY PLAN - SHEET 2 OF 13		SNPOSTING AND RO	SNPOSTING AND RO	SUPOSTING AND RO	SUPOSTING AND RO	SNPUSTING AND RO	SNPOSTING AND RO	SNPOSTING AND RO	SNPOSTING AND RD	SNPOSTING AND RO	DEURNITURE	
MENANCLE POAD TO MEMANGLE PARK AGENERAL MOES SHEET MARKENNESS DS2017/000204 CHARTING STATE CONCEPT DESIGN	MDEH AND CAMPBEI RING FARM PARKWA	илилту соорявлиятион - илилту мираст schedule - sheet 2 ог 2	ACT SCHEDULE - SH	T 13 OF 13		T 90F 13	170F13	1 5 OF 13	T40F13	T20F13	T10F13	FWATER OUALITY BY	TYPICAL SEDIMENT AND PERMANENT WATER QUALITY BASIN - SHEET 2 OF 3		6 Ann	TYPICAL CHANNELS AND DRAIN DETAILS - SHEET 4 OF 4	ALS-SHEET 2 OF 4	UN C CHEFT 1 DE A		- SHEET 6 OF 6	- SHEET 4 OF 6	-SHEET 3 OF 6	SHEET 1 OF 6	N-SHEET 13 OF 13	N-SHEET 17 OF 13	N-SHEET 10 OF 13	N-SHEET BOF 13	N-SHEET 7 OF 13	N-SHEET 5 OF 13	N-SHEET 3 OF 13	N-SHEET 1 OF 13		DELINEATION, SIGNPOSTING AND ROAD FURNITURE - ROAD FURNITURE SCHEDULES	DELINEATION, SIGNPOSTING AND ROAD FURNITURE PEAN - SHEET 13 OF 13	ND FURNITURE PLAN	AD FURNITURE PLAN	AD FURNITURE PLAY	AD FURNITURE PLAN	AD FURNITURE PLAN	AD FURNITURE PLAN	AD FURNITURE PLAN		
DS2017/000204	LTOWN CITY COUNC	NOT FOR CO							and the second			ISIN - SHEET 3 OF 3	SIN - SHEET 2 OF 3														and a subsection of the second s	A CONTRACT OF A DESCRIPTION OF A DESCRIP		And the second s		and a second of the second	AD FURNITURE SOH	1-SHEET 13 OF 13	1- SHEET 12 OF 13	1-SHEET 10 OF 13	1-SHEET 9 OF 13	-SIEET 7 OF 15	I-SHEET 5 OF 13	1-SHEET 4 OF 13	L-SHEET 2 OF 12		
	STK	NOT FOR CONSTRUCTION			a and a second and a second and and a second	and a second			We want and the second se	The second	2 2 2 2 2	and the second se	and a second	The second s			An operation of the second sec							a second of the second second		And the second s	The same as any second of the second		 A second s	and the second s			EDULES		A CONTRACTOR OF A CONTRACTOR A C	A CONTRACTOR OF A CONTRACTOR A						A MARKET A. C. 1	
SHEET 1 OF 2	ß	NOLL			: 1				j		į				!				•					:		: 	<u></u>						.		: 								

 THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

 (a)
 (b)
 (b)
 (c)
 (c)</

	45 45 Convertient A3 CLEE OFFICIAL						
FROMAGE FALL LEXANDER TAXABLE STOLE DETAIL TIME TPETAINESS (E. DEMARDING MONTH TAXABLE TAXABLE STOLE DETAIL TIME TPECHAPTICAL INLESS TO SECTION STOLE TAXABLE STOLE TAXABLE TPECHAPTICAL INLESS TO SECTION STOLE TAXABLE STOLE TAXABLE TRANSPORT				Pr-020 WERNENT PROFILES CONSTRUCTION STAR CS-002 CONSTRUCTION STAR CS-002 CONSTRUCTION STAR CS-002 CONSTRUCTION STAR CS-002 CONSTRUCTION STAR CS-002 CONSTRUCTION STAR	PAULINETIS AND REPORT PAULINETIS AND REPORT		DES (
логит солонор - хоторят деругы - с н - ат изал - есце, кало, такото цера ат - ат изал - есце, кало, такото цера	CONSTRUCTION STAGING PLAN STAGE 3. SHEET 10 F 7 CONSTRUCTION STAGING PLAN STAGE 3. SHEET 20 F 7 CONSTRUCTION STAGING PLAN STAGE 3. SHEET 30 F 7 CONSTRUCTION STAGING PLAN STAGE 3. SHEET 40 F 7 CONSTRUCTION STAGING PLAN STAGE 3. SHEET 50 F 7 CONSTRUCTION STAGING PLAN STAGE 3. SHEET 60 F 7 CONSTRUCTION STAGING PLAN STAGE 3. SHEET 10 F 7	4 4 4 4 4 4 4 4 4	CONSTRUCTION STAGING PLAN STAGE 1 - STEET 1 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 2 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 2 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 2 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7 CONSTRUCTION STAGING PLAN STAGE 1 - STEET 5 0 - 7	PARENENT PROFILES		PROPERTY ACQUISITION - (JT 200 PP22453) PROPERTY ACQUISITION - (JT 201 PP22453) PROPERTY ACQUISITION - (JT 201 PP22454) PROPERTY ACQUISITION - (JT 201 PP22454) PROPERTY ACQUISITION - (JT 10 PP24557) PROPERTY ACQUISITION - (JT 10 PP24557)	SHEET TITLE
Area (1997)				G(G) G(G) G(G) G(G) G(G) G(G) G(G) G(G)			······································
ANTEL REPORT REP						REINFORCED CONCRETE BOX CULVERT ST-DOI ST-DOI ST-DOI ST-DOI TS-DOI TS-DOI TS-DOI TS-DOI TS-DOI TS-DOI TS-DOI TS-DOI SPENIG F-ANN PARKINAY AND MUNE TS-DOI TS-DOI SPENIG F-ANN PARKINAY AND MUNE TS-DOI SPENIG F-ANN PARKINAY AND MUNE TS-DOI SPENIG F-ANN PARKINAY AND MEMA	SHEET NO. SHEET TITLE BRICEE STRUCTURE STOOR STOOR STOOR STOOR GENERAL ARRANGEMENT SHEET A STOOR STOOR GENERAL ARRANGEMENT SHEET A
A construction of the second s						RETE BOX CULVERT COVERS HER GENERAL ANDANGEMENT - SHEET N GENERAL ANDANGEMENT - SHEET N SPENIG FAMI PARKINY NATO SUBDIVISION ACCESS ROAD SPENIG FAMI PARKINY NATO HUNE MOTORWIY NG SUTIY PANE SPENIG FAMI PARKINY NATO HUNE MOTORWIY SIG SCIT NAME SPENIG FAMI PARKINY NATO MEMANGLE ROAD	I. SHEFT A
Rubi balli, Final. Fubi balli, Final. Fubi balli, Final. Rubi balli, Final. Fubi balli, Final. Fubi balli, Final. Balli, Final. Fub balli, Final. Fub balli, Final. Balli, Final. Fub balli, Final. Fub balli, Final. Balli, Final. Fub balli, Final. <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>							
Transport New Boods & Maritime Revision Street Fromer Revision And Investment Services							
NOT FOR CONST CANDENAND CAMPRELITOWN CITY CONNCLS SPENG FANN DARKWY STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT MENANGLE ROAD TO MENANGLE PARK MENANGLE ROAD TO MENANGLE PARK NOESSHEEL MENANGLE ROAD TO MENANGLE PARK MENANGLE SHEEL MENANGLE ROAD TO MENANGLE PARK MENANGLE SHEEL MENANGLE ROAD TO MENANGLE PARK							
NOT FOR CONSTRUCTION							

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED 3 18 109 116 23 13 23 141 145 15 150mm CHARGE DESKRIL 3 18 109 115 23 23 24 145 15 150mm CHARGE DESKRIL

DEVINE R.L.COUNDERFARE VOT CALE TRE ROT B TATABAGUÉ GENULTEROTTI DESponsery HURSDONG-GENETAL LESTANGE R.L.COUNDERFARENCE INTEGRAVIANT AND A TATABAGUE HURSDAW TATABA LESTANGE R.L.COUNDERFARENCE INTEGRAVIANT AND A TATABAGUE HURSDAW TATABAGUE HURSDAW LESTANGE R.L.COUNDERFARENCE INTEGRAVIANT AND A HURSDAW LESTANGE R.L	 UTILITIES 1. THE LOCATION OF UTILITIES TO BE CANFIDMED ON SITE. THE CONTRACTOR SHOULD CARRY OUT THE FOLLOWING PEOPLE COMMEDIAINE ANY WORKS. 6) DETENDING THE EXTENT OF EXISTING UTILITY SURVEY INFORMATION (a) DETENDING THE EXTENT OF EXISTING UTILITY SURVEY INFORMATION BY TELEPHONING 1100 OR FAX 1300 492 OT TO ASCERTAIN THE EXACT LOCATION OF UTILITIES. (a) INATE MAY OTHER ENDINES AS THE CONTRACTOR CONSIDERS INCRESSIVEY TO SATISEY ITSELF AS TO THE EXACT LOCATION OF UTILITY SERVICES. AND (c) ADDR'T WORK METHODS TO AVOID DAMAGE TO UTILITIES. 2. OL-40 DENDES CANSIFICATION OF SUBSURFACE UTILITY INFORMATION QUALITY LEVEL: (c) QL-4 = CRYACE HOUSE TO SUBSURFACE UTILITY INFORMATION QUALITY LEVEL: (c) QL-4 = CRYACE SOLVING SUBJERTING. (c) QL-4 = CRYACE SOLVING SUBJERTING. 	4. UND, ALL DANINGE PIPES SHALL BE REINFORCED CONCRETE FIPE (RCP) CLASS 4 FOR LONGTUDINA. PIPES AND OLASS 4 FOR TRANSPERSE PIPES. 5. ALL PRESARE: TO BE SPECOT AND SCOKET WITH RUBBER RING JOINTS UNLESS OTHERWISE INDICATED. 6. ALL PROFORSED DRAINAGE FIT GRAINERS SHALL BE LOAVE DUTY CLASS.D. 7. EXTENSION TO CLAVERTIS SHALL BE CONSTRUCTED AS PER STRUCTURAL DEFAILS. TO BE PROVIDED AT DETAILS IN FORSING STAFE.	DRAINAGE 1. THE DRAINAGE DRAWINGS INCLUDE DETAILS OF THE FEBWANENT STORMMITTER DRAINAGE SYSTEM 1. THE DRAINAGE DRAWINGS THE CONTRACTOR SHALL BE RESCONSELE FOR ANY TRANSDARY STORMATER DRAWINGS THE CONTRACTOR SHALL BE RESCONSELE FOR ANY TRANSDARY 2. ANY FERMITS RECORDER WORKS AND EDGISION SEMILATE ORFLANED BY CONTRACTOR 3. WAVE CONFECTIVE TO DE MADE TO AN EXISTING CONTAINED BY CONTRACTOR 3. WAVE CONFECTIVE TO DE MADE TO AN EXISTING CONTAINED BY CONTRACTOR 3. WAVE CONFECTIVE TO THE EXISTING DRAWINGE STRUCTURE SHALL BE CONFERMED PRIOR TO POSITION AND LEVEL OF THE EXISTING DRAWINGE STRUCTURE SHALL BE CONFERMED PRIOR TO	PAVEMENT MARKING 1. PAREMENT MARKING SHALL BLAPFLED IN ACCORDANCE WITH PAUS R141 2. PAREMENT MARKING SHALL BE PROVIDED ON ALL LANK EDDE, BARRER, CONTINUITY AND 32. PAREMENT MARKERS SHALL BE PROVIDED ON ALL LANK EDDE, BARRER, CONTINUITY AND 32. SPEARTOWLINES, THE SPEARCH OF RANKED MARKERS SHALL BEN MACONAL 3. WAVER BOOME FAMILY STO DE USED FOR ALL ARCHING AND NUMERALS 4. WAVER BOOME PARTY IS TO BE USED FOR ALL ARCHING AND NUMERALS	KERBS AND GUTTERS 1 FOR STANDARD KERB AND GUTTER SHARES REFER RAS MODEL DRAWING RODOLOL SIGNS 1. SIGNS SHALL BE MANUFACTURED IN ACCORDANCE WITH RAIS 3400. SIGNS SHALL BE INSTALLED IN ACCORDANCE WITH REIMAN FACTORED IN ACCORDANCE WITH RAIS 3400. SIGNS SHALL BE INSTALLED IN	 CSEC NY CLASSIFIED REAMS IN NSW. REFER TO MAURICALIDERS SPECIFICATION AND RMS ACCETTANCE DOCUMENTS FOR THE MAXIMUM ALLOWABLE OWNMO DEFECTION REQUIREMENTS FOR SHETY DARREESS. NOVARAUELE OWNMO DEFECTION REQUIREMENTS FOR SHETY ADDREED FOR SAFETY BARRIER. FOR CLEME ZONE: WIETH REFER TO MASSUPPLEMENT TO AUSTROADS SUIDE TO ROAD DESIGN, PART 6 (2009), ROADSIDE DESIGN, SMETY AND BARREESS. 	AMEA) ZOME 55. SAFETY BARRIERS 1. SAFET BARRIERS PRODUCT, REFER TO NAME ACCEPTANCE DOCUMENTS FOR SAFETY BARRIER PRODUCT ACCEPTED FOR PRODUCT, REFER TO NAME ACCEPTANCE DOCUMENTS FOR SAFETY BARRIER PRODUCT ACCEPTED FOR	GENERAL 1. MX MODEL FILE USED FOR DCCUMENTATION OF ROAD PLANS ARE NOTED. IN THE DESIGN MODEL FILE REFERENCE BOX ON THE PLANS. 2. DRAWKS TO BE FEAST MO COMMUNITIAN WITH BANS SPECIFICATIONS FOR THE PROJECT. 3. ALL DRAFMSIONS IN METRIES UNLESS OTHERMISE STATED. ALL LEVELS, CHANNAGES AND COORDINATES 4. LEVELS ARE TO ANSTRALAW HEIGHT DATUM (A HD)S, CO-ORDINATES ARE TO MAP CRID AUSTRALIA 4. LEVELS ARE TO ANSTRALAW HEIGHT DATUM (A HD)S, CO-ORDINATES ARE TO MAP CRID AUSTRALIA
MICA ZONE 56 AFC		TCS 1. FOR KOTES RELATING TO TRAFFIC CONTROL SIGNALS, REFER TO SHEET No. TS-0011 TO	 TEMPORARY ROAD SAFETY BARRIERS TO BE INSTALLED PRIOR TO COMMENTANE CANCELLAND WORKS. CONSTRUCTION STAGING DESIGN ADDPTS A DB80 TEMPORARY CONCRETE BARRIER FOR THE PURPOSES OF TEMPORARY CEOLETIKE DESIGN. ALL REDUNDANT PALEMENT MARKINGS TO BE REMOVED BEFORE TRAFFIC SMITCH. ACCESS TO RESIDENTIAL AND COMMERCIAL PROPERTIES SHALL BE MAINTAINED AT ALL TIMES DIR CONSTRUCTION. 		 C) INSTALLATION OF SEDIMENT BASIN AND BEFORD DISEMATOR AT OUTLET WITH: AT DOWN SOPER AT DOWN SOPER D) INSTALLATION OF ALL DIFERSION DRAMS AND LEVEL SPREADERS E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. E) INSTALLATION OF ALL REAVABLES EEXCOUNT AND SEDIMENT CONTROLS. 	FRANCE TO BE RUMADE MOWIN (DELYMAN) EROSION AND SEDIMENT CONTROLS 1. THE SECRENCE OF CONSTRUCTION SHALL BE A FOLLOWS A) IDENTIFY LOCATION OF ALL REPORTED EROSION AND SEDIMENT CONTROL MEASURES. B) INSTALLATION OF BARRIER AND SEDIMENT FENCES.	 SPRING FARMI PARKWAY & MENANGLE ROAD WILL BE DESIGNED TO EMDEAVOUR STANDARDS AND WILL BE CONNECTED TO THERE LYSTREEFT LISENT RETWORK. LIGHTING DESIGN ADD INSTRLATION WILL NOT BE PROVIDED FOR THE HUME MOTORWAY OR FOR THE TEMPORARY LIGHTING DURING THE CONSTRUCTION WORK. LIGHTING COLLINAT TYPE TO BE INPACT ASSORBENT AND STRUCTURE TO BE GALVANISED STEEL LIGHT 	LIGHTING 1. SPENKE FARM PARTWAY, BUTRY / EXIT RAMPS AND THE BEDDGE OVER HUME MOTORWAY WILL BE LIT TO 2. SPENKE FARM PARTWAY. BUTRY / EXIT RAMPS AND THE BEDDGE OVER HUME MOTORWAY WILL BE LIT TO CATEGORY 73. 2. ALL RITERSECTIONS MILL BE LIT TO CATEGORY 74. 3. MENANGLE ROAD: CARRINGEMENT LIT TO CATEGORY 74. 4. ENTRY: FEAT RAMPS WILL BE DESIGNED TO RUS STANDARDS AND WILL BE CONNECTED TO AN RMS MAP SWITCH SMARD MASH.
		EF No. 15-0001 TO TS-0004.	XMMENA THE SCHOLDEN XETTE BARRIER FOR THE KEFTC SMITCH MANTANED AT ALL TIMES DURING	TIOR RESTRICT PUBLIC ACCESS DURNIG R TIO ANY EARTHWICHNIG:	nt outret with: Sedwight Fraces 985 987 Mittades Mittades 100MJ, Prior to Ann other	IT CONTROL ME/SURES.	ndeavour standards and Mill E hume motornway or for the : To be galvamsed steel light	L HUME MOTORWAY WILL BE LIT TO ANN UGHTING. LL DE CONNECTED TO AN RMS MAIN
NOT FOR CONSTRUCTION Register a Maritime Register a Maritime <								











THIS DRAVING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF C	OPIED						
AL CADASTRE (DP OVERLAN) PDDE	Existing Levels Design Offset	DESIGN	CUT	Un currant	EXISTING LEVELS DESIGN	DESIGN	CUT
The period	-60.000 99.258 -59.124 99.230	99.022 93.968	EVA1				DATUARL 99.0
ITRIAN SAFE	99.033 -52.303 -50.263 98.695	i	MCBO	-			גר 99,0 גר
TY FENCE			0,4%				
P SAFET	-43.128 - 98.230	09.405	L180				
P SAFETY BARRIER P SAFETY BARRIER Annore Source in Macanon Since two services and services a	-35.388 97.379 -34.467 97.905	99 361	(184				
o A A	97,359		1.0%	-24	,750 98.248	96.248 b	AB0
EXSTING PAVENENT	96.685						ter .
EMENT IN IBACONS DOCUMENT	98.520			-11	.680 95.619 .180 95.559	99.514 E 09.544 E	2H82
				-7.0	95.459 95.432	99.484 H	نم چ
EXISTING SUPFACE	-3.500 96.106 0.000 95.871 0.801 95.816	1	L110 43 LEII MC10 KF1A	-3.5 000 1.00 2.20 4.2	00 95,160	59.594 1	.110 .E11 .E11 .E10 .V10 .M0
RFACE	3.914 95.677 4.700 95.622 8.200 95.503	99.060	KF1A MC20	4.2 4.7 8.2	00 94 963	89.625 L	.E20 🔓
Power Provide Annual Annua	11.700 95.419 12.584 95.365	93.850	LIB5	11. 12.	700 94.802	99.415 L	121 (3) 185 (3
DESIGN SURFACE	15.161 95.346 19.682 95.274		LIB5	17.	54.009		CAB5
URFACE Sector of the sector of	95.263 23.379 95.226	98.730	EHB5	20.	987 94.613	09.268 E	2HB5 5
10220115 10220115 10220115 10220115 10220115 10220115	30.542 95.150	85.150	IAB1	30.	537 94.495	94.495 - 1.	AB1
REALES AND IN							
Transr Roads Service Service							
Roads & Martime Roads & Martime Roads & Martime Roads & Martime Roads & Martime Roads & Martime							
6 CANDE SPRING SPRING ROSS SUB- ROSS SUB- SUB- SUB- SUB- SUB- SUB- SUB- SU							
CANDER AND CAMPER SERVICE AND CAMPER SERVICE AND CAMPER SERVICE ROAD MENANGLE ROAD MENANGLE ROAD CROSS SECTIONS SECTIONS OF CROSS SECTIONS OF CONCEPT DESINU		.d			l a anna an thinn		an e e e e e e e e e e e e e e e e e e e
NOT BELTOWN WAY O DEVELC O DEVELC SPRINT AND TO M SPRINT AND SPRINT AND SPRINT AND SPRINT AND SPRINT AND SPRINT AND SPRINT AND SPRINT AND SPRINT AND SPRINT AND							
NOT FOR CONSTRUCTION STAGE 1 ROAD DEVELOPMENT MENANGLE ROAD TO MENANGLE PARK ROAD ALGUMENT AND FARTHWORKS SCONGET DESIGN							
Noneta and Anorem Market Mar							
CILS CILS CILS CILS CILS CILS CILS CILS							
				• .			









<u>ā</u> ā	DFFSET	EXISTING LEVELS	Design Levels	CUT STRINGS		DESIGN	EXISTING LEVELS	design Levels	CUT STRINGS		
FEDESTRIAN SAFETY FENCE					DATUM RL 840					DATUM R.L. 84.0	
A HELI ROUGH	-40.401	89.800	89.696	IA10	-40.2	.344	88.854	88.854	IA10		12
TY BARRIER EXISTING PAYEAENT	-28.180 -28.180 -23.680	89.555	95.806 95.839 95.356	1810 1809 1F10	28.		88.779 88.677 -	94.936 94.486 -	1810 1F10		10.07
	-11,680 -90,180 -7,000 -7,000 -3,500 -7,000 1,600 4,200 4,700 8,200	89.268 69.252 69.225 89.218 89.155 89.127 89.127 89.106 69.100	101.356 101.388 101.325 101.221 101.118 101.669 100.930 100.975	EH10 EX10 KA10 L110 LA11 KF1X KF1X LE20 L121		000 000 500 600 100	88.713 88.709 88.704 88.704 88.704 88.701 89.664 88.560 83.516 63.515	100.438 100.616 100.458 100.351 100.246 100.147 100.105 100.000	EH10 EX10 KA10 L110 LA11 KF1X LE20 L121		12 12 12 12 12 12 12 12 12 12
DESIGN SUPPROS Autorate me Autorate me Autorate me Autorate me access superson Deservice superson Deservice superson Deservice superson Deservice superson	11.700 16,380	69.046 69.039 69.019	100.765	KA2J EH2J	11.2 11.2 11.2 15.3 15.3	700	89.615 88.639 88.697	99.695 99.945	KA2J EH2J	,	10% y20%
2007 100 100	27.380 31.880	88.921 85.895	94.815 95.265	if2J IB2J		.380	88.563 88.567	93.945 94.395	1F2J 182J	· · · · · · · · · · · · · · · · · · ·	
ALANI ITENSOORT NSW Reeds & Marthme NSW Services Nexusion	44,716	88,848	69.848	1421	43.	.749 -	88.481	89.461	1A21		- K. K.
NOT FOR CONSTRUCTION											

LEGEND									
ARY DIGITAL CADASTRE (DP ONERLAN) PEDE COLET TALE - FADILIE - HADDEN - FADILIE - HADDEN - HADDEN	DESIGN OFFSET	EXISTING LEVELS	DESIGN LEVELS	CUT STRINGS	OFFSET	EVELS	Design Levels	CUT STRINGS	
PEDESTRUM SKETY FEVCE	-60.000	83.470	63.216	IAGO					DATUM RL 830
CE I SAFETY BARRIER	-39.329	88.313	88.313	IA10	-38.692	87 596	87.596	IA10	1.2.2.
COLORANS STRUCTURE	-28.300 -23,800	88.252 88.241	93.827 93.377	IB10 IF10			92.692 92.242	1810 IF10	1001
	-11,600 -10,300 -7,120 -3,560 -2,500	68.172 88.195 88.126 88.118 88.068 87.941	99.377 99.407 99.347 99.234 99.122	EH10 EX10 KA10 L110 LA11		87.640 87.601 87.537 87.520 87.425 87.425	98.242 98.272 95.212 08.066 97.919	EH10 EX10 KA10 L110 LA11	7-20% 40% 40% 40%
EXISTING SURFACE DES EXISTING SURFACE DES EXISTING SURFACE EXISTING SURFACE EXIS	3,300 4,700 8,260 11,820	87.890 87.858 87.836 87.836 87.804 87.796	99.017 98.974 98.862 98.750	KF1X LE20 L12I KA2J	الم معنى الم الم معنى الم معنى الم الم معنى الم معنى الم الم معنى الم معنى الم الم معنى الم معنى الم الم معنى الم	87.279 87.266 87.236	97.787 97.732 97.585 97.439	KF1X LE20 L121 KA2J	40% 40% 40% 40%
ICN SURFACE ICN SURFACE ICN SURFACE ICN SURFACE ICN SURFACE ICN SURFACE ICN SURFACE ICN SURFACE	15.500 27.500	87.770	98.800	EH2J	3 15 693 2 2 2 2 2 7 2 7 2 7 6 2 2 7.693		97.489	EH2J	22 Cr7
	32.000	87.750	93.250	182J	5 5 41.62		91.940 87.125	1B2J 1A21	H A
NSW STREET	43,266	87.817	87,617	1421				Management and a device of the manufacture of the second sec	
NOT FOR CONSTRUCTION	63,000	87.609	87.603	1421					

	0 7	5.0	5 9	4 G	98	16 Q		ឡ ស្	
5 8	dffset	EXISTING LEVELS	Design Levels	CUT STRINKS	DESIGN OFFSET	EXISTING LEVELS	DESIGN	DATUM RL. 80.0 CUT STRINGS	
PEDESTRIAN SAFETY FENCE				220				0.08	
P SAFETY BARRIER	-37.441	87.087	87.067		-36.823	88.537	86.537	IA10	2011
TY BARRIER					-32.580	88.527	88.659	(B10	3
REACTING	-28.680 -24.080	87.022 87.030	91.497 91.047		-28,080	66.539	88.209	IF 10	222
1 8 8 S	-12.030 -10.580	86.962 86.963	97.047 97.077	EHIO EXIO	-12.080 -10.580	88.671 86.539	96,209 06.239	EH10 EX10	p-2.0%
	-7.400	86.981 86.956	97.017	KA10	-7.400	86.485 88.471	00,100	KA10	\$
	-3.700 375 0.000	87.013 86.972	96.869 98.721	L110 LA11 KF1X	-3.700 -	86.503 86.620	96.031	L110	A MC10
	3.300 4.687	86.911 86.690	96,589 96,534	KF1X		66,603			MC20
	8.387	85.836	98.386	574 A		68.285			▲ 10%
DESCHORED OF DESCRIPTION	12.087 15.767	68.782 88.768 86.715	96.238 96.288	KA2J	12.923	63.223 66.206	95.383	ES41	 ආ ප්. දිදි
DESTANDARS INTERACE			WARK Lange - Lanceson Mc	2. Ar	10.466	86.125	95.142	E9141	
RFACE		A DESCRIPTION OF THE PROPERTY				a come construction of the second			1
11220100 11120100 11120100 11120100 11120100	28.446	86.559	89.949	IF41		and the second second second			
	32.946	86.482	90.398	1841					
	41.006	- 86.367	86.367	1441					
Transport Roads & Maritime Services									;
rt Maritim			VELOPMATERIA STATISTICS						
RO STAR		Add and over 1 weeks on the							
DEN AND CAMPBE ING FARM FARMA AGE 1 ROAD NANGLE ROAD NANGLE ROAD NANGLE ROAD NANGLE ROAD STATUS			<u> </u>		60.000	64.954	93.043	EH4I	<u></u>
NOT FOR CONSTRUCTION CANDEN AND CAMPBELTOWN CITY COUNCILS SPENDE FARM FARMAL STAGE 1 ROAD DEVELOPMENT MENANGLE ROAD TO MENANGLE PARK ROAD ALCINEUT AND EARTHWORKS ROAD ALCINEUT AND EARTHWORKS ROAD ALCINEUT AND EARTHWORKS ROAD ALCINEUT AND EARTHWORKS ROAD ALCINEUT AND FARMA PARTICIPATION RECEIPTIONS - SPEND FARMA PARTICIPATION FOR THE SOFT RECEIPTION DISTURDED FOR THE SOFT FOR THE SOFT FOR THE SOFT RECEIPTION FOR THE SOFT FOR THE SO									
R CON UNCLS IT GLE PAS GLE PAS HWORKS									
CONSTRUCTI retus SLE PARK MORKS 204 204 204									













0 5 10 15 20 13 20 15 20 15 20 16 20 mm CMA3 SUE C6:GRUe 0 15 10 15 20 15 20 15 20 16 20 mm CMA3 SUE C6:GRUe 1 1 1 1 15 10 16 20 mm CMA3 SUE C6:GRUe 1 1 1 1 10 <	1								
NGTIAL CADASTRE (DP CVERIAN) - Forter - Forter - South International International - South International International - South International Internatione International Internation	OESKIN CRESET	EXISTING LEVELS	DESIGN LEVELS	CUT STRAKES	OFFSET	EXISTING	DESIGN	CUT STRINGS	DATUM RL 77.0
I PEDESTRUM SVEETY FENCE P SVEETY BARGER EXERTING PAVEMENT <	-20 605 -15 563 -14 080 -7400 -7400 -7400 -7700 -700 -700 -700	83.518 83.806 83.891 84.034 84.279 84.499 84.718 85.079 85.209 85.239 85.454 85.779 85.454 85.779 85.454 85.641 95.714 85.805 85.632	03.518 26.030 52.060 85.760 85.712 85.564 65.366 85.062 85.062 85.068 24.906 24.935 85.6892	AE0 AE0 EHED AB EXE0 AB L10 AB L10 AB L10 AB L110 AB L110 AB L110 AB L110 AB L111 AB EV11 AB EV11 AB EV11 AB EV11 AB EV20 AB EV20 AB EV20 AB EV21 AB EV21 AB EV21 AB EV21 AB EV21 AB EV22 AB AB AB EV20 AB EV21 AB EV21 AB EV21 AB EV21 AB AB AB AB AB AB AB AB AB EV21 AB AB AB <t< th=""><th>-19,537 -15,520 -14,030 -14,030 -10,900 -7,409 -3,760 -3,760 -3,760 -3,760 -3,760 -3,760 -3,270 -3,384 9,384 -13,085 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -</th><th>83.652 83.826 83.974 84.159 84.352 84.570 84.724 85.001 85.1021 85.623 85.451 85.451 85.913</th><th>83.652 85.625 85.625 85.625 85.625 85.169 85.041 85.021 84.873 84.873 84.873 84.863 84.863 84.863 84.853 85.913</th><th>IAE0 EHE0 EXE0 IA10 IA11 EV11 IM11 EV11 IA21 IA21</th><th>P215 432 433 403 403 403 403 403 403 403 403 403</th></t<>	-19,537 -15,520 -14,030 -14,030 -10,900 -7,409 -3,760 -3,760 -3,760 -3,760 -3,760 -3,760 -3,270 -3,384 9,384 -13,085 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -15,725 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -12,031 -14,030 -	83.652 83.826 83.974 84.159 84.352 84.570 84.724 85.001 85.1021 85.623 85.451 85.451 85.913	83.652 85.625 85.625 85.625 85.625 85.169 85.041 85.021 84.873 84.873 84.873 84.863 84.863 84.863 84.853 85.913	IAE0 EHE0 EXE0 IA10 IA11 EV11 IM11 EV11 IA21 IA21	P215 432 433 403 403 403 403 403 403 403 403 403

LEGEND									
(OP OVERLAN) [PEDI Umorine invision (Autorine in Neuronal Matternal Social vision en converting and	DESIGN	EXISTING LEVELS	DESIGN	CUT STRIVES		Design	EXISTING	design Levels	STRINGS
L PEDESTRUM SAVETY FENCE P SAFETY BARRIER EXISTING PARELEXT EXISTING SURFACE DESIGN SURFACE NOT FOR CONSTRUCTION man man <th>-15,678 -14,176 -10,991 -7,200 -3,600 -3,600 -177 -775 -3,600 -1,200 -1,18 -1,18 -8,600 -9,600</th> <th>83.909 84.000 84.152 84.193 84.421 84.602 84.773 84.950 84.974 85.056 85.191 85.237</th> <th>83 641 85 605 85 637 84 983 84 925 84 870 84 84753 84 84753 84 84753 84 84 753 84 851 84 851 85 854 84 851 85 854 85 854 84 851 85 854 85 855 85 855 85 85 85 85 85 85 85 85 85 85 85 85</th> <th>ME0 0 CHE0 0 CHE0 0 EXE0 0 L10 0 L10 0 L10 0 L11 0 EX11 0 EX21 0 L121 0 EX21 0 L121 0 L122 0 L121 0 L121</th> <th>8</th> <th>-27.755 -23.336 -21.718 -12.0596 -12.275 -7.650 0.000 3.300 9.244 9.600 13.322 21.460 21.460 21.460 23.568</th> <th>83.107 R1.353 83.413 83.657 83.656 84.001 84.149 84.201 84.616 84.651 84.701 84.656 84.701 84.656 84.701 84.656 84.701 85.059 85.199 85.248 85.350 85.452</th> <th>83.107 24.195 84.247 24.234 84.301 84.303 24.355 24.439 24.439 24.439 24.439 24.439 24.439 24.439 24.333 364.227 24.212 24.380 65.452</th> <th>IAED Image: Second se</th>	-15,678 -14,176 -10,991 -7,200 -3,600 -3,600 -177 -775 -3,600 -1,200 -1,18 -1,18 -8,600 -9,600	83.909 84.000 84.152 84.193 84.421 84.602 84.773 84.950 84.974 85.056 85.191 85.237	83 641 85 605 85 637 84 983 84 925 84 870 84 84753 84 84753 84 84753 84 84 753 84 851 84 851 85 854 84 851 85 854 85 854 84 851 85 854 85 855 85 855 85 85 85 85 85 85 85 85 85 85 85 85	ME0 0 CHE0 0 CHE0 0 EXE0 0 L10 0 L10 0 L10 0 L11 0 EX11 0 EX21 0 L121 0 EX21 0 L121 0 L122 0 L121 0 L121	8	-27.755 -23.336 -21.718 -12.0596 -12.275 -7.650 0.000 3.300 9.244 9.600 13.322 21.460 21.460 21.460 23.568	83.107 R1.353 83.413 83.657 83.656 84.001 84.149 84.201 84.616 84.651 84.701 84.656 84.701 84.656 84.701 84.656 84.701 85.059 85.199 85.248 85.350 85.452	83.107 24.195 84.247 24.234 84.301 84.303 24.355 24.439 24.439 24.439 24.439 24.439 24.439 24.439 24.333 364.227 24.212 24.380 65.452	IAED Image: Second se

1415 DRAWING MAY BE PREPARED NICOLOURAND MAY BE INCOMPL 1 1 1 2 13 3 14 14 14 14 14 1 1 1 2 13 3 14 14 14 14 1 1 1 2 13 3 14 14 14 14 1 1 1 13 13 14 14 14 14 1 1 1 15 14 14 14 14 1 1 14 14 14 14 14 1 14 14 14 14 14 14 1 14 14 14 14 14 14 1 14 14 14 14 14 14 1 14 14 14 14 14 14 1 14 14 14 14 14 14 1 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 14 <th>DESIGN</th> <th>EXISTING LEVELS</th> <th>DESIGN LEVELS</th> <th>CUT STRINKS</th> <th>DATUM RL 77.0</th> <th></th> <th>DESIGN</th> <th>EXISTING</th> <th>DESIGN</th> <th>STRAKS</th> <th>DATUM RL 77.0</th>	DESIGN	EXISTING LEVELS	DESIGN LEVELS	CUT STRINKS	DATUM RL 77.0		DESIGN	EXISTING	DESIGN	STRAKS	DATUM RL 77.0
NI SAFETY FENDE P SAFETY BARRIER EXISTING PALENENT EXISTING SURPACE	825	83.405 83.462 83.601 84.149 84.511			MC 10 MC 20		15.208 12.580 3.500 0.000 5.598 5.500	02.955 83.049 83.153 83.317 83.240 83.460 83.460 83.460 83.616 83.527 83.723	82 955 83,812 83,434 83,479 83,689 83,689	IAEI EHEJ LEEI LI10 LEII KFIE KFIM	AD1 20% AD25 AD26 MC10
ISN SURFACE	26.395 54.592	84.853 85.050 85.050 85.060 85.590 85.690 85.627 85.634 85.634	83.615	KFF2	MCF0	2	18 300 22.380 23.328	83,957 83,976 84,061 84,085	83,434 83,612 64,085	KAFO EHFO JAFO	
NOT FOR CONSTRUCTION SPRING FRAM PROVING INF COUNCIES STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT MENANAUGLE ROAD TO MEDIAINGLE PARK ROAD ALICAIMENT AND EARTHWORKS ROAD ALICAIMENT AND EARTHWORKS MENANAUGLE SPRING STATE 110 F 17 CONSTRUCTIONS SPRING FAIL OF 17 CO		I						1			





1 10 13 20 23 23 23 24 55 50 16 50 17 18 18 17 18 18 10 11 </th <th>DESIGN</th> <th>EXISTING</th> <th>DESIGN</th> <th>CUT STRINGS</th> <th></th> <th>DESIGN OFFSET</th> <th>EXISTING</th> <th>DESIGN</th> <th>CUT STRINGS</th> <th></th>	DESIGN	EXISTING	DESIGN	CUT STRINGS		DESIGN OFFSET	EXISTING	DESIGN	CUT STRINGS	
DASTRE m m m m m m m	19 SI	LS	25	ŝ	DATUMRL 220	ЦX	SN	61 ž	8	DATUM RL 820
(DP OVERLAY) PEDESTRUM SNEETY FENCE DECEMBER 18200					8					20 () 1 1 1 1 1 1 1 1 1 1 1 1 1
P SAFETY	-22.145	84.628	84.628	1A40		-19,434	85.165	85.155	1440	
CCARSTRUCTURE CCARSTRUCTURE CCCARSTR					12.22			And a final sector of the first		ta se
	-8.678 -7.378 -6.493	85.643 85.677 85.692	91.360 91.388 91.280	EH40 BF40 ES40		- 8 ,235 -8.035	86.129 88.217	90.749	EH40 BF40	
	-4.498	85.735	91,204	LE40	3.8% 3.8%	-8.055 -4.055	86.248	90.668 90.853	ES40 LE40	0.3%
	125 0.000	85,831	91.033	L140	NC40 55	0,000	88.342	90.623	L140	MC40 날 공
•	4.322 6.197 7.197	85.692 85.915	90.869 90.798 90.798	LE4I KO4I CB4I	<u>ل</u>	4.100 5.100 6.100 6.300	86.329 86.328 86.338	90.592 90,584 90,584 90,580	LE4I KO4I CB4I BG4I	-0.7%
DESIGN SURPACE	7.187 7.397 8.697	85.92 8 85.977	90.784 90.768	EG4I EH4I	g_200%	7.600	86.361	90,554	EH41	100%
I I I I I I I I I I I I I I I I I I I I					201	15.560	86.579	- 88.579	- 1441	
ALAY Transport Reads & Maritime NSW Services Parazora	17.436	66.403	E6.408	IA4I						
CARLEN AND CANOBEL SCRING FRANCPERSON SCRING FRANCPERSON STAGE I ROAD MENARCIE COAD HOAD ALIGNMEN HOAD ALIGNMEN HOAD ALIGNMEN HOAD ALIGNMEN HOAD ALIGNMEN										
NOT FOR CONSTRUCTION										


















THES DRAWANG MAY BE PREPARED IN COLCUR AND IN COLCUR AND IN COLCUR AND INCLUDES THAN SAFETY FRANCE IN COLLINE INCLUDES THAN SAFETY FRANCE IN CONTROL LINE INCLUDES THAN SAFETY FRANCE INCLUDES THAN SAFETY FRA		OFT STRINGS LEVELS LEVELS DESIGN DESIGN DESIGN
PEDESTRUAN SAFETY FENCE P SAFETY BURGHER 	-24,787 99,334 58.334 IA40	-22,117 100,178 100,178 1440
EXSTING F.MEUERT EXSTING SURFACE	-8.000 102.235 166.727 EH40 -1.273 102.733 106.753 67.40 -1.273 104.476 100.631 IE40 -1.273 104.476 100.631 IE40 -1.273 104.476 100.631 IE40 -1.273 -1.273 -1.273 104.476 100.631 IE40 -1.273	3.500 107.126 107.255 LE41
DESIGN SURFACE	8.948 106.313 106.313 LEE0 12.753 106.194 106.194 L180 18.457 106.063 108.083 LE81	7.808 107.118 107.118 LE80
NOT FOR CONSTRUCTION NOT FOR CONSTRUCTION SPRING FAME PRAVIOW SPRING FAME PRAVIOW STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT REPART ROAD DEVELOPMENT ROAD ALIGNMENT AND EARTHWORKS ROAD ALIGNMENT AND RATIO RECOMMENTATION AND REAL RATIONAL AND REAL RECOMMENTATION AND REAL RATIONAL AND REAL RECOMMENTATION AND REAL RATIONAL AND REAL RATIONAL AND REAL RECOMMENTATIONAL AND REAL RATIONAL AND RATIO		

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOM	PLETE IF COPIED NAL									
	OFFSET	EXISTING	DESIGN	CUT STRINGS		DESIGN OFFSET	EXISTING	DESIGN	CUT STRINGS	DATU
DIGITAL CADASTRE (DP OVERLAN) PEDESTRUM SAFETY FENCE SAFE - Training - Statistic Statistics - Statistic					DATUM R.L. 101 9					DATUM RL 102.0
TYBARRER EXISTING PAVEN PARAECT PERANDINASTING TWEE SOLUTION AND AND AND AND AND AND AND AND AND AN	-17.004 -6.750 5.450	103.091	103.091	EH40	20%	-12.229	105.834	105.834	IA40 EH40	2.05
LEAT EXTING SURFACE	-5.450 -4.750 -4.250 0.000 3.£00	105.508 105.842 107.149 107.951	108,251 108,155 103,255 103,255 108,127 108,127	BF40 CF40 E340 LE40 LE41	NCCO 2007 2007 2007 2007 2007 2007 2007 200	-4.669 -3.969 -3.469	106.901 107.229 108.158 108.735 108.664	108.995 108.899 108.999 108.895 108.895 108.790 108.664	BF40 CF40 KO40 LE40 LE41 LE80	а 33 35 36 36 36 36 36 36 36 36 36 36 36 36 36
	6.795 10.593 14.301	107.898	- 107.898 107.781 - 107.647	LE80 L160 LE81		9.713 13.415	108.571 108.451	108.571	L180 LE81	24%
Transport Reads & Maritime Services										
NOT FOR CONSTRUCTION ISPACE 1 ROAD DEVELOPMENT MENANGLE ROAD TO MENANGLE PARK MENANGLE PARK MEN										

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLET											
(新) S (云)	DESIGN	EXISTING	DESIGN	CUT STRINGS		Urrsei	DESIGN	EXISTING LEVELS	Design Levels	CUT STRINGS	
DIGITAL CADASTRE (DP OVERLAY) PEDESTRUM SAFETY FENCE =					DATUM R.L. 106.0						DATUM R L. 107.0
											1
P SAFETY BARRIER P SAFETY BARRIER WINDER PROJECT INFORMATION WINDER PROJECT INFORMATION P SAFETY BARRIER P SAFETY B										 Control of the second seco	
SKISTING PAVED					b	-8.1	652	112.111	112.111	1440	/ <u>1</u> 35
40000	-8.877 -5.500 -4.000 -3.000	109,409 109,435 109,502 109,523	109.409 109.703 109.696 109.696	IA40 EH40 CB40 ES40	0.5%	-4,	500 000 000	111.323 110.948 110.742	110.535 110.490 110.490	EH40 C840 E540	
CHILDREN CORRECT	0.000	109.578 -	109.662	LE40	13 NC40	725 0.0	000 -	110.440	110.430	LE40	20% MC48
DEST RECEIVENCES	3.500 5.153	109.527 109.481	109.557 109.481	LE4I LE80	4.5% 5%	3.£ 4.8	500 524 -	110.309 110.283	110.325 110.283	LE41 LABO	4.1% 2.7%
SN SLIPEF	8.955	109.380	109.380	L 180	23 72		325	110.179	110.179	L180	3.7%
102.02 10 102.02 10	12,658	109.257	109.257	LE81						An and Anno Anno Anno Anno Anno Anno Anno An	
									1		
Transport Roads & Mantume Services											т.
			and a second							and a second	
NOT FOR CONSTRUCTION											
RUCTION A3 SHEEF 15 OF 21 Nature Services											







THIS DRAWING HAV BE PREPARED IN COLCUR AND MA	A BEARDONALE E COPED			<u></u>		
SEND COMTROL LINE BOUNDARY DIGITAL CADASITIE (DF OVERLAY) PEDESITIANI SWETY FENCE In LICONTER MARK - KOTE - KNOTE - KNOTE INTERNAL - STERM ALL CARACTER MARK - KNOTE - KNOTE INTERNAL - STERM LICE RUSS - CENTRAL CONSTRUCTION - STERMAN LICE RUSS - STERMAN - ST	EXISTING LEVELS DESIGN OFFSET	STRINGS DESIGN LEVELS	DATUH RL 120	LEVELS DESIGN OFFEET	- CUT STRINGS DESIGN LEVELS	DATUM RL 1120
P SAFETY EAROLER EXISTING PARENT MALET REPARENT TINCTORE MISSION ANNUALES EXAMPLES IN COMPANY ANNUALES IN AUG DAVISOR ANNUALES IN AUG DAVISOR	-6.979 116.021 -5.500 115,725 -4.000 115.466 -3.000 115.293	116.021 (A40 115.281 EH40 115.236 EB40 115.236 ES40		-7.024 116) -5.500 116; -4.000 115; -3.000 115;	298 115.774 EH40 956 115.729 CB40	292 - 303 - 375
EXISTING SURFACE DESIGN SURFACE INFORMATION INFORMATION INFORM	8 0.000 115.103 1.792 116.071 5.594 114.840 9.299 114.791	116.124 LE40 115.071 LA80 114.940 L180 114.791 LE81	<u><u>,</u> <u>MC40</u> <u>(a)</u> (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c</u>	4.847 115.	617 115.629 LE40 505 115.505 LA80 469 115.469 L160 334 115.334 LE81	
NOT FOR CONSTRUCTION CANDEN AND CAMPELLIOWN ONLY CONNELLS AS STAGE 1 ROAD DEVELOPMENT & Markhime STAGE 1 ROAD DO MENANGLE PARK & Markhime MEANANGLE ROAD TO MENANGLE PARK SS CONSTRUCTION ENT AND EARTHWORKS SECONS SECONS EET OFFICE MEANANCE IN DATIONOLULA IN ADD CARTHWORKS EET OFFICE MEANANCE IN DATIONOLULA IN ADD CARTHWORKS EET OFFICE MEANANCE IN ADD TO MENANCE INFORMATION SHOLL IN ADDITION SHOLL IN AD						



THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMP 0 15 10 115 120 125 130 131 145 145 145 150 145 122 06/0004	LETE IF COPIED		
LEGEND LEGEND AND AND AND AND AND AND AND AND AND AND			
IV FEINCE P SHEETY BARRIER EXISTING PAVEMENT EXISTING SURFACE Image: State of the state of	OR MINGO		
DESIGN SUFFACE Training of the subtrain su			
NOT FOR CONSTRUCTION SPRING FAMILY PARAMET SPRING FAMIL PARAMET SPRING FAMILY PARAMET <th colspan<="" td=""><td></td><td></td></th>	<td></td> <td></td>		































LEGEND	0.2			60		05	50		<u>ୟ</u> ପ୍ର	
L CADASTI Hol St Rotes L CADASTI Rotes Rotes Rotes Rotes	OFFSET	EXISTING LEVELS	Design Levels	CUT STRINCS		OFFSET	EXISTING	Design Levels	CUT STRINGS	
NTY DIGITAL CADASTRE (OP OVERLAY) PEDE DIGITAL CADASTRE (OP OVERLAY) PEDE UNIVERSIDATE - HOUSE NOT INSCRAMME REPORT INSCRAMENT LEXX					DATUMRL 55.0					DATUM RL SEO
L PEDESTRUM SAFETY FEINCE			n vien en al de la constante d							
P SAFE	(8.000	00.022	00.020	14.00		-17.03	99,689	93.089	IA80	
C DRAWN	-18.922	98.962	98.962	1880 1	4.07					4.02
	-12,100	99.431	100.167	EHB0	-2.0%	-12.10		100.321	EHB0	20%
EXISTING PAVEMENT	-9.900	99.707	100.211		3.0%	-9.900 -7,400		100.365	ESRO LEBO	3.05
×400	-7.400	100,008	- 100.137	- LEBO	30%	-7,400	39.630	100.291		3.0%
IBAT EXISTING SURFACE	-3.700	100.030	100.02 6 99.915	L180 LF81	୍ଥ ତୁଟ୍ ଲ(280	-3,700 125 0.000		100.150	L180	는
EXISTING SURFACE	0 0000 0.194	99.892 99.883	99,908	LFAI	400%	රිති 0.000 0.911	100.048 103.022	100.070 100,043	LFBI	44 70 70
FLOT INVERTIGATION SURF	3895	99717	99 797	LIAI	<u>بن</u> چې	4.613	99.847	99.932	LIAI	
LESURER LESURE	7.597	99.546	99.687	LAAI	د. چ	8.315	99.690	99.822	LAAI	.3.0%
DESIGN SURFACE	10.597 11.278	99.363 99.278	99.597 99.708	KAAD CBAD	2.0%	11.31 11.99	1	99,732 09.842	KAAQ CBAD	
तान्त्र क मान्द्र म	13.779	98.876	99.758	EXAI	· 2	14.49		99.693	EXAJ	9
C: 10200 WES	15.329	98.537	99.769	EHAI	201	16.04		99.924	EHAI IAAI	201
	18.598	98.156	98.158	IAAI		17.95	99,250	99.177	AM.	and a support
							-		And a second	
CANDEN AND CAMPBEL SPRING FARM PARAMAN STAGE 1 ROAD DI MERANGLE CRAD ROAD ALLGNMEN GOOS SECTIONS - MBA REGENERATION DS		A MARINE A MARINE A MARINE A MARINE A MARINE A MARINE A MARINA A MARINA A MARINA A MARINA A MARINA A MARINA A M							for a ran frank of the statement	
INOT FOR CONSTI CWIDEN AND CAMERELITOWN OT FOR CONSTI STANE FRAM PRAVINY STAGE 1 ROAD DEVELOPMENT MENANAGE 2004 TO MEMANGLE PARK ROAD ALIGNMENT AND EARTHWORKS CROSS SECTIONS - MEMANGE FROM WORTHEOHING ACT CROSS SECTIONS - MEMANGE FROM WORTHEOHING ACT ACT ACT ACT ACT ACT ACT ACT ACT ACT ACT										
NOT FOR CONSTRUCTION ITOW CITY COUNCILS A3 EVELOPMENT TO REMANGLE PARK T AND EARTHWORKS T AND EARTHWORKS T AND EARTHWORKS 10000204 REC 1000204 REC 10000204 REC 1000000000000000000000000000000000000										

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INC 0 15 10 115 130 125 130 135 140 145 150mm 0/HA3 0/E 0/	SOME CONTRACT	***	nin da anna an					
LEGEND								
THE JONG THE SHORE THE SHO								
	DESIGN OFFSET	EXISTING LEVELS	CUT STRINGS DESIGN		DESIGN	EXISTING	CUT STRINCS DESIGN	
RY DIGITAL CADASTRE (DP OVERLAN) PEDESTRI DIGITAL CADASTRE (DP OVERLAN) PEDESTRI DIGITAL CADASTRE (DP OVERLAN) PEDESTRI NGC 1955-00000 (DC 1950) (DC 1950) (DC 1950) NGC 1955-00000 (DC 1950) (DC 19	. ~			DATUM RL S.D			Anno el conce - Allania cu el coltera y la envelo	CATUMR R. 95.0
CIGTAL CADASTRE (DP OVERLAY) CPEDESTRUM SVETY FENCE C - ROUP C-ROUP - Shatan - States - Sta								
SULLE Anaturouse	-17.521 5	99,081 99,	.081 ¹ AB0		-17.656	99, 154 99	9.154 IABO	
TY BARRIER				4.027				
		al control of	0.436 EHBO 0.480 ESBO	2.0%	-12.100 -9.900		0.543 EHBO 00.587 ESBO	2.0%
			0.405 LEBO	3.0%	-7.400		X0.511 LEBO	3.0 3.0
12000		i - j	et e agan anna an de de te	3.0%			an on a constant of the second	54 10 38
	-3.700	100.101 100	0.293 L180	30°%	-3.700	100,038 10	00.400 LiBO	3.0%
	-1000 -1.000	100.172 10	0,182 LFBI		175 0.000	100.228 10	00.288 LFBI	MCB0 삶
	1.797	100.122 10	0.128 LFAI	ມ	2.849	100.187 10	90.203 LFAI	
MOLEC MILL MILL MILL MILL MILL MILL MILL MIL	5.500	59.980 10	0.018 L1A		6.553	100.061 10	00.093 L1AI	
POLIDALLINE VOLIDALLINE MOLOCIAL BROCICE CESEVIORI CESEVIO CESEVIORI CESEVIORI CESEVIORI CESEVIORI CESEVIORI CESEVIORI CESEVIO	9 203	99.778 99	1.907 LAAI		10.257	99.682 91	9.982 LAAI	-3.07%
DESIGN SURFACE	1		9.818 KAAO 9.928 CBAO		13.280		9.893 KAAO	မ အ
AGE subman autorar autorar autorar autorar autorar autorar			9.928 CBA0	20% 20%	13.941	1 6	00.003 CBA0	2.0%
	16.438	\$9.657 \$9	9.999 EHAI 9.632 WAI	A A A A A A A A A A A A A A A A A A A	16,444 17,495 18,076	09.801 1	00.053 EXAI 00.074 EHAI 19.979 IAAI	203 401
							a - a chu - a'n annan an an	U
Transport Services Research Resea				: : : :		and the second se		
Transport Roads & Maritime Services								i
P STAN								- 5
CANDEH AND CAMPBE SPRING: FAND CAMPBE STACE 1 ROAD RAKIN MENANGLE ROAD MENANGLE ROAD MENANGLE ROAT MENANGLE ROAT M			an an ann an Ad Morrison					
NO NO ROAD TO ROAD TO S. MEMMAY S. MEMMAY SISN								
NOT FOR CONSTRUCTION SPRIG FAND CAMPBELLTOWN OTV COUNCILS SPRIG FAND PROVEM STAGE 1 ROAD DEVELOPMENT ROAD DEVELOPMENT ROAD DIJONMENT AND EARTHWORKS ROAD ALLONMENT AND EARTHWORKS CROSS SECTORS MEMORY ADA INSTRUMENT SUSCENTIASING DS2017/000204								
R CON XXNGUS XXNGUS VICLE PA THWORK THWORK THWORK THWORK THWORK								
CONSTRUC CILS CILS CILS CILS CILS CILS CILS CIL								
K MICERNET 4 OF 18 RC-0174 01 RC-0174 01 ADD MICENTIC SHOULD IN								
		·						

0 3 9 15 3 3 13 40 45 1254004X 1 1 13 13 13 14 14 1254004X 1 1 13 13 14 14 1254004X 1 1 13 14 15 1264004X 1254004X 1 1 10 10 10 10 10 10 1 10 10 10 10 10 10 10 10 1 10 10 10 10 10 10 10 10 10 1 10	DESIGN OFFSET	EXSTINC LEVELS	LEVELS	CUT STRINGS	CATUMRL 550	DESIGN OFFSET	EXISTING LEVELS	DESIGN	CLIT STRINCS	24TUM RL 95.0	
Y FENCE P SAFETY BARRIER EXISTING PARAMENT PROCE PROCED INSCRIPTIONS M000 HAMPOOR PROCED INSCRIPTIONS M000 MACA ZONE 5550 APD M00	-17.701 -12.160 -9.900	99.247 99.508 99.590	99.247 100.648 - 100.692	LABO EHBO ESBO	-2.0% E.0%	-18.224 -12,100 -9,900	99,204 99,592 99,730	99.204 - 100.735 100.779	IAB0 - EH80 - ES80 -		4.01 2.0% 2.9%
	-7.400 -3.700 1.000 2.004 3.081 4.063	\$9,685 \$9,832 100,154 100,200 100,167 100,165	100,618 100,503 100,369 100,369 100,202 100,307 100,278	LEBO L1BO LFBI EVBI IMBO EVA1 LFAI	30% MC80	-7.400 -3.700 -0.000 1.000 2.674 4.432 5,434	99.868 100.019 100.159 100.276 100.276 100.259 100.230	100,705 100,596 100,487 100,458 100,179 100,355 100,325	LEBO L1BO LFBI EVBI IMBO EVA1 LFAI		129% 129%
DESIGN SURFACE	7.768 11.473 13.478 14.157 16.681 17.772 15.403	100.058 99.891 99.762 99.700 99.720 99.773 100.038	100.167 100.057 99.997 100.107 100.158 100.158	LIAI LEAI KAAO CRAO EXAI EHAI IAAI	30% 30% 20% 40y	9.140 12.846 14.850 15.531 18.036 19.088	99.883 99.734 99.735 100.244	100.214 100.103 100.043 100.153 100.203 100.224	LIAI LEAI KAAO CBAO EXAI EHAI		30% 3.0% 3.0% 2.0% 2.0%
INCLUSION AND WARTING SERVICES AND CAMPELLIONN CITY COUNCILS AS A MARKING COMPELLIONN CITY COUNCILS AS A MARKING COUNCILS AS A MARKING AND A LIGHT AND EVELOPMENT IN COUNCILS AND A LIGHT AND EARTHWORKS AND A LIGHT AND EARTHWORK AND A LIGHT A											

1 5 10 15 10	KY BE INCOMPLETE IF CO CA A3 502 OFICEPUL				<u></u>			
BOUNDARY	DESIGN	LEVELS EXISTING	DESIGN		OFFSET	EXISTING LEVELS	DESIGN	CUT STIENNOS
CONSTRE (DP OVERUAY)				DATUM R.L. 550				DATUM RT. 550
P SAFETY BARRIER	-19.384	09,967 98.94	45 IABO	407	-19.336	98.921	98.921	LABO
ANYS SPROTUKE E	-12.100	99.482 - 100.7	766 EHB	-2.0%	-12.100	99.517	100.730	EHB0
AHD	-9.900	99.631 100.6	510 ES8	3	-9,900	\$9.647	100.774	ESBO
NTX 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	-7,400	99.811 100.7	735 LEB		-7,400	99.801	100,699	LEBO
AND DOCUM			ena 140	3.0%	-3.700	100.082	100,587	L180
	-3.700	100.113 100.6	624 L1B0	2 2 7	-3.103	100.002	100,307	S.
	25 -0.000 1.000	100.245 100.5 100.277 100.4		MCBO	275 0.000 1.000	100.253 100.277	100.478 100.446	LFD) MCBD EVBI
	3,415	100.343 100.0	dət imə		4.224	100.309	99.909	INBO
PLOT WATER TIME PLOT WATER TIME TITLE TITLE DESIGN WATER DESIGN WATER DESIGN WATER	5.953 6.955	100.297 100.3 100.269 100.3	-	· · · · · · · · · · · · · · · · · · ·	7.610	100.211	100.247	EVA1
DESIGN				3.0°5	8.612	100.165	100.217	LFAI
DESIGN SURFACE	10.883 -	100.112 100.	193 L1A	3.0%	12.196	99,971	100.108	L1A)
1 DATE 21 00 2014 21 00 2014 21 00 2014 21 00 2014 21 00 2014 21 00 2014	14.371	99.851 100. 99.649 100.		3.03	15.801	99,698	99.999	LEAI မှ
		99.543 100.			17.803 18.483	99.697 99.825	99.939 100.049	KAA0
ALAT ALAT NSW Roads & Martin NSW Services MEMORY AND MARTINE SERVICES	19 563 20.615	100.324 100. 100.713 100.		20%	20.986	100.670	100,098	EXAI
Transport Roads & Maritime Services					22.037	101.088	100.119	EHAI
ort Maritir VICES	24,494	102.058 102.	.136 IAA			1000 C		X.
ŝ					28.495	102.345	102.345	- IMI
NUCLENAN NUCLENAN TAGE FAT TAGE FAT TAGE FAT NOAD AL COAD AL COAD AL COAD AL COAD AL COAD AL COAD AL				1				
CALITEN AND CAMPBEL STARE 1 ROAD ROAD ROAD ALIGL ROAD ROAD ALIGL ROAD CROSS SECTIONS-MEN CROSS SECTIONS-MEN DS SECONCEPT DESIGN								
NOT NATE TOWN AT TO NOT NOT NOT NOT NOT NOT NOT NOT NOT								
NOT FOR CONSTRUCTION								
RT CON								

THE DRAWING MAY BE PREPARED IN COLLOUR AND MARY		DESIGN LEVIELS	CUT STRINGS	OFFSET	LENCES	DESIGN	CUT STRINGS	DATUM RL 550
P SAFETY BARRIER	-18.459 99.227	99.227	KB0	-18.2	0 99.09	1 - 99.091	IABO	407
- Parenting	-12.767 09.567	100.650	EH80		19 89.32	8 100,416	EH80	
EXISTING F	-10.567 - 99.717	100.694	ESB0		99.47	7 100.460	E\$80	3.0%
AT 25	-7.400 99.957	100,594	LAB0	-7.19	99.70	9 100.356	LAB0	10001111/11/1000 - mm 11 11 11 11
10012	-3,700 100.151	100.477	L180	3.56	1 99.95	0 100.249	L180	30%
EXISTING SURFACE DC-RC-D/177 TELEMENTERSURF TELEMENTERSURF SUBJECT SUBJECT SUB- SUBJECT SUBJECT SUB- SUBJECT SUBJECT SUB- SUBJECT SUBJECT SUB- SUBJECT SUBJECT SUBJECT SUB- SUBJECT SUBJECT SU	<u>ප</u> ර 0.000 100.224	100.361	LFBI	<u>المحمد المحمد المحمد</u>	0 100.0	40 100.144	LFBI	ж. мсво
SURFACE	1.000 100.249	100.329	EV8I	1.00	100.0		EVBI	
epseed Brand	4.692 100.222	99.714	илво	4.83	100.0	73 99.508	IMBO	
물질법 및 급급 !	8.486 100.062		EVA1	8.50	1	1	EVA1	
ILIR HALE ILIR SURFACE ILIR SURFACE SUBJECT SUBJECT SU	9.486 100.009	100.063	LFAI	9.60	99.86	iO 99.664	LFAI	3.0%
sahman	12.966 99.797	99.958	LIAJ	13.0	99.6	5 99.759	LIAI	30%
	16,486 99,423	99.853	LEAI	ن يو يو	0 99.3	99.654	LEAI	
NOVES TELLS	18.487 99.822 19,167 100.115	99.792 99.902	KAAO CBAO	18.5 19.1	1	1	KAAD CBAD	203
Research International States and Internationa	21.667 101.191 22,717 101.561		exaj Ehaj	21.6 5 7 22.7		1	exaj Ehai	
	28.079 102.65	102.654	IAAI		a	103.607	16.61	
INCT FOR CONSTRUCTION	28.079 102.65	102.854		28.3	94 102.	102.807		

THIS DRAWING MAY BE PREPARED IN COLOUR AN	ID MAY BE INCOMPLETE IF COPIED Som CHAI SZE ORISINIL													
LEGEND														
LINE BOUNDARY OIGTAL CADASTRE [DP OVERLAY)	DESIGN OFFSET	DESIGN	CUT STRINGS				DESIGN	EXISTING LEVELS	LEVELS DESIGN	CUT STRINGS				
ASTRE []		5 0, 2	63	R				6 		[2			
2:61				DATUM R.L. 95.0							DATUM RL 540			
L PEDESTRIAN SA'ETY FENCE							-23.851	97 479	97.479	IABO				
P SAFETY BARRIER	-19,828 98.341	98.341	IABO		- -			and a subsection of the second s			4.82			
555112 2.6 2.6 2.6 2.6 2.6 2.6 2.6 2.6 2.6 2.					١		-14.757	97.917	99.737 99.768	EHB2 EXB2				
AHH	-12.739 98.674	100.034	EHBO		1.		-13.254	97,966 98.171	99.814	CBB2				
EXISTING PAVEMENT	-10 539 98.80		E\$80	· · · · · · · · · · · ·	212		-10.069	08.223	99.703 99.641	KAB2	20%			
ENT CLOSE DOO DESIMPLY DESIMPL	-7,039 - 89.09	99.984	LABO	1	2.1%		-7.000	60,400	03,041		2.0%			
Ерит соотверственные соотверственные соотверственные соотверственные на соотверственные на соотверственные на соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные соотверственные на соотверственные на соотве на соотверственны	-3,500 \$9.38	69.908	L180				-3 500	98.718	99.570	L180				
	හි -0.000 99.61	99.834	LF8I		MCB0	375	-0.000	98.921	99,500	LFBI	MCB0			
- EXISTING SUPFACE	1.000 99.67		EVBI			Ci Ci	1,000	98,963	99.479	EVBI				
	4 379 99.74	99 349 1	14160				4,348	99.115	98.921	14:80				
PICTESLATES	8.500 99.65	4 99.661	EVA1				8.500	99,159	69.337	EVA1				
DESIGN SURFACE	0.500 99.82	1	LFAI	· · · · · · · · · · · · ·	<u>ل</u>		9.500	99.150	99.307	LFAI	-3-0%			
Xorer K	13,000 \$9.44	0 99.526	LIAI		-3.0% -3.0%		13.000	99.063	99.202	LIAI	17 			
110200 110000 11000000	16.500 99.15	7 09.421	LEAI				18 500	98.941	99,097	LEAI				
RESTERN S	18,500 99.40	3 99.361	KAAO		ن 10%		18.500 19.180	98.844 99.002	99.036 99.146	KAAO CBAD				
	19.180 4 99.65		CBAO		2.0% 2						2.0%. 2			
Ease Transport Roads & Maritime NSW Services NONZING NORTHER SPACETOFICE ROADS AND LARTINE SERVICES	21,680 100,6 22,730 101.0		EXAJ EHAJ		2.0%		21.680 22.730	99.846 100.227	99.196 99.217	exaj Ehaj	20%			
TOFFICE					2001		25,511	100.608	100.608	IAAI				
	28.134 102.3	44 102.244	IAAI		X									
COMDENAND CAMPRE SPENCE FACULATION STAGE 1 ROAD MENANGLE ROA MENANGLE ROAM MENANGLE			A set of the set of th	Angelo Januar Vinne et										
CEPT DESIN			- Accel and a second second		1									
CAMPERAND CAMPERLITOWN OTF FOR CONSTRUCTION SPENS FARM PARKING STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT ROAD ALIGNMENT AND EARTHWORKS CROSS SECTIONS - NARWING EARTHWORKS CROSS SECTIONS - NARWING EARTHWORKS CROSS SECTIONS - NARWING EARTHWORKS CONCEPT DESIGN														
DINOLS DINOLS														
S S S S S S S S S S S S S S S S S S S														
AUG9)- SHEET 8 OF 18 AUG9)- SHEET 8 OF 18 AUG9177 07 RCC-01777 07 AUG97 AUG97 07														
TION A3														
anica Fizi Specifica - Vari Englando J., Samari Kristato S.º D Stan Morris: Preteci (USE) First Stan Morris: Preteci (USE) First														
---	---------------------------------	-----	--------------------------	----------------------------	----------------------------	----------------------	---------------	-----	-------------------------------	----------------------------	--	--	------------------------	-------
ייניטי געריין איזער ססטאפאראזוטאסיי די ססטאפאראזוטאסיי די	BOUNDARY													
0 10 10 2010 10 20	DIGITIAL CADASTIRE (DP OVERLAY)		DESIGN	EXISTING LEVELS	DESIGN	CUT STRINGS			DESIGN	EXISTING	Design Levels	CUT STRINGS		
ATE -	DASTRE								-35.000 Ĩ			ร เม	10	1
SYNDDIA IS: INDE AN INDEGANE IS: INDE AND IS: IS AND IS AN	DP OVE	-3	35.000	95.929	99.500	EHIO	DATUM RL 93.0		~QQ.QQU	40.011	521504	1.7.7	DATUM R.L. SJ.D	0.5%
TIPOS AM TE / RE VISION				96.126					-31.238	95,514	99.181	LE20	<u>ន</u>	0.6%
4 DESCRIPT	1 PEDE								-29.907 -23.603 -28.463	05.560 05.604 95.609	99.172 09.164 09.156	E020 EV20 SS10	:	0.6%
S ANNO	STRIAN S										-			-0.4%
	I PEDESTRUM SWETY FENCE								-23.304	95.936	99.178	CB1A		
	INCE								-22.795 -21.653	95.979 96.063	99.037 99.016	KF1A LE11		
2 Harshrova	P SAFF		19.886	98.811	99.391	EH82								ω
SCALES ON A	SAFETY BARRER		18.244	95.834	99.431	EXB2								3.8%
N N N N N N N N N N N N N N N N N N N	R		15.576 14.855	93.997 97.062	99.492 99.385	CB82 KA82			-14.515	96,141	93.744	L110		
PROJECT (PRESMOORN STRACT)			1100				2 %		-14.010	20,141	04.111			
2	EXIST EXIST			97.348										
22,	EXISTING PAVEMENT	4	10.090 8.714 8.499	97.493 97.617 97.637	99.286 99.258 99.254	LIB4 CFBX CFBX	20%							1.1%
41E	-CARENT		7.000	97.718	99.223	LIB4	20%						-	2
							20%							
	m		-3.500	97.882	99.152	L1BO	2.0%							
	XISTING	400	-0.000	800.88	99,081	LFBI	мсво	425	-0.000	96.747	98,585	MCB0	1	MCBO
8	EXISTING SURFACE	1	1.000	98.039	99,061	EVBI								
•			4.340	98.111	88.604	IMB0	1 1						-	-2.0%
3/10/2018 TITLE DRAN CHECK DESIGN CHECK DESIGN CHECK												10 Jan 10		
XX 00 00 00 00 00 00 00 00 00 00 00 00 0	- Design		8.479 9.500	98.444 98.478	98.919 98.888	EVA1	<u>i</u>		9,500	97.766	98.395	MCAO		-
R. LENCONCOL	DESIGN SURFACE						30%							
salamen	ň		13.000	98.555	98.783	LIAI	30%			97.868				-3.0%
DATE DATE DATE DATE DATE											and the state of t			
NSW A			16.500 18,500	98.452 98.360	98.678 98.616	LEAI KAAO	14 ON		18.500	97.875	98,125	KAAO		
			19.180	98.312	98.728	CBAO	2.0%		19.180	97.864	98.235	CBAO	i	20%
ervice			21.680 22.730	98.804 99.117	98.778 98.799	EXAI EHAI	2.0%		21.680 22.730	97.880 97.928	98.285 98.306	EXAI EHAI		2.0%
n ITATSport			23,885	99.214	99.146	IAAI			23.963	97.993	97.998	IAAI		10
n line														
SPRING FARM FARMIN STAGE 1 ROAD D MENANGLE ROAD ROAD ALIGNMEN CROSS SECTIONS - MEI CROSS SECTIONS - MEI CROSS SECTIONS - MEI													V III 1 - L-L-L-MARKEN	
NMENT														
EVELO I TO MI I AND 2017/	NOT									}		. I	1	
STAGE 1 ROAD DEVELOPMENT MENANGLE ROAD TO MENANGLE PARK ROAD ALIGNMENT AND EARTHWORKS Gassissentons- MENMELEROAD RATHWORKS Instruments DS2017/1000204 Ratholic Control Control Instruments DS2017/1000204 Ratholic Control Con	NOT FOR CONSTRUCTION													
T SLE PA WORK														
RK (S ND (MCB0) - SHEET	USTR													
)-St	0													

THIS DRAWING MAY BE PREPARED IN COLCUR AND MAY BE INCOMP 0 [6 13 15 20 13 145 150mm CM A3 \$22E CRESHW 1 [2] [45] [40] [45] [50mm CM A3 \$22E CRESHW												
BOUNDARY DIGITAL CADASTRE (JP OVERLAY)	ersen ofisen	EXISTING	DESIGN	CUT STRINGS			0++SE	DESIGN	EXISTING	DESIGN LEVELS	CUT STRINGS	
DASTRE ()	및 골 -35.000 -	57 gg 95277		ស្តី [IAB1	I		<u>1</u>	12	SNG	νz	8	. S
1 3 3 1	-31.009				3.7.7 PATUM R.L. 93.0						Normal Participation and an accelerate statement of the second statement of the se	CATUM R.L. 92.6
FEDESTRIAN SAFETY FENCE	-27.042	95.495	98.493	EH85		200						
I SAFE	-22.139 -21.474 -20,598	95.599 95.608 95.628	98,491 98,335 98,360	BFB5 CFB5 KAB5								
P SAFETY BARRIER PROJECT DRAFTIER PROJECT DRAFTIER PROJECT DRAFTIER PROJECT DRAFTIER PROJECT DRAFTIER						1.7%	-16	871 (95-657	95.657	IA81	12
R R	-12.719 -11.546	95.855 95.886	98.227 98.355 98.355	LIB5 CTB3			-12.	680 9	95.749	97.753	EH85	
EXISTING PAREMENT	-11,423	95.688	98,355	CBB3			-9.8 -9.0		95.847 95.860	97.837 97.733	BFB5 KAB5	
PAVEMENT JACOUST	-7.944 -7.000	95.960 95.994	98.357 98.216	CBB3 LI85			-7.0	00	95.972	97.725	LE80	
						1.7%	-3.5	00	95.958	97.719	L180	0.2%
EXISTING SURFAC RC-DT88 ISOURDEADED IF ISOURDEADED IF ISOU											and a rest of the second	0.2%
EXISTING SUPPACE	450 -0.000	96.213	98,098	MCB0		MCBO	475 -0.0 0.6 0.7		98.024	97.714 97.843 97.843	UFBI GTA1 CBA1	MCB0
						-2.0%	2.6 2.9		98.037 98.039	97.803 97.673	CBA1 KFA1	1.1.28
RGTD AV002 DEEXA DEEXA DEEXA DEEXA DEEXA		96.445		1.0000 - 1.000 - 1.0000		2	6.2	00 -	96.359	97.617	L1A0	provide and the second second
DESIGN SURFAC TITLE INVE TITLE INVE TIN	4 544	07.005	03.000	14040			9.5	~	96.694	97,562	LAAD	1.78
DESIGN SURFACE	9.500	97.095	97.908	MCAO			9.0		50.034	51,552		4 15 17
1 Pr			A NATIONAL PROPERTY AND A REAL PROPERTY A REAL PROPERTY AND A REAL PROPERTY A REAL PRO			ង	13.	000	97.047	97.457	LIAI	-3.0%
							16.	500 -	97.157	97.352	LEAI	ے پی
CLEN SCAN SCAN READS AND	18.500 19.160	07.369 97.382	97.639 97.748	KAAO - CBAD		64			97.200 07.195	97.292 97.402	KAAO CBAO	
aav Transport NSW Fransport Roads & Marten NSW Services Rowsand Rowsa	21.680	97.331	97.798	EXAI		2.0% 2.0%			97.175	97.452	EXAI	2.0%
Transport Roads & Martitme Services where Rolect OFFCE	22,730	97.278	97.819	EHAI		4.		-	97.163 97.069	97.473	- EHAI IAAI	
SS CE TIPE	25,240	97.192	97.192	IAAI	1							:
			10.011									
STAGE 1 ROAD DI ROAD ALIGNARIA ROAD												
N(INCOMPTONIC INCOMPTONIC INCOMPTONIC INCOMPTONIC INCOMPTONIC												
NOT FOR CONTRACT NOT FOR CONTRACT NUT FOR CONTRACT NO DEVELOPMENT OD DEVELOPMENT OD DEVELOPMENT NUT FOR MENTAL NEW CONTRACT DS2017/10000000000000000000000000000000000												
NOT FOR CONSTRUCTION												
CONSTRUCTION CITS CEPARK WORKS OTHEOWNO (MCER) - SHEET 10 04												

P SAFETY BARRIER EXISTING FAILuEAT Annu-contracting barrier barrier for the formation of th			NAME AND ADDRESS OF TAXABLE AND ADDRESS OF TAXABLE ADDRESS OF TAXAB		DATUMRI 910			EXERN EXERNE	CUT STRINGS	
REAL PARTY AND LANG LANG LANG LANG LANG LANG LANG LANG	-18.004	94.601	94.601	- IAB1	2.02A	-18	.043 94.97	3 04.973	1AB1	15:
A K	-12.680	94.689	97.263	EHB5		-12	1.683 95.02	3 07.126	EHB5	3.0%
	-9.880 -9.000	94.846 94.870	97.347 97.243	BFB5 KAB5			883 <u>94.94</u> 003 94.92		BFB5 KAB5	च दः र
	-7.000	84.941	97.284	LEBO	2 2 2 3 4	-7.1	000 94.84	8 - 97,168	LEBO	్
	-3.500	95.042	97.361	L180	-2.2%	-3.	500 94.77	2 97.275	L180	<u>ئ</u> ا 1
SUC SUBSCIENCES		95.119 95.207	97.438 97.579 97.579	LF81 CTA1 CBA1	<u>ж</u> <u>МС80</u>	525 0.0	000 84.65 70 84.65 40 94.65	96 97,382 15 97,527 17 97,527	LF8I CTA1 CBA1	MCB0
SURFACE	0.740 2.560 2.900	95.218 95.469 95.600	97.579 97.582 97.452	CBA1 CBA1 KFA1		0.7 2.6 2.8	57 94.70	3 97.579	CBA1 CBA1 KFA1	
PROTUNER 37102018 BROWN BROWN BROWN	6.200	95.934	97.419	L1A0	-1.0%	6.1	197 94.81	19 97.416	LIAO	
DESIGN SURFAC THOTOME: INC. STOCOMPC: INC. STOCOM THE STOCOM STO	9.500	96.355	97.386	LAAO	2017 2017	9,4	197 95.5	79 97.383	LAAO	
DESIGN SURVEY MUSC MUS					50 028					300%
CE stabilities sta	13.000	96.741	97.281	L1AI	20 20 20	12.	.997 - 98.4	56 07.278	LIAI	30%
	16.500 18.524	97.038 97.112	97.176 97.116	LEAI KAAO	<u>ئې</u>	18.	,497 97.0	79 97.173	LEAI	4.30%
	19.204 21.705	97.137	97.226	EXAL	13%		.649 97.14		ESAJ EHAJ	-2.3% 1
Aler NSW Transport Roeds & Maritime Services Trainagrow Westersy Stokey ProJect OFFICE	22.755 23.466	97.096 97.081	97.260 97.031	EHAI		22	.329 97.1	70 97.170	- IAAI	A:
					ī					
CANCENARD CAMPER SPENS FARM PARAMA SPENS FARM PARAMA STAGE 1 ROAD D MENANGLE ROAD MENANGLE ROAD ALIGNMERT MERCENTRAL MERCENTRAL MERCENTRAL							And and a set of the 	 A characterization of the second secon		
NOT FOR CO									<u>I</u>	
NOT FOR CONSTRUCTION										
CONST SONST PARK										

DESIGN	EXISTING	DESIGN	STRINGS	DATUM R.L. 220			DESIGN	EXSTING LEVELS	LEVELS	CUT STRANS	DATUM RL 93.0	
							-18.018	98.178	96.178	IAB 1	·	
-15.457 -11.500 -9.300	96.210 96.392 96.410	98.210 97.200 97.266	EHB1 ESB1	4.01	- - 		•10.625	96.354	97.526	EHB1	4.01	
-6.300	96.459	97.358	LEBO					96.420	97.592	ESBI	4 9 2	
-3.500	96.505	97.440	LABI					98.503 98.557	97.682	LAB1	and the second second	
0.000 0.670 0.740 2.598 2.838	90.572 90.588 98.590 96.623 96.628	97.545 97.620 97.690 97.747 97.747 97.617	LFBI CTA1 CBA1 CBA1 CBA1 KFA1		MCBO			96.612 96.625 93.627 93.681 95.687	97.845 97.990 97.990 98.038 97.909	LFBI CTA1 CBA1 CBA1 CBA1 KFA1	54 MCB0	-
6 138	96 972	97 583	LIAO								-1.0%	
9.438	96.569	97.550	LANO		-		9.304	97.013	97.845	LAAO		
12.938	97.081	97.445	L1AJ				12.604	97.445	97.739	LIAI		
18.438	97.292	97.340	LEAJ				16.304	97.568	97.634	LEAI	40%	
18.938 20,438 20,866	97.337 97.334 97.327	97.265 97.220 97.327	ESAJ EHAI IAAI		30% A.D.1		18.805 20.305 20.575	97.600 97.584 97,581	97.550 07.514 97.581	ESAI EHAI IAAJ	407 4 07	
	-15.457 -11.500 -9.300 -6.300 -3.500 0.670 0.740 2.638 6.133 9.438 12.938 13.438 13.438 13.938 20.438	-15.457 96.210 -11.500 93.352 -9.309 93.410 -6.300 96.459 -3.500 96.505 0.700 96.553 0.700 96.523 0.700 96.523 0.700 96.523 0.701 96.523 0.702 96.523 0.703 96.563 0.704 96.563 0.705 96.563 0.706 96.563 12.938 97.081 13.438 97.292 13.938 97.337 20.438 97.334	-15.457 96.210 96.210 96.210 -11.500 96.392 97.200 -9.300 96.410 97.283 -6.300 96.459 97.356 -3.500 96.505 97.440 0.707 96.526 97.545 0.370 96.505 97.440 0.607 96.505 97.650 96.628 97.671 96.628 97.671 6 133 0.6.232 97.550 97.650 9.438 96.569 97.550 12.938 97.081 12.938 97.081 97.445 13.436 97.292 13.438 97.292 97.340 13.438 97.292 97.340	-15.457 96.210 93.210 AB1 .11.500 96.392 97.200 EHB1 .9.300 96.312 97.263 ESB1 .6.300 96.459 97.358 LEB0 .3.500 96.505 97.440 LAB1 0.700 96.505 97.450 LFB1 0.300 96.505 97.440 LAB1 0.700 96.528 97.570 LFB1 0.740 96.528 97.747 CBA1 2.598 96.628 97.747 CBA1 2.598 96.529 97.550 L1A0 9.438 96.529 97.550 LAA0 9.438 96.529 97.550 LAA0 12.938 97.031 97.445 L1A1 13.438 97.292 97.340 LEA1 13.938 97.337 97.265 ESA1	-15.457 96.210 93.210 JAB1 -11.500 96.392 97.200 EHB1 -9.000 96.410 97.288 ESB1 -6.300 96.459 97.358 LEB0 -3.500 96.505 97.440 LAB1 -0.001 96.572 97.545 LFB1 -0.3500 96.623 97.747 CBA1 -0.701 96.628 97.600 CTA1 0.670 96.628 97.600 CBA1 2.599 96.628 97.617 CBA1 2.599 96.628 97.650 LAM0 9.438 96.569 97.550 LAM0 12.938 97.292 97.340 LEAI 13.438 97.292 97.340 LEAI 13.439 97.297 97.285 ESAI 20.868 97.327 97.327 WA1	15.457 96.210 98.210 JAB1 -11.500 96.322 07.200 EHB1 -11.500 96.322 07.200 EHB1 -9.200 63.410 97.263 ESB1 -6.300 96.459 97.358 LEB0 -3.800 96.505 97.440 LAB1 -0.700 80.529 97.550 LFB1 -10 92.599 97.550 LFB1 -0.700 96.572 97.545 LFB1 -10 92.599 97.550 LFB1 -11.500 96.623 97.747 CBA1 2.599 96.628 97.7617 KFA1 -10 92.599 97.590 LIA0 2.598 96.628 97.617 KFA1 -13 96.722 97.583 LIA0 -13 96.723 97.590 LAN0 -13 96.724 97.583 LIA0 -143 96.593 97.590 LAN0 -143 <td>-15.457 98.210 98.210 M81 -1</td> <td>D D -15.457 66.210 95.210 J81 -16.018 -11.500 95.392 07.200 EH91 -16.018 -11.500 95.392 07.200 EH91 -16.018 -0.520 03.410 97.268 ES91 -16.018 -0.520 03.410 97.268 ES91 -16.018 -0.520 95.505 97.400 LAB1 -16.210 -5.428 -0.700 95.558 97.600 CTA1 -16.018 -5.428 -0.700 95.558 97.600 CTA1 -16.010 -5.278 96.623 97.610 CTA1 -16.210 -5.278 -5.428 94.38 96.559 97.600 CTA1 -5.284 -5.284 -2.847 9.438 96.559 97.550 LAM0 -5.284 -5.284 -5.284 -5.284 -5.284 12.2938 97.081 97.345 LIA1 -5.284 -5.235 -5.304 -5.335 -5.304</td> <td>-15.457 95.210 95.210 M81 -15.457 95.210 95.210 M81 -11.500 95.332 97.200 EH91 -11.500 95.332 97.200 EH91 -11.500 95.332 97.339 LES0 -5.30 95.555 97.460 LAS1 -5.30 95.555 97.460 LAS1 -5.31 95.557 97.450 LS1 -5.32 95.555 97.460 LAS1 -5.32 95.555 97.460 LAS1 -5.33 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.430 96.577 97.550 LAA0 -1.430 96.577 97.550 LAA0 -1.430 96.577 97.550 LAA0 -1.430 96.581 97.550 LAA0 -1.430 96.581 97.455 L1JJ -1.430 96.581 97.455 L1JJ -1.430 96.581 97.455 L1JJ -1.430 97.555 LAA0 -1.430 97.555 LAA0 -1.430 97.555 P7.465 L1JJ -1.430 97.555 P7.465 L1JJ -1.430 97.555 P7.465 L1JJ -1.430 97.555 P7.465 P7.455 L1JJ -1.430 97.555 P7.665 P</td> <td>In In In -15.457 62.210 92.210 M81 -16.016 93.178 94.178 -11.500 92.322 07.500 EH81 -16.016 93.178 95.178 -11.500 92.322 07.500 EH81 -16.016 93.178 95.178 -3.000 94.410 97.528 ES51 -10.655 95.400 97.592 -3.000 94.593 97.555 IF81 -10.655 95.503 97.592 -3.000 95.505 97.400 L481 -10.655 95.503 97.892 -3.000 95.505 97.400 L481 -10.655 95.503 97.892 -3.000 95.505 97.401 L481 -10.655 97.690 97.790 -3.000 95.505 97.401 L481 -10.655 96.632 97.890 -3.000 97.997 0.997 0.997 0.997 0.997 0.997 -3.000 97.691 0.997 0.997</td> <td>D D D D D D -15.457 95210 92210 92210 92210 9210</td> <td>B B</td>	-15.457 98.210 98.210 M81 -1	D D -15.457 66.210 95.210 J81 -16.018 -11.500 95.392 07.200 EH91 -16.018 -11.500 95.392 07.200 EH91 -16.018 -0.520 03.410 97.268 ES91 -16.018 -0.520 03.410 97.268 ES91 -16.018 -0.520 95.505 97.400 LAB1 -16.210 -5.428 -0.700 95.558 97.600 CTA1 -16.018 -5.428 -0.700 95.558 97.600 CTA1 -16.010 -5.278 96.623 97.610 CTA1 -16.210 -5.278 -5.428 94.38 96.559 97.600 CTA1 -5.284 -5.284 -2.847 9.438 96.559 97.550 LAM0 -5.284 -5.284 -5.284 -5.284 -5.284 12.2938 97.081 97.345 LIA1 -5.284 -5.235 -5.304 -5.335 -5.304	-15.457 95.210 95.210 M81 -15.457 95.210 95.210 M81 -11.500 95.332 97.200 EH91 -11.500 95.332 97.200 EH91 -11.500 95.332 97.339 LES0 -5.30 95.555 97.460 LAS1 -5.30 95.555 97.460 LAS1 -5.31 95.557 97.450 LS1 -5.32 95.555 97.460 LAS1 -5.32 95.555 97.460 LAS1 -5.33 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.439 96.623 97.47 -1.430 96.577 97.550 LAA0 -1.430 96.577 97.550 LAA0 -1.430 96.577 97.550 LAA0 -1.430 96.581 97.550 LAA0 -1.430 96.581 97.455 L1JJ -1.430 96.581 97.455 L1JJ -1.430 96.581 97.455 L1JJ -1.430 97.555 LAA0 -1.430 97.555 LAA0 -1.430 97.555 P7.465 L1JJ -1.430 97.555 P7.465 L1JJ -1.430 97.555 P7.465 L1JJ -1.430 97.555 P7.465 P7.455 L1JJ -1.430 97.555 P7.665 P	In In In -15.457 62.210 92.210 M81 -16.016 93.178 94.178 -11.500 92.322 07.500 EH81 -16.016 93.178 95.178 -11.500 92.322 07.500 EH81 -16.016 93.178 95.178 -3.000 94.410 97.528 ES51 -10.655 95.400 97.592 -3.000 94.593 97.555 IF81 -10.655 95.503 97.592 -3.000 95.505 97.400 L481 -10.655 95.503 97.892 -3.000 95.505 97.400 L481 -10.655 95.503 97.892 -3.000 95.505 97.401 L481 -10.655 97.690 97.790 -3.000 95.505 97.401 L481 -10.655 96.632 97.890 -3.000 97.997 0.997 0.997 0.997 0.997 0.997 -3.000 97.691 0.997 0.997	D D D D D D -15.457 95210 92210 92210 92210 9210	B B

i.



THIS DRAWING MAY BE FREPARED IN COLUME AND MAY BE INCOMPLET Image: Im	DESCR	EXISTING LEVELS	DESIGN	CUT STRINGS	DATUMAL 930		DESIGN OFFSET	EXISTING	DESIGN	CUT STRINGS	DATUM RL 540
P SKETY ENORER EXISTING PARENT EXISTING SURFACE Provide seasoes converting Incorrect and the provide seasoes converting Source for the provide seasoes converting Intervide seasoes converting Not converting to the provide seasoes converting Not converting to the provide seasoes converting to t	-13.598 -8.618 -6.546 -3.500	56 753 96,910 96,950 97,008 97,151	96.753 96.004 98.061 98.152 98.257	IAB2 EH03 MZB1 LE84 LF81	4.127 -2.7% -3.0% -3.0% -3.0%	675	-13.632 -8.700 -6.500 -3.500 8.000	98.651 98.800 98.885 98.993 101.385	97.884 97.950 98.040 93.145 99.255	EH83 ES82 LERI LF8I	4.07. 4.07. 4.07. 4.07. 4.07. 1.07. 1.07.
DESIGN SURFACE CLANK VALIDALI INE Koli IN VALIDALI IN Koli IN VALIDALI INFORMATI IN Koli IN <td>4,085 8.146 11,647 15,116 17,818 19,119 19,497</td> <td>97.372 97.659 97.092 98.007 97.068 97.665 97.635</td> <td>98.379 99.257 99.153 98.049 97.974 97.929 97.835</td> <td> (MAQ) LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ</td> <td>30% 30% 30% 40g</td> <td></td> <td>7.391 13.656 18.162 17.685 17.983</td> <td>97.819 97.895 97.852 97.759</td> <td>98.145 97.958 97.883 97.833 07.759</td> <td>LFAO LEAI ESAJ EHAI</td> <td>30% -30% -30% -207</td>	4,085 8.146 11,647 15,116 17,818 19,119 19,497	97.372 97.659 97.092 98.007 97.068 97.665 97.635	98.379 99.257 99.153 98.049 97.974 97.929 97.835	(MAQ) LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ LFAQ	30% 30% 30% 40g		7.391 13.656 18.162 17.685 17.983	97.819 97.895 97.852 97.759	98.145 97.958 97.883 97.833 07.759	LFAO LEAI ESAJ EHAI	30% -30% -30% -207
NOT FOR CONSTRUCTION CAUDEN AND CAMPELLITOWN CITY COUNCILS SEPARGE FRAN PARYWAY TAGE 1 ROAD DEVELOPMENT ROAD ALLONMENT AND EARTHWORKS COSSSECTIONS - MERVINGLE PARK AND ALLONMENT AND EARTHWORKS COSSSECTIONS - MERVINGE PARK INSTRUMENT AND EARTHWORKS CONSESSECTIONS - MERVING - PARK INSTRUMENT AND EARTHWORKS INSTRUMENT - MERVING - PARK INSTRUMENT - MERVING - MERVI											



0 5 10 13 20 24 10 43 65m OHAD2E OFFICE 0 5 10 14 20 24 10 43 65m OHAD2E OFFICE 1 10		DESIGN LEVELS	STRINGS	EXISTING LEVELS DESIGN OFFSET	Sur Strangs Design Levels	DATU
DIGITIN, CADINSTIRE (DP OVERLAN) PEDESTRIAM SAFETY FENCE P SAFETY BARRIER	-12.621 965	.00 96,500	NB3			DATUM RL SSO
EDUSTING PANEMENT EXISTING SUBFACE	-5.200 959 -6.000 97.1 -3.500 97.2 2.069 97.6 4.146 97.6	159 97.605 117 97.671 125 97.746 137 97.851 1305 97.914	EH84 4	-12.100 96.618 -8.200 - 96.954 -6.000 97.22A -3.500 97.523 75 0.000 97.523 0.000 97.687 1.413 97.708 2.878 97.721 6.333 97.708	96.618 AB3 97.593 EHB4 97.599 ESB3 97.734 LEB1 97.839 LFB1 97.839 LFA0 97.734 LEA1	A 0.7 3.0% 3.0% 3.0% 3.0% 3.0% 3.0%
DESIGN SURFACE	7.876 97.4 10.150 97.4 11.852 97.3 12.046 97.3	512 97.671		8.837 97.553 10.339 97.525 10.844 97.489	97.859 ESAI 97.814 EHAJ 97.488 IAAI	50% 407
NOT FOR CONSTRUCTION CANDEN AND CAMPBELLI JOWN CITY COUNCILS STAGE IN CONSTRUCTION CITY COUNCILS STAGE IN COUNT COUNCILS STAGE IN COUNCILS STAGE INCLUMENT AND EARTHWORKS SUBJECT TO MENT AND EARTHWORKS <						



	DESICN	EXISTING	DESIGN	CUT STRMCS			DESIGN OFFSET	EXISTING	DESIGN	CUT STRAKES	
ON CONTRUCT FENCE		ŭ			0470MRL 930						DATUM RL 940
TY BARRER EXISTING PAYEJENT PAGET and an and the second at	-12.102 -8.200 -8.000 -3.500	98.228 98.228 97.897 88.007	96.846 97.822 97.888 97.963	LAD3 EHB4 ESB3 LEDI	407 - 30% - 30%		-11.106 -7.971 -5.771 -3.500	97.228 97.727 98.049 98.168	97.228 98.013 98.079 98.147	IAB3 EH84 ESB3 LEBI	ADT 30% 30% -30%
850 ANSTING SUPPACE ANSTING SU	-0.000	98 082	98.068	UFB)	3 ⁸ <u>NCB0</u>	. 875	0.000	98.254	98,252	LFBI	KCED
n aust aust aust aust aust aust aust aust					- 1 - - - -						
CHUREN AND CHARGELLTOWN CITY COUNCLS A3 STARE 1 ROAD DEVELOPMENT A3 STAGE 1 ROAD DEVELOPMENT A3 STAGE 1 ROAD DEVELOPMENT A3 ROAD ALIGNMENT AND EARTHWORKS COSS SECTIONS MEMANGLE PARK RUSS SECTIONS MEMANGLE DON DEVELOPMENT BEAMAGLE ROAD TO MEMANGLE PARK MEMANGLE ROAD TO MEMANGLE PARK COSS SECTIONS MEMANGLE PARK COSS SECTIONS MEMANGLE DON DEVELOPMENT EARTHWORKS MEMANGLE ROAD TO MEMANGLE PARK EARTHWORKS COSS SECTIONS MEMANGLE PARK EARTHWORKS CONCEPT DESIGN EARTHWORKS CONCEPT DESIGN EARTHWORKS											







THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED 10 | 5 | 10 | 15 | 20 | 25 | 30 | 35 | 40 | 50mm (CH X) 028 0 FROMUL



THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED 10 | 5 | 10 | 15 | 20 | 25 | 20 | 35 | 49 | 45 | 50mm CHAI \$22 ORDAUL

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED |0 |5 |13 |32 |32 |33 |35 |32 |32 |33 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35 |35</td







THIS ERAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETED OFFICE 0 | 5 | 16 | 15 | 25 | 25 | 31 | 35 | 43 | 45 | 35mm of Al \$285,000 AU





THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED
0 |5 |10 |11 |20 |25 |30 |31 |40 |40 |40 |40mm ON AS SZE ORIGAW











THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED 9 | 5 | 10 | 15 | 21 | 23 | 23 | 24 | 35 | 16 | 16 | 16 m of an 32 20 organiz







15 10 15 20<	CROSSEALL LEFT DESIGN LEVELS EXISTING LEVELS	CATUM RL 740 VERTICAL ALIGINAENT HORIZONTAL ALIGINAENT	
	300,000 88/64 100,248	14-2290 / IVA VC=152200	
55 10 10 10 10 10 10 10 10 10 10	VTP 319.204 99.396 320.000 = 86.098 99.362 = -	STEAIGHT	
ach i fran ann an ann ann ann ann ann ann ann a	340,000 67.519 98,400 HTP 344,578 98,180		
	360.000 87.216 97.438		
	VTP 371.946 96.864		
RECONTINUE 25 25 25 25 25 25 25 25 25 25 25 25 25	380.000 66.777 96.507		IP STN 386.946 RL 90.142
SACTEL INFORMACIAN ELEVERATION	LONG TUDINAL SECTION VTP 401348 86.620 85.881 420.000 86.059 85.537 55.479 55.479 55.479 55.479 56.479 57.572 57.5		NB ENTRY RAMP
	CO CT CT VTP 417.188 420.000 86.059 85.479 85.479 85.479		····
Junces com IA186400-1 ESSANSAD	RING FJ 440.000 85.572 94.972 RR M D D	K= VC=55 955	IP STN 450.688
A CALL AND	RKWAY 460.000 84.922 94.208		// RL 94.867
	480.000 84.197 93.457 WTP 434.187 93.259	R=50000	
A BYG285 X D14680000 A D14680000 A D14680000 A D14680000 MONE MONE	500.000 84.212 92.501		
nor er srehmen an uzen an uzen an uzen an szen an szen	520.000 83.251 91.541	C	/
	540.000 83.105 90.581		
Transport Roads & Maritime Services	560.000 B2.838 89.821		
8	VTP 570.833 89.102		/
CANDEN AND CAMPBEL SPENIS FARM PARKW STAGE I ROAD D MENANGLE ROAD ALIGNMEN ROAD ALIGNMEN ROAD ALIGNMEN RECEITING DATES	580 000 82.317 88.680	K-2251 / 100.	
	800.000 <u>84.209</u> 87.891		
NOT FOR CONSTRUCTION			



THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCO [0] [5] [15] [15] [15] [16] <t< th=""><th>DAMEETE IF CORED</th><th></th></t<>	DAMEETE IF CORED	
a b <td></td> <td></td>		
More and		
Maria Julia Maria Maria Casar Image www. www. www. Image source www. Image Image www. www. Image Image wwwwwwww. <td< th=""><td></td><td></td></td<>		
NOT FOR CONSTRUCTION		









THIS DRAWING MAY BE PREPARED IN COLOUR AND 0 15 10 115 20 23 30 53 40 45 55	MAT BE INCOMPLETE IF COM	-1511					
UNAMOTIFIC COLUMN NAME FIG	STATION	EXISTING LEVELS	DESIGN LEVELS	CROSSFALL LEFT	HORIZONTAL ALIGNMENT	VERTICAL ALIGNMENT	DATUM RL 75.0
E : TRIE SUBDO-RDI THIS DRAW	VTP HTP 0.000	86.393	95.602				IP STN 0.000 RL 95502
10-11E : TINE - ACJ 07 - XY02/19 In2-do M - EXTLan NUCANO DO LANZA - XY02/19 In2-do M - EXTLan CU - Hig Downood (Ref) 10-11E 11 - Hig Downood (Ref) 10-1				ZUUP,	- 1991 - 1 (an an a		
07 3142 8142, 416 8142, 416	20.000	65.89 2	94.700		STRAIGHT		
N 102718 102 KSULED FOR KSULED FOR		reaction within the property			H		
4:56 AM 17 REPUBLICATION 30% CONCEPT	40.000 HTP 43.983	84.891	93.898 93.738			G=-4.010% L=68.828	
- soperation							
8	60.000	84.646	93.096				
	VTP 68 828		92.742			H	
Soorthur Soorthur Soorthur Viteraus Thinks	60.030	85,458	92.315	4 507	R		
NGC PERFORMANCE	5				R=110,000		
Hodat: Inforcement	LONGITUDINAL SECTION NORTHBOUND ENTRY RAMP MC40	85.378	91.659				
	ECTION 120.000	85.709	91.141				
	NO HTP 124.391		91.046	ALL L			
IA1854	HBOUN 140,000	86.089	90.761				
	D ENTF				SIKAUGHI		
Safar NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA NZA	Y RAM 160.000	85.627	90.519		3	á	/
	P MC4						
³ Ø	HTP 173.711 180.000	87.139	90.432 90.414				/
arcon desc unter desc desc desc desc desc desc desc desc						WC=270,000 K=29017100	STN 185.160 RL 90.410
AND INTELLINE AND IN	200.000	87.453	90.448			88	
A HILIHOUTA	200.000	07.455	30.440				jip STN 203 821 (RL 67.329
PLOI IF							
H 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	220.600	87.750	90.619				
					7	0-16	
	240,000	87.873	90.928	3.00%	N-HOLVOO	200	
Roads & Marthme NSW Services Roads & Marthme Services Roads & Marthme Roads Roads & Marthme Roads Roads & Marthme Roads & Mart							<u> </u>
	260.000	87.851	91.375				
5							
CANDEN AND CAMPBE SPRING FAMIL PARKWA STRAGE 1 ROAD ROAD ALIGE 1 ROAD ROAD ALIGNER ROAD ALIGNER	280.000	86.922	91.960	+			
ND CAMP TRANSPARE TRANSPARE INAL SEC TRANSPARE EPT DESI							
NO DEVE NO DEVE	300.000	87.022	92,682	1.11			ПЦ
NOT FOR CONCLEASE PRELITOWN CITY COUNCILS WANN TO DEVELOPMENT D D D D D D D D D D D D D D D D D D D							
DR CC							
CONSTRUC CONSTRUCT LIS LORKS LORKS LORKS LORKS RD-02 R							
NOT FOR CONSTRUCTION							
at the second							

0 6 13 15 23 75 13 35 10 15 5m	YA ON AJ SIZE CARGINAL						
DAADD NEEDDATRIK NAME VARRAYAN BANKINSKI VA UDANTAN KRON LUDG, REShinski VA UDANTANTAN	STATION	EXISTINGLEVERS	DESIGN LEVELS	CROSSFALL LEFT	HORIZONTAL ALIGNMENT	VERTICAL ALIGNMENT	DATUM RL 780
TEALET THE	300.000	B7.022	92.632			/	
20 PR	320.600	87.693	93.543			VC=270.000 K=2901.100	
101 155-001 101 101 101 101 101 101 101 101 101							
BN www. 12-12	VTP 338.828 340.000	87.101	94.479 94.541			$\frac{1}{1}$	
Solution Science Science	360.000	68.947	95.601				
					R=-480.000		
Secondary Secondary Tradition	380.000	90,896	96,660		Ϋ́Ω		
		93.000	97.719			L=111.424 G=5 297%	
Server a conversion structure of the server	LONGITUDINAL SECTION NORTHBOUND ENTRY RAMP MC40			3 00 E			
	CTION 420,000	94.699	98.779				
2 3	ORTHBOUL HTP 439.158 440.08	96.089	99.793 99.838				
	ND EN VTP 450,253		100.361				
	Y RAMP 460.000	97,425	100.890				
		98.001	101.889				
A DE GATE TRE 3 TUDOTE TRE 1002/018 2005/42	480.000	30.001	101,003				
TE FINE VILLE VILE	500.000	99.529	102,826				
PCDT 07 Srahman 201951 201951 10005					SI		
0A12 20102211 20102211 20102211 20102211 20102211 20102211	520.000	100.814	103.702		STRAIGHT	K=-8512/100 VC=145.000	IP STN 622.753 T RL 104.221
NEW Transport	540.000	101.664	104.516				
Transp Roads & Services Services							
Transport Roads & Maritime Services	560.000	102.779	105.269				
	580.000	103.744	105.981				
SANDERAND CHIPPEL STANDERAND CHIPPEL STANDE FRANK PARKMA STARGE 1 ROAD DI STAGE 1 ROAD STARGAN ROAD ALGANNES CONCEPT DESIGN STARGAN CONCEPT DESIGN	VTP 595.252		106.447	3,00%	SAUG C		ହ <u>ଳ</u>
NOT FOR CONSTRUCTION	600,000	104.900	108.592				
FOR C IT COUNCIL							
CONSTRUC CILS LE PARK MIRY PAMP MCAD- MIRY PAM							
NOT FOR CONSTRUCTION TOWNOTF COUNCES EVELOPMENT TAND EARTHYDRXS 1- NORTHBOLIND ENTRY PANAP (MCL40) - SHEET 2 OF 20177/000204							
						<u></u>	

THIS DRAWING MAY BE PREPARED IN COLCOR AND MAY BE INCOMPLETE IF COPIED 10 15 115 115 115 120 125 130 125 140 145 150mm0NAISZECKGANL

1 660.000 107.543 108.434 109.844 660.000 107.543 108.434 109.845 100000000 1000000 1000.339 109.846 1000000000 1000.474 109.845 10000000000 1000.474 109.845 100000000000 1000.474 109.845 100000000000000 110.216 110.276 1000000000000000000000000000000000000
Image: second
Image: series of the
NOTE: Note: <th< td=""></th<>
780.000 111.979 112.111 000000000000000000000000000000000000
780.000 111.979 112.111 0000ceex 500.000 112.699 112.689 111.070 112.699 112.689 112.689
780.000 111.979 112.111 900/0657 1000/016 112.599 112.689 112.000 112.599 112.689 112.699
780.000 111.979 112.111 1000000000000000000000000000000000000
788.000 111.979 112.111 000000000000000000000000000000000000
780.000 111.979 112.111 1000000000000000000000000000000000000
Bit Dig Prof UKEE Source Imple Source Source
XX XX<
820.000 113.171 113.237
Bit Delim Bit Delim Bit Delim Bit Delim Bit Delim IP STN 835.002 Rutaria
840.000 113.698 113.754 113.754 113.897 SK 70 TT HTP 845.634 113.698 113.754 113.691 7 SK 70 TT HTP 854.028 114.099 88 114.099 88 114.099
Strating HTP 645.834 113.604 R HTP 654.028 114.099 8 St & ACO 880.000 114.204 114.211
B80.000 1114.072 114.697 CONDECTIONAL CONSTITUTION OT FOR CONSTI STAGE FASH CONDITION OT FOR CONSTI STAGE FORD DELLOWINGTY COUNCLS STAGE FORD TO MENANCLE FOR CONSTI STAGE FOR CONST STAGE FOR CONST STAG
900.000 115.103 116.124
CONNECT 10 1000 1000 1000 1000 1000 1000 1000

10.111/16 10.011/17.021-00.0010 10.011/17.021-0000000 10.011/17.000000000000000000000000000000
1977 - 2010/2011 01/2010 AM - 20Ane 2027 - ANDONALOT I REVISION RESORTING 2027 - ANDONALOT I REVISION RESORTING 2027 - 2021 - 2021 - 2021 - 2021 - 2021 2021 - 2021
NACEAL CELENA Niceal Celena Niceal Celena
PHONE STORE 10
Contraction of the second seco
ANDIA COLORADO VALCAR NA 1850/LO CO (D. 172) DOWNIC I DESCHI FOUNDUIT DOWNIC I DESCHI FOUNDUIT NOT PLACE (NY MER. 1717 PLACE (
ACTIONNE THE JACONNE THE JACONNE SEALUTION DESIGNATION SEALUTION DESIGNATION SEALUTION DESIGNATION SEALUTION DESIGNATION ANTI-ACCE
ru) er srahman
New Services
ಹ
NOT FOR CO
NOT FOR WY WY DEVELOPMEN DEVELOPMEN DEVELOPMEN NT AND EART NT A
ONS PARK PARK PARK
MC401-SHEET 4 OF 5 MC401-SHEET 4 OF 5 MC402-SHEET 4 OF 5 MC402-SHEET 4 OF 3 MC402-SHEET 4 OF 3 MC402-SHEET 4 OF 3 MC403-SHEET 4 MC403-SHEET 4 MC403-SHEE

THIS DRAVING MAY BE PREPARED IN COLOUR AND 10 [5] 10 [15] 20 [25] 20 [45] 41 [45] 50	D MAY BE RECOMPLETE IF COPIED	
ONVINCE/RELECCIDENTANCE - FOLLANE TIME - FOLLANE DEXTRUST-South REPORT TO CONTRACT CONTRACT - FOLLANE SECONDOR - FEEDER FOR CONTRACT - FOLLANE SECONDOR - FEEDER FOLLANE SECONDOR - FEEDER FOLLANE SECONDOR - FEEDER FOLLANE SECONDOR - FEEDER SECONDOR - FOLLANE SECONDOR - FOLLANE		
APPROV MAXCT INJUNITY MUSCI DESCRIPTION VILLOOF VILLOOF VILLOOF VILLOOF VILLOOF </td <td></td> <td></td>		
Mol unit: The Worth Case 3H02018 uniter International Stateward International Stateward 00004000 Stateward Stateward Stateward 00004000000 Stateward Stateward Stateward 000040000000 Stateward Stateward Stateward 0000400000000000000000000000000000000		
NOT FOR CONSTRUCTION CONDENT FOR CONSTRUCTION GROUDENT FOR CONSTRUCTION STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT ROAD ALIGUE ROAD TO MENANCIE PARK ROAD ALIG		

15 15 15 23 23 30 35 45 45 10 5 5	SOME ON AS SIZE ORIGINAL					
Annual Control (1996) - (1966) - (1966) - (1966) - (1966)	STATION	EXISTING LEVELS	DESIGN LEVELS		VERTICAL AUGNMENT	DATUM RL 550
	HTP 0.000 VTP	1.965 101	1528	00%		P STN 0.000 RL 101.528
PLGT BW PLGT BW PLG	20.060 97	12.023 100	0.928		1=39.877	
Reaction loss extended and reaction loss of the rea	VTP 39.877 40.000 - 92	100	0.332 0.328 -	1	L	5P. P SIN 44 877
DF DESCRIPTION	VTP 49.877	100	0.025	11UIA	2	P STN 44.877 RL 100.162
к К	60.000 9	1,916 - 99.	.708			
vidthona Vidthona Vidthona Vidthona Vidthona		19.792 93.	1.082	3.00%	L=77	
HORC T BREADOWN SIT	LONGITUDINA	XQ.172 98.	1.456	e ^x	L=11 174	
	EC 120.000 9	91.611 97	7.830	3		
	VTP 127.851	97.	1.591	3.00%		
	140.000 9 OUND	91.586 97	7.239		STRAIGHT	
	LONGITUDINAL SECTION - SOUTHBOUND EXIT RAMP (MC50)	91.276 96	5.815		3HT	
		91.136 96	3.573			
Acronation and account of a second account of	200.000 9	91,135 96	6.512	2.98%	K=220/ /100	주
PLOT BY	220.000 9	91.897 96	5.632			
	240.000 \$	92.691 98	8.934			
REARI REARIE ITANSPORT ROADE & Maritime ROADE & Maritime REARIEST OFFICE REARIEST OFFICE REARIEST OFFICE	260,000 \$	93.932 97	7.416 -			
Aaritime	VTP 272.651		7.815	3.02%		
			a.068			
CONTIGENTAND CAMPEEL STARE 1 ROAD MENANGLE ROAL MENANGLE ROAL MENANGLE ROAL MENANGLE ROAL MENANGLE ROAL MENANGLE ROAL MENANGER THEORY	HTP 291.745		8.472	7-1	G=3,440%	
NOT FOR CONSTRUCTION GAMPBELITOWN OTF COLVCIS STATE & TRANCARWAY STATE 1 ROAD DEVELOPMENT MENANGLE ROAD TO MENANGLE PARK NGAD ALGENRENT AND EARTHWORKS INFORMATIONAL SECTION OF A CONSTRUCTION INFORMATION SOUTHOOD EARTHWORKS INFORMATION SOUTHOOD EARTHWORKS INFORMATION SOUTHOOD EARTHWORKS INFORMATION SOUTHOOD EARTHWO IN THE INFORMATION INFORMATION SOUTHOOD EARTHWO IN THE INFORMATION INFORMATION IN THE INFORMATION IN THE INFORMATION IN THE INFORMATION INFORMATION IN THE INFORMATION INFORMATION IN THE INFOR	300.000	95.204 96	6.756	R- 100,000	50.57	

•

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED 10 15 1/3 1/3 1/3 1/3 1/3 1/3 1/4 1/45 1/5/mr (MA) 525 (RIGHML

THIS DRAWING MAY BE PREPARED IN COLOUR AN	D MAY BE INCOMPLETE IF CO Tres ON ALQUE OR GRAL	PIC()					
Supervision - Litocyliger, even	STATION	EXISTING LEVELS	DESIGNLEVELS	CROSSFALL LEFT	HORIZONTAL ALIGNMENT		DATUMARL 87 3
STELLES STELLES STELLES STATE	300.000	95.204	98.758			\prod	
62 102 102 102 102 102 102 102 102 102 10	320.000	96.924	99.444	3			
9 9 1007/11 10	340.000	98.979	100.132	3.02%			
CCPT DESSAN							
	360.000	100.327	100.820				
						G=3,440	L=176302
	380.000	101.220	- 101.608			88	9
	LONGITUDINAL SECTION - SOUTHBOUND EXIT RAMP (MC50)	102.105	102.196				
e Interest (B)	NNAL SEC				R=1503.500		
	100N 420.000	102.917	102.884		500		
*T 8 1	OUTHEO 440.000	103.583	103.572	a and a second			
Access data way in the access	UD TT VTP 448,958	and a set of the set o	103.860				
13686 E	17 RAMP 460.000	104.253	104.255			VC-30.001	1P STN 463.958 RL 104.396
			101.070			100	
	VTP 478.959 480,000	104.890	104.876 104.910	0.00%			
FOLDINIE INE SUIDAIE INE FOLDINIE INE FOR OFFICE FOLDINIE INE FOLDINIE INE FOLDINIE INE	500.000	105.539	105.550				
32004TH4 35004TH4 25004BH1 25004BH1 25004BH1 25004BH1 2004 2004 2004 2004 2004 2004 2004 200							
AGI Br srahman Qal Qal Qal Qal Qal Qal Qal Qal Qal Qal	520.000 HTP 524.449	106.167	106.190				
						6-32	110-0-05-05-05-05-05-05-05-05-05-05-05-05-
	540.000	106.807	108.830			10%)	
Transport Roads & Maritime Services	560.000	107.448	107.470		R=4862.200	- a succession	
SS and Street					8	ORNER - COMPANY	
	580.000	108.120	108.110				
CONVENTION	HTP 569.418 VTP	108.411	108.411		<u>! </u>		I IP STM 589.418 RL 108.411
OT F ELOPN ND EA ND EA ND EA ND EA ND EA ND EA							
NOT FOR C							
ROWLE IN THE PARK							
CONSTRUCT CLE PARK MORKS MORKS MORKS MORKS MORKS MORKS MORKS MORKS MORKS							

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

THIS DRAYANG MAY BE PREPARED IN COLCUR AND MAY BE IN [0] [5] [10] [15] [20] [25] [20] [35] [40] [43] [45 [45 [45 [45 [45 [45 [45 [45 [45 [45	COMPLETE IF COPIED	
Actual Listence with 1990 and 100 and 1		
Manual Status Manua Status Manual Status Manual St		
Koljune rok poljur cubra Vilozitič sodiman cubra Ima nost sodiman Iso etco: subovo koljučan Iso etco: nubovo koljučan		
NOT FOR CONSTRUCTION CANDEN AND COMPETITION OT FOR CONSTRUCTION STRIKE FAM INSTRUCT STRIKE FAM INST		

n nar - en men en e semen MMM et d'het e men e

THIS DRAWING MAY BE PREPARED IN COLCUR AND MAY BE INCOMPLETE IF COPIED

	n co	0 X	<	
STATION	DESIGN LEVEL	IORIZONTAL A	ERTICAL ALIG	
ţ	<u> </u>	lignment FT		
300,000 99.99 9 301,516	97 100.050 100.038			0 55 7 00 0
	34 99.890 99.832		738	
340.000 99.7	23 - 99.718			
760.000 00 A	34 90 490			
300,007 03.4	uq 02.700		K=-6694 /100 VC=80 285	P STN 387.397 RL 99.511
380.000 98.9	77 99.202			
P 438390	98.091		0 849 999%	
440,000 97.3	24 98.059	STRAIGHT		
460.000 96.8	24 97.723			
480.000 96.6	10 97.497			
500.000 98.5	97 380		VC=116 K=3543 J	IP STN 496.770 RL 60.524
			100	STN 611.222 RL 97.383
520.000 95.3	97.374			
540.000 96.5	69 97,477			
	97.528 798 97.686			
			L=38.5	
P 590.120	98.050			- ~
1			$ \geq$	
	p 300,000 99,91 320,000 99,81 327,255 99,41 340,000 99,41 380,000 99,41 380,000 99,41 380,000 99,41 400,000 99,41 400,000 99,41 400,000 99,41 400,000 99,41 400,000 99,41 400,000 99,41 400,000 97,3 460,000 96,81 500,000 96,81 500,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 520,000 96,21 530,000 96,21 530,000 96,21	STIDU STIDUC STIDUC STIDUC a 300.000 99.997 100.059 a 300.000 99.894 99.890 a 300.000 99.723 99.71 a 300.000 99.723 99.731 a 300.000 99.434 99.490 a 300.000 99.434 99.490 a 300.000 89.777 99.202 a 400.000 89.732 98.854 p 400.000 97.855 94.490 a 400.000 97.855 94.490 a 400.000 96.824 97.723 a 400.000 96.810 97.497 a 60.000 96.810 97.497 a 60.000 96.810 97.497 a 60.000 96.810 97.497 a 60.000 96.275 97.380 a 500.000 96.569 97.477 a 500.000 96.569 97.477 a 500.000 90.768 97.588 a 500.000 97.208 97.588 a 500.000 97.208	STITION SPS SPT SQL EFF SPS SQL EFF <thsps eff<="" sql="" th=""> <thsps eff<="" sql="" th=""></thsps></thsps>	STITUO EXISTING [EVELS] DESC SOLUCION SOL

 THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

 0
 15
 1/3
 1/3
 1/40
 1/45
 1/3/mm COLAD VAL

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE 0 5 10 115 120 123 120 125 140 143 120mm CN A1 121 12 12 120 125 140 143 120mm CN A1	I SZE ORGANI						
A definition of the second sec	STATION	EXISTING LEVELS	DESIGN LEVELS	CROSSFALL LEFT	HORIZONTAL ALIGNMENT	VERTICAL ALIGNMENT	
E TALE SALA SCHEDTAR	600.000	97.624	93.162			$\left[\right]$	
	620,000	97.814	98.295			K=29677100 VC=49130	IP STN 618.396 RL 98.301 STN 629.610 RL 98.310
Dispect removes a	640.000 VTP 642.961	97.878	98.292 98.230				
A K	660.000	97.846	98.203				
A the provided of the provided	680,000	97.807	95.113			G=-0.450% L=73.947	
	700.000	97.743	93.023		Ŧ		
LONGITUDINAL SECTION - MENANGLE ROAD SOUTHBOUND (MCA0)	VTP 716.908 720.900	97.695	97.947 97.934	3002	R=-5500.000		· · · · · · · · · · · · · · · · · · ·
	740.000	97.654	97.868				
NGLE ROAD SOUTHER GERUPAS GERUPAS GERUPAS GERUPAS GERUPAS GERUPAS MARINE VIEWS MARINE VIEWS MARINE VIEWS MARINE VIEWS MARINE VIEWS MARINE VIEWS	760.000	97.666	97.838			VC=86.847 K=10928/100	IP STN 760.331 RL 97.752 STN 760.033 RL 97.837
BOUND (MC)	780.000	97,746	97.845				
zindermark zons webn ap- kennen	800.000 VTP 803.755	97.829	97.889 97.902				· · · · · · · · · · · · · · · · · · ·
4401 B7 100 B0 100 B	820.000 HTP 830.544	97.910	97.958 97.994			C=0.345%	
CENT CENT CENT CENT CENT CENT CENT CENT	HTP 843.443 VTP	98.015 98.034	98.027 98.038		STRAIGH		IP STN 843.443
Roads & Maritime	УТР						PiL 98 038
NOT FOR							
CONSTRUCTION ICLS A3 ILE PARK WORKS SOUTHBOUND (ICLA) - SHEET 3 OF 5 SOUTHBOUND (ICLA) - SHEET 3 OF 5 MO-02443 CP							

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

THIS CRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCO 10 4 10 14 23 23 20 25 49 45 60mm on M3 0/26 0R	2).PLETE IF COPIED Varia	
Operation Operation Operation Second state Second state		
Rof bake ine 7002009 Rof of sedown CLSR external provent CLSR maximum Image recover num num Image recover num num Image recover soutown num Image recover num num		
NOT FOR CONSTRUCTION SPRIC FAMILIAN CONVENTION SPRIC FAMILIAN CONVENTION MENANGLE ROAD TO MENANGLE PARK ROAD ALIGNMENT AND FARTHWORKS INSERTIMANA CONSTRUCTION SWITCH MENANGLE ROAD TO MENANGLE PARK ROAD ALIGNMENT AND FARTHWORKS MECHANISM CONSERTIONS MELANAST DESIGN		

BE INCOMPLETE IF COPIED

THIS DRAWING MAY BE PREPARED IN COLOUR 0 5 10 15 20 25 20 35 40 45	R AND MAY BE INCOMPLETE IF COPIED	
0 3 2 3 4		
norsie (10517.84) - "silvan Businonalii (1801500-16650emts). Reneti (18076-20028) Teleon Reset (180 nr. Sincht) Eleon		
Authonomic Scherosoft (2004)		
ANDERS NOLDERS I I LEARNING NOLDERS I LEARNING NOLD		
ate Hot if and		
NOT FOR CONSTRUCTION CONSTRUCTION ON FOR CONSTRUCTION Spendic Frame PRRMIX A3 STAGE 1 ROAD DEVELOPMENT A4 MELANGLE ROAD DEVELOPMENT A5 MELANGLE ROAD TO MELANGLE PARK A5 ONETUDAL SCITON - MELANGLE ROAD TO MELANGLE PARK A5 ONETUDAL SCITON - MELANGLE ROAD TO MELANGLE PARK A5 ONETUDAL SCITON - MELANGLE ROAD TO MELANGLE FORD A55 ONETUDAL SCITON - MELANGLE ROAD SUTHBOLND MCAULT SCITONAL SCITONAL SCITON - MELANGLE ROAD SUTHBOLND MCAULT SCITONAL SCITON - MELANGLE ROAD SUTHBOLND MCAULT SCITONAL SCITONAL SCITONAL SCITON - MELANGLE ROAD SUTHBOLND MCAULT SCITONAL SCITO		

DOWING FLL COLTONIIMIC DOWING FLL COLTONIIMIC DESTANDED E MENALINEER TO DOWNER THE SOURCE OF DOWNER DESTA WOOL FLAG WID FOR DOWNER THE SOURCE OF DOWNER THE DOWNER FLAG WID FOR DOWNER THE SOURCE OF D	STATION	EXISTING LEVELS	DESIGN LEVELS	CROSSFALL LEFT	VERTICAL ALIGNMENT	DATUM R.L. 510	
ALEE INTE ALEEVOOD DE CASE ALEEVOOD DE C	VTP HTP 0.000	99.038	99.640				IP STN 0.000 RL 69.640
1110.00 二 20 1110.00 二 20 1110.00 二 20 1110.00 二 20 1110.00 二 20	20,000	99,659	99.663		<u>G=0,115%</u>	1 -28.57	
97 97 - 314223 Т.С.С.19 А.Д Еміна 12417 - Андоновскі Лецька, одсклаготом 12417 - Калоновскі Лецька, одсклаготом 12417 - Калоновска Сол. Дог. С.С.Бани 12417 - 13426 г. Ал. Сол. Дог. С.С.Бани	VTP 35.521 40.000	89.657	99.681 99.687		8		
CEPF CESSON CEPF CESSON					STRAIGHT		
	60.000	99.691	99.732		N=100120	VC-70.334	IP STN 70.668 RL 99,722
A RECEVUE	80,000	99.783	99.808			2	
SET 1520 2 CLUCK 100 10 CLUCK	HTP 96.980 100.000 5 VTP 105.856	99.892	99.897 69.915 99.952	JIII .			
	T VTP 112.158	100.035	99.993 100.042		6	E630Z	
	SECTION				VC-27257		IP STN 125.766 RL 100.082
Autors (Incompany) in Maleria (Autors) (Incompany) in Maleria (Incompany) in Maleria (Incom	VTP 139.415 140.000	100.098	100.138 100.141			/	
	n 160.000	100, 193	100.224	\$600.E		·	
ŝ. Ķ	97/UEO0 180.000	100.232	100,310		N= 144920 / IW	VC-63/391	IP STN 171,205 RL 100,268
ALIGNARY DESERVACEO DI CONDOCIO	200.000 VTP 202.996	100.154	100.399		R=450,000	<u>\</u>	
accord lancad Accord Ac					8		
	220.000	100.118	100.475				
	240,000	100.238	100.511		VC=74.556		IP STN 240.324 RL 100.531 STN 247.784 RL 100.514
Transport Roads & Maritime Services	260.000	100.290	100.506		and an		
CAUDENT STARKE F STARKE F STARKE F UNGAD A LONGINUM	VTP 277.651 280.000	100.261	100.469 100.461		8		
NOT FOR CONSTRUCTION CAUDE HAND CAMPTELITOW OTF COUNCILS STRING FARM FARMAN STAGE 1 ROAD TO MENANGLE PARK ROAD ALIGNMENT AND EARTHWORKS ROAD ALIGNMENT AND EARTHWORKS MORTOPINAL SCOTTO - SEMENTION OF SEMENTIONS METADALISE DOS/017000204	300.000	100.224	100.381		VC-62.233		
OT FOR C WINGIT COUNCIL ELOPMENT DIMENANGLE ND EARTHW WINGLE ROADHW WINGLE ROADHW							
SONSTI LS DARK ORKS ORKS							

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

0 6 10 15 20 25 30 35 40 45 55mm CM A3	SIZE ORIGINAL							
0 3 19 15 12 17 18 20 15 18 18 18 18 18 18 18 18 18 18 18 18 18	STATION	EXISTING LEVELS	DESIGN LEVELS	CROSSFALL LEFT	HORIZONTAL ALIGNMENT R=450.000	VERTICAL ALIGNMENT	DATUM R.L. 550	
in the second seco	300.000 HTP 303.555	100.224	109.361 109.336	3.17%	R=450 000	$\left \right\rangle$		IP STN 308.758 Rt. 100.375
2.0169 100 100 101 101 101 101 101 101 101 10	320.000	100.120	100.195			K=-6164.1100 VC=62.233		
0.1221 0.1221 0.1221 0.1221 0.1221 0.1221 0.1221 0.1222 0.122 0.	VTP 339.884 340.000	93.791	59.956 99.955			1		
SCA SCA SCA SCA	380.000 VTP 368.483	99.373	99.703 99.618			C=-1,312%		
Pancortan v Bancortan v Parcoass	380,000	98.670	99.425	2.05%	A ANALY IN A REPORT OF	K=5816 /100 VC=23.998		IP STN 398.462 RL 99.355
PROJECT DE SEALES ON HORAZ, L100 VIERT 1, X00 VIERT 1, X00	400.000 VTP 406.462	98.006	99.081 98.956		STR			
	420.000	96.926	98.685		STRAIGHT	0-13077 L=33777		
Actual socowear is where a more a more and a	440.000 VTP 440.165	96 356	98.285 98.231	E SECTOR		K=36451100		IP STN 449.288 RL 98.099
	VTP 458.390 460.000	96.174	97.963 97.939		A CONTRACTOR OF	100 L=17.806		
	VTP 476.198 480.000	05.986	97.696 97.641			8/ 5		
	500.000	- 95.119	97.438					
All Color All Color <t< td=""><td>HTP 517 974 520,000</td><td>94.106</td><td>97.378 97.378</td><td></td><td>and Construction of the second se</td><td>K=2847/100</td><td>67 192 27 - 55</td><td>IP STN 514.693 STN 518.900 RL 97.118 RL 97.375</td></t<>	HTP 517 974 520,000	94.106	97.378 97.378		and Construction of the second se	K=2847/100	67 192 27 - 55	IP STN 514.693 STN 518.900 RL 97.118 RL 97.375
	540.000	96.108	97.454					
REMARKED THE REAL OF THE RELATION OF THE RELAT	VTP 553.191 560.000	96.475	97.682 97.664		R=8350,003		··	
	580.000	96.626	97.905	3.00%	83	6=1.205%		
CONDENTIND CONFEE STARGE I ROAD ROAD ALICIAMEA LONGITUDINAL SECTION INSTITUTIONAL SECTION INSTITUTION	VTP 595.735	And a second	98.094					
NOT F(BELTOWN CITY N DEVELOPM ND ATO DEVELOPM NAD TO MENT ON-MENNICLER DISZO17700	600.000	97,120	98.143				K	1
COUNCILS COUNCILS COUNCILS TENT TENT TENT TENT COUNCILS C								
NOT FOR CONSTRUCTION CONDENTADO CAMPBELLTOWN CITY COUNCLES SERIA FRAN PREVENT SERIA FRAN PREVENT STAGE 1 ROAD DEVELOPMENT MERNANGLE PARK COLORITIZE STATUS - MERNANGLE PARK MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 MERNANGLE ROAD NORTHERUND (NCB0) - SHEET 2 OF 5 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
					.			

 THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE NICOMPLETE IF COPIED

 0
 16
 10
 15
 20
 25
 44
 45
 55mm CMAS SEE 0465841

DRAWING MAY BE PREPARED N COLOUR AND MA 1 10 15 20 23 30 13 00 14 50m C 1 10 15 20 5 20 10 10 10 10 10 10 10 10 10 10 10 10 10	A DE INCOMPLETE IN CONTES				
Separate Fat Docymer van c	EXISTING LEVELS	DESIGNLEVELS	HORIZONTAL ALIGNMENT GROSSFALL LEFT	DATUM RL 5:9 VERTICAL ALIGNMENT	
	600.000 (97.120	98.143			
0. 16.2016 12. 07.2016 12. 07.2016	620.000 97.120	98.288		RE-2972/100	IP STN 620.316 RL 98.391 SIN 631.534 RL 98.310
In contract of the second s	640.000 97.039 VTP 644.898	98.208 98.260	R=6300 000		
nove Too	660,000 97 179	98.212	100		
силинания Вилонания Украинания Украинания	680,000 97.212	98.122		G=-0,450%	
AND	HTP 693616 97.270	98.039 68.032			
	TUDIA VTP 719.053 720,000 97.323	97.947 97.942			
Michael Coordination	TTON - MEL 740.000 97.474	97.873	STRAIGHT		
	LOCKITTUDINAL SECTION . VTP 719.053 97.323 720.000 97.474 720.000 97.474 740.000 97.474 760.000 97.474 760.000 97.587 760.000 97.587 97.587 97.714	97.840	3,00%	K=10876 /100	₽° STN 762.276 RL 97.752 STN 767.954 RL 97.837
	RTHB HTP 779247 780,000 97.714 D (M	97.842 97.843	*		
Image: Section (Section (800.000 97.790 VTP 805.499	97.884			
Pool Fr statutan stat	820.000 97.693	97.951		L=31.597	
NSW	VTP 837.055 840.000 97.993	98.011 98.021	R=-1000.000		
Roads & Maritime	860.000 93.125	98.130		VC-43.805	IP STN 858,926 RL 98.068
ಹ	880.000 99.30 HTP 880.901 98.310 VTP	98.300 98.310			IP STN 880.601 RL 98.310
NOT FOR CONSTRUCTION					
CONSTRUCTION TILS A3 TILS A3 TILE PARK MORKS MORKS MORKS					

THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

	8 3 3 5 4 4 5 mm CHARDE ORGENE	
ON MODEL F		
etamrik 1804 11. F(S) USED		
Forebody		
(Density)(1)		
01-1145-02 01-1145-02		
00-80-0251-0255.de 5.080/0780 91 12 12		
41X 61 20 21 21X 61 20 21 21X 61 20 22 21X 61 20 20 21X 61 20 20 21X 61 20 2		
STOPPOT TREASA M		
1SION DESC ONCEPT DES ONCEPT DES		
States St		
A HATTHORE		
SCALES ON AS SEE DO		
	The second se	
PERFECTION TUN		
N N N		
AD-0254		
ND 0254		
AUG/2010	2007ut	
SCAL		
HORRE 24		
shimari	szhmei	
n Transport Investigation Services Investigation Research Services Investigation Research Services Investigation Research Services		
Ads & Ads & Adds		
Roods & Martime NSW Services		
	40	
SPRING FARM PARKWI STAGE 1 ROAD MENANGLE ROAD ROAD ALIGNMEN LONGTIDINAL SECTION INCREASING STATE STATE STATE STATE STATE STATE DESIGN		
EPT DES		
IGN		
O MEN NANGLET		
AENT RIANGLE 20202		
E PARK ORKS ATHBOUN	NSNO.	
SPING FRAMPARYMY STAGE 1 ROAD DEVELOPMENT STAGE 1 ROAD DEVELOPMENT ROAD ALIGNMENT AND EARTHWORKS ROAD ALIGNMENT AND EARTHWORKS LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 LONGINUMAL SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MEDI SECTION - MENNELE ROAD NORTHBOLVID (MCBBI - SHEET 4 OF 5 MED SEC	NOT FOR CONSTRUCTION	
<u>第一</u> 第一 4 一 4		
彩招招 2	<u>کا</u>	

THIS DRAWING MAY BE PREPARED IN COLCUR AND MAY 10 [5] 10] 15] 29] 25] 30] 25] 49] 45] 50mm (24		
N Home Party Constraints and the party of the party o		
MICY 200E 20 MIC TOTAL MICH 120 2 MICH 120 1 MICH 120 2 MICH 120 1 MICH 120 1 MICH 120 1 MICH 120 1 MICH 120 MI	OPPH	
Nacus Science Barry Nacus Nacu		
Terne screw terne screw screw resource screw resource scre		
NOT FOR CONSTRUCTION		

Incorporation (Non- angeles of the sector of sector as a sector as a sector of sector	STATION 0.000 P.6.17 PP 9.647 VTP 20.000 40.998 HEE 49.998 HEE 49.998 HEE HEE	EUSTING LEVELS EUSTING EUSTIN	IP STN 0.000
Autronie n.e. Kul in Autronie n.e. Kul in m.e. Kul in commercial sources commercial sources commercia			
NOT FOR CONSTRUCTION			

 THIS DRAWING MAY BE PREPARED IN COLOUR AND MAY BE INCOMPLETE IF COPIED

 10
 15
 16
 16
 16
 17
 10
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16
 16

Standard ALLEGADINAL EVEN STATION STAT	DATUM RL 75.5 VERTICAL ALISIMENT HORIZONTAL ALISIMENT
OF Issues OTO VTP 0.000 64.522 65.168 Image: State Sta	IP STN 0.000 RL 85.168 RL 85.168
	STN 19.374 IP STN 19.355 RL 8487 RL 84.585 NL 84.585
USE VI 20,000 85,137 84,699 1	
СК 2000 Казарана Казар	약 (1.65)
VTP 74.344 85.627	P STN 82310
Mathematical 85,851 85,851 85,851 Mathematical 700,000 85,851 85,851 85,851	부 전 문 전 문 85.651
out 1000 1 10000 1 1000 1 1000 1 1000 1 1000 1 1000 1 1000 1 1000 1 1000	
In the transfer GROUT/I Instrumentation Internet Homes Internet Homes Internet Homes	
PLOT DUE / ML THE THE CONST	
Anonesso Ano	
Transport Roads & Maritime Services	
70	
NOT FOR CONSTRU CANDENAND CAMPBELLTOWN GITY COUNCILS STRAKE FARM PARKWAY STAGE 1 ROAD TO DEVELOPMENT MENANGLE ROAD TO MENNINGLE PARK MENANGLE ROAD TO MENNINGLE PARK	
NOT FOR CONSTRUCTION	
CONSTR CONSTR Jus Jus FPARK VORKS VORKS	

Corrs Chambers Westgarth

Annexure B

Milestones and Contributions

See attached.

COLUMN 1 Milestone	COLUMN 2 Funding Date	COLUMN 3 Funding Amount Allocation (\$AUD)	COLUMN 4 Funding Party	COLUMN 5 Security Party
Milestone 1	The date this Deed is executed.	(100%) \$3,347,826 in respect of the Property B Land	Landcom	N/A
		(100%) \$1,652,174 in respect of the Property A Land	Landcom	N/A
Milestone 2	The Funding Date for Milestone 2a.	(100%) \$38,834,784 in respect of the Property B Land	Landcom	N/A
		(100%) \$19,165,216 in respect of the Property A Land	Landcom	N/A
		(100%) \$23,000,000 in respect of the Dahua Land	Dahua 3	N/A
If the Funding Allocation Am partiy secured by a Security:	If the Funding Allocation Amount for Milestone 2 will be partly paid and partly secured by a Security:			
Milestone 2a	The date which RMS, acting reasonably, nominates as being 10 weeks before the date on which RMS	(25%) \$9,708,696 in respect of the Property B Land	Landcom	N/A
	issues a Request for Tenders (or similar document) for the head construction contract for the Required	(25%) \$4,791,304 in respect of the Property A Land	Landcom	N/A
	vorks, or such later date as specified in a written notice from RMS to Landcom and Dahua 3 following approval of the Final Business Case. Landcom may, in its discretion, provide the funding at an earlier date.	(25%) \$5,750,000 in respect of the Dahua Land	Dahua 3	N/A
Milestone 2b	The date which RMS, acting reasonably and with due diligence, nominates in a written notice to	(25%) \$9,708,696 in respect of the Property B Land	Landcom	Landcom

Corrs Chambers Westgarth

3437-2025-8564v54 L\329906431.1 ME_152649452_1

page 37

COLUMN 1 Milestone	COLUMN 2 Funding Date	COLUMN 3 Funding Amount Allocation (SAUD)	COLUMN 4 Funding Party	COLUMN 5 Security Party
	Dahua 3 and Landcom under clause 5.2(c)(l) , as the date on which 25% of the Required Works will	(25%) \$4,791,304 in respect of the Property A Land	Landcom	Landcom
	have been completed. Landcom may, in its discretion, provide the funding at an earlier date.	(25%) \$5,750,000 in respect of the Dahua Land	Dahua 3	Dahua 3
Milestone 2c	The date which RMS, acting reasonably and with due diligence, nominates in a written notice to	(25%) \$9,708,696 in respect of the Property B Land	Landcom	Landcom
	Dahua 3 and Landcom under clause 5.2(c)(i), as the date on which 50% of the Required Works will	(25%) \$4,791,304 in respect of the Property A Land	Landcom	Landcom
	discretion, provide the funding at an earlier date.	(25%) \$5,750,000 in respect of the Dahua Land	Dahua 3	Dahua 3
Milestone 2d	The date which RMS, acting reasonably and with due diligence, nominates in a written notice to	(25%) \$9,708,696 in respect of the Property B Land	Landcom	Landcom
	Dahua 3 and Landcom under clause 5.2(c)(i), as the date on which 75% of the Required Works will have been been concluded 1 and on in ite	(25%) \$4,791,304 in respect of the Property A Land	Landcom	Landcom
	discretion, provide the funding at an earlier date.	(25%) \$5,750,000 in respect of the Dahua Land	Dahua 3	Dahua 3
		Total Milestone 2: \$81,000,000		

Corrs Chambers Westgarth

3437-2025-8564v54 L\329906431.1 ME_152649462_1

page 38

Grand total : \$86,000,000

Annexure C

Property A Land, Property B Land and Dahua Land

- 1 Property A Land
 - a. Lot 641 DP600334
 - b. Lot 1001 DP1219028
 - c. Lot 32 DP1101983
 - d. Lot 1 DP598067
 - e. Lot 7 DP801774
 - f. Lot 1002 DP1234642

2 Property B Land

- a. Lot 2 DP554242
- b. Lot 59 DP10718
- c. Lot 12 DP251335
- d. Lot D DP19853
- e. Lot 32 DP1105615
- f. Lot 1000 DP1219023
- g. Lot 1 DP1091474
- h. Lot 9 DP249530
- i. Lot 15 DP251335
- j. Lot 17 DP251335
- k. Lot 4 DP628052
- I. Lot 2006 DP1234643
- m. Lot 2007 DP1234643

- n. Lot 2008 DP1234643
- o. Lot 2009 DP1234643
- p. Lot 2010 DP1234643
- 3 Dahua Land
 - a. Lot 3 DP236059
 - b. Lot 124 DP1097090
 - c. Lot 7 DP787284
 - d. Lot 1 DP727098
 - e. Lot 1 DP708770
 - f. Lot 1 DP707225
 - g. Lot 1 DP249393
 - h. Lot 33 DP1101983
 - i. Lot 31 DP1101983
 - j. Lot 125 DP1097138
 - k. Lot 1 in DP1247661





3437-2025-8564v54 L\329906431.1 ME_152649462_1





Corrs Chambers Westgarth

3437-2025-8564v54 L\329906431.1 ME_152649452_1





3437-2025-8564v54 L\329906431.1 ME_152649462_1

page 43

Corrs Chambers Westgarth

Annexure D

Dedicated Land

Column A	Column B
Title reference	Timing of dedication
Lot 2012 in	No later than:
DP1234643	 4 months after the date the Dedication Notice is issued by RMS; or
	 2 months after the date of execution of the Relevant VPAs by all parties to those Relevant VPAs,
	whichever is the later.
Lot 2011 in	No later than:
DP1234643	 4 months after the date the Dedication Notice is issued by RMS; or
	 2 months after the date of execution of the Relevant VPAs by all parties to those Relevant VPAs,
	whichever is the later.
Additional Dedicated Land.	No later than:
	 4 months after the date the Dedication Notice is issued by RMS; or
	 2 months after the date of execution of the Relevant VPAs by all parties to those Relevant VPAs,
	whichever is the later.

Annexure E

Lease Land

The Lease Land is the land comprising all or part of the parcels specified in part 1 of this Annexure E, as depicted on the plan specified in part 2 of this Annexure E.

1 Title reference

- (a) Lot 1 DP 598067
- (b) Lot 2011 in DP1234643
- (c) Lot 2012 in DP1234643
- (d) Lot 2007 in DP1234643
- (e) Lot 2006 in DP1234643
- (f) Lot 1002 in DP1234642
- (g) Lot 9 in DP249530
- (h) Lot 2008 in DP1234643



2



3437-2025-8564v54 L\3299D6431.1 ME_15264945_1

page 46

Annexure F

Spring Farm SRG Terms of Reference

1 Background

Spring Farm Parkway is a potential east-west road that would link Camden Bypass at Spring Farm to Menangle Road at Menangle Park, including connections to the M31 Hume Highway. When completed, the proposed road would service existing and future residential land releases including Spring Farm, Elderslie, Menangle Park and Mount Gilead and will provide an alternative route to Narellan Road with access to the Hume Highway.

The NSW Government 2016-17 budget has committed \$30 million under the Housing Acceleration Fund Round 4 to fast track planning and Stage 1 construction for the Parkway. RMS has commenced investigations and plans to deliver a strategic design for the corridor and a business case for construction of Stage 1 within the available funding.

RMS is working with Landcom and the Dahua Group to develop a funding agreement for funding towards the delivery of Stage 1 to support the Menangle Park rezoning as endorsed by Campbelltown Council in November/December 2016. Endorsement was on condition of satisfactory arrangements towards the Stage 1 infrastructure prior to sub-division. The Landcom and the Dahua Group contribution to funding is estimated at over 74% of the total funding for Stage 1 of the Spring Farm Parkway.

The design and cross-section developed by Landcom for Stage 1 of the Spring Farm Parkway are generally being used to guide design development and the limits of the Required Works. RMS can vary the ultimate design of Stage 1 to enable delivery of access to the Menangle Park development within the total funding available.

The specifications and the project estimate will be refined as investigations and INSW Gateway approvals progress.

2 Role and Purpose

The intent of this stakeholder reference group is to facilitate cooperation between key stakeholders and RMS during planning, investigations and feasibility assessment for delivery of Stage 1 for the Spring Farm Parkway.

This process will allow for open communication and transparency through the development of the project and finalisation of the specifications and final project estimate for Stage 1.

As funding contributions from Landcom and the Dahua Group are capped and RMS does not have the capacity to provide for any funding gaps, this group will work together to share information and maintain a high level of awareness of investigations, risks and mitigation measures to achieve the outcomes necessary to provide access to Landcom's and the Dahua Group's development in Menangle Park within the available resources.

Key aspects will include:

- ensuring the SRG is fully informed of progress of the Required Works during the term of development and delivery, including the design options and estimates to be included in the project business case;
- (b) ensuring the SRG is fully informed of changes to key program milestones/specifications and/or funding issues and contribute to problem solving;
- discussing major issues and risks with the potential to impact the cost and delivery of the Required Works;
- (d) facilitating an understanding at the SRG of the project management and INSW Gateway framework that is used by RMS to manage the project, control costs and provide value for money;
- (e) facilitating understanding and co-operative problem solving; and
- (f) reporting and making recommendations to the Department/RMS/TfNSW HAF PCG regarding program, specifications and/or funding to support Stage 1 of the Spring Farm Parkway.

3 Term

This Terms of Reference is effective during the period of development and delivery of the Required Works, commencing on the date of the Transport Infrastructure Contributions Deed.

4 Membership

The group will comprise key strategic agencies and partners, including:

- (a) RMS
- (b) Landcom
- (c) The Dahua Group
- (d) The Department

Others to be invited as needed, include:

- (a) Council
- (b) Transport for NSW
- (c) Greater Sydney Commission
- (d) others as agreed.

All parties must ensure that suitably informed and authorised representatives are made available to contribute to meetings and workshops as required.

5 Meetings

- (a) All meetings will be chaired by RMS.
- (b) The meetings will include discussion on:
 - progress in design and construction of the Required Works (as appropriate);
 - progress against the Indicative Program, including having regard to the Target End Date for the opening of the Carriageway to traffic and completion of as much of other the Required Works as possible;
 - (iii) any proposed variations to the scope of the Required Works;

- (iv) progress against the budget for the Required Works; and
- (iii) to the extent reasonable practicable, utilisation of Contaminated materials (including soils and building and demolition waste) on the Dedicated Land for the Required Works and mechanisms to avoid the disposal of waste offsite.
- (c) Without limiting the other issues which they may raise at the meetings, the Developers may raise material issues arising from their review of information relating to the Cost of the Required Works.
- (d) Except as set out below, all decisions are to be made by consensus (i.e. members are satisfied with the decision even though it may not be their first choice).
- (e) If:
 - (i) a material issue arises during the design phase of the Project;
 - (ii) consensus cannot be reached at a meeting on that issue; and
 - RMS, acting reasonably, determines that the resolution of that issue will not interfere with, or cause delays to, the Indicative Program,

then the disputing attendees will be given an opportunity to escalate that issue to their respective senior executives, who will promptly meet and endeavour to negotiate in good faith to resolve the issue in dispute.

- (f) if:
 - (i) a material issue arises at any other time during the Project; and
 - (ii) consensus cannot be reached at a meeting on that issue; and

the process for the escalation of issues to senior executives set out in (e) of this **Annexure F** may occur concurrently with RMS continuing to progress the Project in accordance with the Indicative Program.

- (g) The parties agree that the key objective of the SRG is:
 - to facilitate the opening of the Carriageway to traffic and completion of as much of other the Required Works as possible by the Target End Date and within the budget of the Funding Amount plus the HAF Funding; and

- (ii) if, that objective is not achievable, to achieve delivery of the Carriageway by the Target End Date and within the budget of the Funding Amount plus the HAF Funding.
- (h) RMS will:
 - prepare the agendas and any supporting documents, including regular headline funding reports showing how the Funding Amount and the HAF Funding were expended during the last reporting period, projected expenditure and the balance of funding available for the Required Works;
 - (ii) if requested by a Developer provide:
 - (A) an itemised breakdown and details of the costs towards which the Funding Amount and the HAF Funding have been or is proposed to be applied;
 - (B) copies of any contractor invoices and documentary evidence that RMS can provide as determined by it acting reasonably, to verify the Cost of the Required Works;
 - (iii) prepare and coordinate meeting notes and information;
 - (iv) consult with and agree action items and next steps in relation to the delivery of the works with the Dahua Group and Landcom, in particular, in relation to any variations to the proposed scope of the Required Works; and
 - (v) prepare meeting minutes to be circulated to all parties in attendance.
- Meeting minutes will capture actions and assigned parties responsible for said actions, and these will be raised at subsequent sessions for redress.
- (j) Meetings will be held at least bi-monthly until the Final Business Case. After this, the frequency can be altered by agreement of all parties.
- (k) If sub-groups or specific meeting are required, these can be arranged outside of these times at a time convenient to sub-group members.
- (I) If site inspections are required and agreed to by RMS, these can be arranged at a time convenient to the members in accordance with the construction contractor's induction/other requirements.

6 Relationship to Existing Groups

The Stakeholder Reference Group will seek to resolve issues with the consensus of attendees.

At the outset, the group will seek endorsement of these Terms of Reference.

The Stakeholder Reference Group will continue to report and make recommendations to the HAF PCG regarding program, specifications and/or funding to support the Project.

7 Exclusions

This is not a Project Control Group and relies on the willingness of the parties to work together to resolve issues and successfully develop and deliver optimum outcomes for the Project.

It does not replace existing governance, contractual, statutory or RMS procedures/process. RMS will follow NSW Government and RMS guidelines during project development including for Stakeholder and Community consultation and property acquisition.

The HAF/RMS Governance processes for the Project will apply including INSW Gateway approvals.

Any disputes will be referred to the HAF PCG for consideration.

The stakeholders identified in this group are encouraged to input into these processes.

Membership does not ensure favourable decisions or outcomes but allow issues to be aired and discussed.

This does not replace obligations made in relevant Voluntary Planning Agreement or Transport Infrastructure Contributions Deeds.

Annexure G

Lease

1 Lease Terms

ROADS AND MARITIME SERVICES

AGREEMENT FOR LEASE OF LAND TO ROADS AND MARITIME SERVICES

AGREEMENT made the of 201....

BETWEEN:

[Landcom ABN 79 268 260 688 of Level 14, 60 Station Street, Parramatta NSW 2150 (Landcom)

Dahua Group Sydney Project 2 Pty Ltd (ABN 80 606 391 235) of Suite 20.02, Level 20, 201-217 Elizabeth Street, Sydney NSW 2000 (Dahua 2)

Dahua Group Sydney Project 3 Pty Ltd (ABN 49 606 391 922) of Suite 20.02, Level 20, 201-217 Elizabeth Street, Sydney NSW 2000 (Dahua 3)]

[as applicable]

(Land Owner)

and

Roads and Maritime Services (ABN 76 236 371 088) of 20-44 Ennis Road Milsons Point NSW 2061, a NSW Government agency and corporation incorporated under the *Transport Administration Act 1988* (NSW) (**RMS**)

2 Definitions

In this lease these terms have the following meanings:

- (a) Approval means any consent, approval, authorisation, determination, licence, registration, order, permission or concurrence required by any Law required for the commencement, execution, Completion or use of the Required Works.
- (b) Completion means that the Required Works have been completed to the relevant percentage as specified for each Milestone set out in Column 2 in Annexure B of the TIC Deed, as determined by RMS acting reasonably and evidenced by the status reports given by RMS under clause 9.1 of the TIC Deed.
- (c) **Contamination** has the same meaning as in section 5 of the *Contaminated Land Management Act* 1997 (NSW), including hazardous materials such as asbestos in structures and soil.
- (d) Environmental Law means any law relating to the environment
- (e) Environmental Notice means any direction, order, demand, notice, declaration or other requirement from any Government Agency (whether written or oral) under an Environmental Law.
- (f) **Government Agency** means a department of State, statutory or public authority, instrumentality, statutory corporation, body or person whether Commonwealth, State, territorial or local.
- (g) Law means:
 - (i) any statute, regulation, rule, proclamation, order, ordinance or bylaw whether present or future and whether Commonwealth, State, territorial or local;
 - (ii) the common law;
 - (iii) any Approval (including any condition or requirement under an Approval); and
 - (iv) any standards, criteria and guidelines made or approved by the NSW Environment Protection Authority in relation to Contamination or the Remediation of Contamination
- (h) Lease Land means the land comprising all or part of folio identifiers [insert title reference] as depicted on in the plan in the Schedule. [Dahua note: Please put this definition in alphabetical order. Dahua has left it in its current location to show the mark-up on the definition wording.]
- (i) Make Good Works means:
- (i) restore the Lease Land and any land accessed by RMS or its Personnel as nearly as practicable to its condition prior to the Required Works or access, excluding any changes to the condition of the Lease Land authorised as part of the Required Works, unless otherwise agreed with the Land Owner; and
- (ii) Remediate any Contamination:
 - (A) that is caused by RMS or its Personnel; or
 - (B) that existed prior to the commencement of the Required Works and is disturbed by or exacerbated (but only to the extent the Contamination was disturbed or exacerbated by the Required Works,

other than any Contamination which:

- (C) is permitted to remain under this Deed or pursuant to any Approval obtained in respect of the Required Works; or
- (D) is permitted to remain as otherwise agreed with the Land Owner.
- (j) **Personnel** means the employees, agents, officers, contractors, subcontractors, suppliers, consultants and invitees of RMS and [Landcom or the Dahua Group] (as applicable).
- (k) Remediation means any clean up, remediation or restoration measures relating to any Contamination with respect to the Dedicated Land, whether in existence prior to, on or after the date of the TIC Deed, including any remediation required under any Environmental Notice or under any Law in relation to such Contamination
- Required Works has the same meaning as in the TIC Deed.
- (m) **TIC Deed** means the Menangle Park Spring Farm Parkway Transport Infrastructure Contributions Deed between RMS, the Land Owner and [the Dahua Group/Landcom] dated [insert].

3 Grant of lease

- 3.1 The Land Owner leases the Lease Land to RMS for a term of 3 years from the date the TIC Deed is executed, and thereafter, month to month, for the purposes of undertaking the Required Works, on the terms in this document.
- 3.2 RMS must pay a one-off rental fee of \$1.00, on demand if requested by the Land Owner, by way of consideration for the grant of this lease.

4 Requirements of access

RMS must (and must ensure that its Personnel) at all times when accessing the Lease Land:

- (a) comply with all relevant Laws and Approvals;
- (b) obtain and maintain appropriate levels of insurance;
- (c) cause as little disturbance or inconvenience to any other occupiers and users of the Lease Land as is reasonably possible;
- (d) take all reasonable steps to prevent damage to any property or harm or nuisance to any persons in or near the Lease Land;
- (e) not cause any damage or loss to the Lease Land other than to the extent necessary for the carrying out of the road works; and
- (f) subject to clause 6(c), comply with all reasonable site access requirements (including workplace health and safety requirements) required by the Land Owner, or their Personnel.

5 Variation to Lease Land

- 5.1 If, at any time prior to the termination of the Lease, RMS considers, acting reasonably, that it requires a lease over additional land for the carrying out of the Required Works, RMS may give notice to the Land Owner indicating on a plan the additional land it requires for the Required Works.
- 5.2 On RMS giving notice under clause 5.1:
 - the parties are to negotiate in good faith to agree the new boundaries of the Lease Land;
 - (b) if the parties reach an agreement on the new boundaries of the Lease Land:
 - (i) the additional land is deemed to form part of the Lease Land for the purposes of this clause 5; and
 - (ii) RMS must prepare and provide to the Land Owner a plan showing the new boundaries of the Lease Land; and
 - (iii) the plan in the Schedule of this Lease is deemed to be replaced by that plan provided under clause 5.2(b)(ii) and the Lease under clause 3 applies to the Lease Land as depicted in the new plan; and

6

(c) if the parties are unable to reach an agreement on the new boundaries of the Lease Land, the dispute resolution procedures in the TIC Deed apply.

Access for Landcom and Dahua

If, at any time prior to the termination of the Lease, the Land Owner requires access to the Lease Land to carry out works as part of its Development on the Lease Land (as the case may be):

- (a) the Land Owner must notify RMS at least 2 weeks prior to accessing any part of the Lease Land of that part of the Lease Land required for their proposed works, and the nature and timing of those works;
- (b) the parties agree to use best endeavours to negotiate and coordinate their respective activities (and RMS will procure the coordination of its head contractor's activities) so as to minimise disruption to the Required Works and the works proposed by the Land Owner; and
- (c) once RMS' head contractor has taken possession of the Lease Land, when accessing the Lease Land, the Land Owner must comply with all reasonable site access requirements (including workplace health and safety requirements) as required by RMS' head contractor.

7 Release

- 7.1 RMS releases the Land Owner from all actions, proceedings, claims, demands, liability, loss, damages and costs arising from the access, use or occupation of the Lease Land under this Lease by RMS or its Personnel except:
 - (a) to the extent caused or contributed to by a negligent or wrongful act, omission or default of the Land Owner; and
 - (b) as a result of any breach of this Lease by the Land Owner.

8 Work, Health and Safety

- 8.1 In this clause 8:
 - (a) WH&S Act means the Work Health and Safety Act 2011 (NSW).
 - (b) WH&S Legislation means the WH&S Act and the WH&S Regulation.
 - (c) WH&S Regulation means the Work Health and Safety Regulation 2011 (NSW).
 - (d) "Workplace", "construction project", "construction work", "person conducting a business or undertaking" (PCBU) and "principal contractor"

have the same meanings assigned to those terms in the WHS Legislation.

- 8.2 The Land Owner authorises RMS to exercise such authority and management and control of the workplace in connection with the road works as is necessary to enable RMS to discharge the responsibilities imposed on a principal contractor under the WHS Legislation.
- 8.3 Without limiting any other provision of this Lease, RMS acknowledges and agrees that:
 - (a) it is the PCBU that commissions the construction work in connection with the road works for the purposes of the WHS Legislation;
 - (b) it has management and control of the workplace in connection with the road works for the purposes of the WHS Legislation;
 - (c) it is the principal contractor in connection with the road works unless RMS engages another PCBU as the principal contractor for the road works; and
 - (d) it has sufficient authority and management and control of the workplace in connection with the Required Works to comply with its obligations as principal contractor, or to enable another PCBU it engages as principal contractor to comply with its respective obligations under the WHS Legislation.
- 8.4 Without limiting RMS' obligations under any other provision of this Lease, RMS:
 - (a) must at all relevant times exercise and fulfil its functions and obligations in relation to work, health and safety under the WHS Legislation and this Lease in connection with the Required Works, including as principal contractor;
 - (b) if RMS engages another PCBU as principal contractor, must:
 - (i) require the principal contractor to comply with the obligations imposed on RMS in relation to work, health and safety under this Lease; and
 - ensure that the principal contractor fulfils its functions and obligations in relation to work, health and safety under the WHS Legislation and this Lease;
 - (c) must not:

- (i) carry out; or
- (ii) permit any other PCBU engaged by, or under the control or direction of, the RMS to carry out,

any construction work unless RMS, or a PCBU engaged by RMS, is exercising and fulfilling the functions and obligations of principal contractor under the WHS Legislation in respect of all construction work carried out under this Lease; and

- (d) to the extent not prohibited by Law, must indemnify the Land Owner against any damage, cost, expense, loss or liability suffered or incurred by the Land Owner arising out of or in connection with:
 - (i) any failure of a PCBU engaged by RMS to exercise or fulfil the functions and obligations of the principal contractor under the WHS Legislation or under this Deed or to otherwise comply with this clause 8; and
 - (ii) any work health and safety claims in connection with the Required Works or RMS' workplace except to the extent that they are directly caused by a wrongful, negligent or unlawful act or default of the Land Owner.
- 8.5 Without limiting any other obligations under this Lease, RMS must, and must ensure that any PCBU it engages as a principal contractor must:
 - (a) comply, and must ensure that its subcontractors comply, with the WHS Legislation (including the obligation under the WHS Legislation to consult, cooperate and coordinate activities with all other persons who have a work health and safety duty in relation to the same matter);
 - (b) if requested by the Land Owner or required by WHS Legislation, demonstrate compliance with the WHS Legislation, including providing evidence of any approvals, certificates, authorisations, licences, prescribed qualifications or experience, or any other information relevant to work health and safety matters;
 - (c) notify the Land Owner immediately (and in any event within 12 hours of such matter arising) of all work health, safety and rehabilitation matters arising out of, or in any way in connection with, the Required Works;
 - (d) carry out the Required Works safely so as to protect persons and property and the environment;

- (e) insofar as the Principal Contractor, in carrying out the Required Works is under any duty imposed by the WHS Legislation, do everything necessary to comply with any such duty; and
- (f) ensure that it does not do anything or fail to do anything that would cause the Land Owner to be in breach of the WHS Legislation.

9 Make Good

- 9.1 If requested by the Land Owner by notice in writing to RMS, RMS must, within 60 days after the termination of this Lease, carry out and complete the Make Good Works.
- 9.2 RMS must obtain at its own cost and risk all relevant and necessary Approvals for all parts of the Make Good Works before commencing the Make Good Works.
- 9.3 RMS may not seek any modifications to any Approvals relating to the Make Good Works without the prior written endorsement of the Land Owner, which must not be unreasonably withheld.
- 9.4 RMS must ensure the Make Good Works:
 - (a) comply with all Approvals;
 - (b) are carried out in a proper and workmanlike manner;
 - (c) do not cause any loss, injury, death or damage to Land Owner or to any person or entity, and
 - (d) comply with all Laws and requirements.
- 9.5 RMS must notify the Land Owner in writing once the Make Good Works are complete.
- 9.6 If RMS does not complete the Make Good Works by the date stipulated in the notice served under **clause 9.1**:
 - the Land Owner may serve notice on RMS providing a further period of 30 days to complete the Make Good Works; and
 - (b) if RMS fails to complete the Make Good Works after service of notice, the Land Owner may undertake the works necessary to complete Make Good Works and RMS is responsible for the costs incurred by the Land Owner in carrying out the Make Good Works. RMS must reimburse the Land Owner within 15 days after receipt of a tax invoice issued by the Land Owner or the Land Owner's contractor.

10 Contamination

On and from the date that RMS commences the Required Works, RMS:

- (a) releases the Land Owner from any claim or liability arising from Contamination that is caused by RMS or its contractors on the Lease Land or that existed prior to the commencement of the Required Works, to the extent that the pre-existing Contamination is disturbed or exacerbated by the Required Works; and
- (b) must comply with all Environmental Laws and Environmental Notices in relation to the Required Works, to the extent they relate to Contamination referred to in clause 10(a).

11 Assignment

This lease must not be assigned or transferred without the prior written consent of the other party, which must not be unreasonably withheld or delayed.

12 Termination

The parties acknowledge and agree that this lease may be terminated by either party by notice to the other party in writing that:

- (a) the Required Works have been completed; or
- (b) the TIC Deed has been duly terminated.

THE SCHEDULE HEREINBEFORE REFERRED TO:

The Lease Land is the land comprising all or part of the following folio identifiers: [insert]

[Insert plan of Lease Land]

Executed by Roads and Maritime Services by its duly appointed officer in the presence of:)))	
Witness		Officer
Name of Witness (print)		Name of Officer (print)
Executed by [Land Owner]))	
Company Secretary/Director		Director
Name of Company Secretary/Director (print)		Name of Director (print)

Annexure H

See attached.

•

3437-2025-8564v54 L\329906431.1 ME_152649462_1



7





3437-2025-8564v54 L\329906431.1 ME_152649452_1







page 65

3437-2025-8564v54 L\329906431.1 ME_152649462_1

Annexure I

Terms of Works Easement

[Drafting note: The following terms cover easements for batters and easements for drainage. If a particular easement is required only for batters or only for drainage, then the provisions and heading text which relate to the purpose for which the easement is not required should be deleted for that easement.]

1. DEFINITIONS

- 1.1 In this easement, the following terms have the following meaning:
 - (a) Approval means a consent, approval, authorisation, determination, licence, registration, order, permission or concurrence required by any Law required for the commencement, execution, carrying out and completion of any works or activities on land.
 - (b) Contamination has the same meaning as in section 5 of the Contaminated Land Management Act 1997 (NSW), including hazardous materials such as asbestos in structures and soil.
 - (c) **Drainage Infrastructure** means any line of pipes, swales, drainage culvert or other structure or infrastructure constructed for the purpose of draining water (whether rain, storm, spring, soakage or seepage water).
 - (d) Easement Site means that part of the land shown on the Plan as burdened by the easement.
 - (e) Law means:
 - (i) any statute, regulation, rule, proclamation, order, ordinance or by-law whether present or future and whether Commonwealth, State, territorial or local; and
 - (ii) the common law.

- (f) Plan means DP [insert]. [Drafting Note: Plan of the easement site will be prepared at the time the easement is being registered]
- (g) **Remediate** has the same meaning as in the Contaminated Land Management Act 1997 (NSW).
- (h) RMS means Roads and Maritime Services (ABN 76 236 371 088) a NSW Government agency and corporation incorporated under section 46 of the *Transport Administration Act 1988* (NSW).
- (i) **RMS Works** means the works to be carried out by or on behalf of RMS in respect of the Spring Farm Parkway Roadway.
- (j) Spring Farm Parkway Roadway means the proposed east-west road that would link Carnden Bypass at Spring Farm to Menangle Road at Menangle Park, including connections to the M31 Hume Highway.

2. WORKS BY RMS OR THE OWNER OF THE LAND BENEFITTED

- 2.1 The owner of the land benefited may, or (if the owner of the land benefited is not RMS) must permit RMS to:
 - (a) construct and maintain on the land burdened, but only within the Easement Site, whatever batter or embankment is reasonably necessary to support the surface or subsurface of the land benefited or any part of it, or any structure or works on the land benefited, in so far as that surface, subsurface, structure or those works are required to support the Spring Farm Parkway Roadway; and
 - (b) use any Drainage Infrastructure already located in, on or under the Easement Site (including any replacement Drainage Infrastructure) and, where no such Drainage Infrastructure exists, to construct, maintain and use Drainage Infrastructure on or under the Easement Site, for the purposes of draining any batter or embankment located on the land burdened to support the Spring Farm Parkway Roadway or capturing and diverting overland flows which otherwise would flow onto the RMS Works; [Drafting note: Delete whichever of the yellow highlighted words do not apply for any particular easement.]
 - (c) inspect, clean, repair, maintain, renew, replace or remove Drainage Infrastructure identified in clause 2.1(b);
 - (d) do anything reasonably necessary for purposes identified in clause **2.1(a)**, (b) or (c), including:
 - (i) entering the land burdened;

- (ii) taking anything on to the land burdened; and
- (iii) carrying out work.
- 2.2 If any batter or embankment or other support works are constructed or carried out on the land burdened, RMS or (if the works were carried out by the owner of the land benefited) the owner of the land benefited must:
 - (a) construct all such works in accordance with all RMS standards which are relevant to such works; and
 - (b) provide the owner of the land burdened with a statement from RMS' Project/Contract Manager for the RMS Works confirming that such works have been carried out in accordance with RMS' Specification R44 Earthworks (as amended from time to time, or any other standard which replaces or supersedes that standard)."

3. WORKS BY THE OWNER OF THE LAND BURDENED

- 3.1 After the completion of the RMS Works, the owner of the land burdened may carry out civil works, filling, earthworks and related activities on the land burdened:
 - (a) so that the ground levels on the land burdened align substantially with the finished levels of the Spring Farm Parkway Roadway; and
 - (b) to install drainage infrastructure.
- 3.2 The owner of the land burdened must not:
 - interfere with the batter or embankment, other than in the course of civil works, filling, earthworks and related activities on the land burdened contemplated by clause 3.1 which are carried out in accordance with an Approval;
 - (b) undermine the support which the batter or embankment offers;
 - (c) cause or allow to be caused any damage to any Drainage Infrastructure or interfere with the free flow of water through any Drainage Infrastructure, other than in the course of civil works, filling, earthworks and related activities on the land burdened contemplated by clause 3.1 which are carried out in accordance with an Approval; or
 - (d) use the Easement Site, or any other part of the land burdened, or any other land, in a way which may:

- (i) adversely affect the stability of or the support provided by the batter or embankment; or
- (ii) cause damage to the Drainage Infrastructure or interfere with the free flow of water through the Drainage Infrastructure, other than in the course of civil works, filling, earthworks and related activities on the land burdened contemplated by **clause 3.1** which are carried out in accordance with an Approval, and without RMS' prior consent,

except in accordance with the terms of this easement.

- 3.3 If the owner of the land burdened does or allows anything to be done which:
 - (a) damages the batter or embankment;
 - (b) impairs the effectiveness of the batter or embankment; or
 - (c) damages the Drainage Infrastructure or interferes with the free flow of water through the Drainage Infrastructure, other than in the course of civil works, filling, earthworks and related activities on the land burdened contemplated by clause 3.1 which are carried out in accordance with an Approval, and without RMS' prior consent,

other than in accordance with the terms of this easement, the owner of the land benefited may, or (if the owner of the benefited land is not RMS) must permit RMS to, serve not less than 14 days' notice on the owner of the land burdened requiring the damage to be repaired or the impairment removed. If the owner of the land burdened does not comply with the notice, the owner of the land benefited may, or (if the owner of the benefited land is not RMS) must permit RMS to, enter the land burdened and repair the damage or remove the impairment and may recover any reasonable costs from the owner of the land burdened.

- 3.4 In exercising the rights under clause 2.1 and clause 3.3, the owner of the land benefited must:
 - (a) ensure all work is done properly;
 - (b) cause as little inconvenience as is practicable to the owner and any occupier of the land burdened;
 - (c) cause as little damage as is practicable to the land burdened and any improvement on it;

- (d) restore the land burdened as nearly as is practicable to its former condition; and
- (e) make good any collateral damage; and
- (f) Remediate any Contamination which is caused, disturbed or exacerbated by it or its personnel.

4. TERMINATION OF THIS EASEMENT

- 4.1 This easement terminates on the date on which the owner of the land benefited gives written notice to the owner of the land burdened that it no longer requires the benefit of this easement.
- 4.2 Upon the termination of this easement:
 - (a) the owner of the land benefited must promptly:
 - (i) restore the land burdened as nearly as practicable to its condition prior to the grant of the easement, unless otherwise agreed with the owner of the land burdened; and
 - (ii) Remediate any Contamination which is caused, disturbed or exacerbated by it or its personnel; and
 - (b) the owner of the land burdened and the owner of the land benefited must promptly do all things necessary (including making application to NSW Land Registry Services) to remove the record of this easement from the title of the land burdened and the land benefited.

Annexure J

Catchment Plan

See attached.

3437-2025-8564v54 L\329906431.1 ME_152649482_1



page 72

Corrs Chambers Westgarth

3437-2025-8564v54 L\329906431.1 ME_152649462_1 Annexure B

Novation Deed

Minister for Planning and Public Spaces

Dahua Group Sydney Project 3 Pty Ltd

[New Developer]

Novation Deed for Voluntary Planning Agreement

Contents

1	Definitions		1
2	Novation		2
	2.1	Original Agreement	2
	2.2	Reference in Original Agreement	2
	2.3	Address for notices	2
3	Affirmation of the Original Agreement		2
4	Indemnities		2
5	5 Warranties and representations		3
	5.1	Warranties	3
	5.2	Survival of warranties	3
6	GST		3
	6.1	Construction	3
	6.2	Consideration GST exclusive	4
	6.3	Payment of GST	4
	6.4	Timing of GST payment	4
	6.5	Tax invoice	4
	6.6	Adjustment event	4
	6.7	Reimbursements	4
	6.8	Calculations based on other amounts	5
	6.9	No merger	5
7	Star	np duty and costs	5
8	Further acts		5
9	Amendment		5
10	Gov	erning law and jurisdiction	5
11	Counterparts		6
12	Gen	eral	6
	12.1	Construction	6
	12.2	Headings	6
Exe	cution		7
		····	

Date

Parties

Minister for Planning and Pubic Spaces ABN 20 770 707 468 of Level 15, 52 Martin Place, Sydney, New South Wales, 2000 (**Minister**)

and

Dahua Group Sydney Project 3 Pty Ltd ACN 606 391 922 of Suite 02, Level 20, 201 Elizabeth Street, Sydney NSW 2000 (Original Developer)

and

[New Developer] ACN [#] [Address] (New Developer)

Background

- A The Minister and the Original Developer are parties to the Original Agreement.
- B The Original Agreement relates to the whole of the Land.
- C The Original Developer proposes to transfer the whole of its interests the Land to the New Developer and wishes to novate all of its rights and obligations under the Original Agreement to the New Developer.
- D [Insert any other appropriate recitals]

Agreed terms

1 Definitions

In this deed these terms have the following meanings:

Effective Date	[X]
Land	Has the meaning given to that term in the Original Agreement.
Original Agreement	The voluntary planning agreement dated [X] and made between the Minister and the Original Developer.

2 Novation

2.1 Original Agreement

Subject to clause 2.2 and with effect from the Effective Date:

- (a) the New Developer is substituted for the Original Developer as a party to the Original Agreement;
- (b) the New Developer will be bound by the Original Agreement, and will be entitled to the benefit of the Original Agreement, as if the New Developer was a party to the Original Agreement instead of the Original Developer; and
- (c) the Original Developer is released and discharged from all obligations and liabilities, and from all claims (whether for costs, damages, fees, expenses or otherwise), arising under the Original Agreement, except in respect of any obligation, liability or claim that arose prior to the Effective Date.

2.2 Reference in Original Agreement

All references to the Original Developer in the Original Agreement are to be construed as references to the New Developer.

2.3 Address for notices

The Minister must address all notices and communications to be given or made by it to the New Developer under the Original Agreement to the following address:

New Developer:

Address:	[X]
Fax:	[X]
Contact Person:	[X]
Email:	[X]

3 Affirmation of the Original Agreement

The Original Agreement will be read and construed subject to this deed, and in all other respects the provisions of the Original Agreement are ratified and confirmed, and, subject to the variation and novation contained in this deed, the Original Agreement will continue in full force and effect.

4 Indemnities

The New Developer indemnifies the Original Developer on demand against all liabilities, claims, damages and loss which the Original Developer suffers or incurs in relation to the Original Agreement including those which arise or relate to acts or omissions occurring on or after the Effective Date.

5 Warranties and representations

5.1 Warranties

Each party represents and warrants that, at the time of execution, and at the Effective Date:

- (a) it has capacity unconditionally to execute, deliver and comply with its obligations under this deed;
- (b) it has taken all necessary action to authorise the unconditional execution and delivery of, and the compliance with, its obligations under this deed;
- (c) this deed is a valid and legally binding obligation and is enforceable against it by each other party in accordance with its terms; and
- (d) its unconditional execution and delivery of, and compliance with its obligations under, this deed do not contravene:
 - (i) any law or directive from a government entity;
 - (ii) its constituent documents;
 - (iii) any agreement or instrument to which it is a party; or
 - (iv) any obligation of it to any other person.

5.2 Survival of warranties

The warranties and representations in **clause 5.1** survive the execution of this deed and the novation of the Original Agreement.

6 GST

6.1 Construction

In this clause 6:

- (a) unless there is a contrary indication, words and expressions which are not defined in this deed but which have a defined meaning in the GST Law have the same meaning as in the GST Law;
- (b) GST Law has the same meaning given to that expression in the A New Tax System (Goods and Services Tax) Act 1999 (Cth) or, if that Act does not exist for any reason, means any Act imposing or relating to the imposition or administration of a goods and services tax in Australia and any regulation made under that Act; and
- (c) references to GST payable and input tax credit entitlements include:
 - notional GST payable by, and notional input tax credit entitlements of the Commonwealth, a State or a Territory (including a government, government body, authority, agency or instrumentality of the Commonwealth, a State or a Territory); and
 - (ii) GST payable by, and the input tax credit entitlements of, the representative member of a GST group of which the entity is a member.

6.2 Consideration GST exclusive

Unless otherwise expressly stated, all consideration, whether monetary or nonmonetary, payable or to be provided under or in connection with this deed is exclusive of GST (**GST-exclusive consideration**).

6.3 Payment of GST

If GST is payable on any supply made by:

- (a) a party; or
- (b) an entity that is taken under the GST Law to make the supply by reason of the capacity in which a party acts,

(**Supplier**) under or in connection with this deed, the recipient of the supply, or the party providing the consideration for the supply, must pay to the Supplier an amount equal to the GST payable on the supply.

6.4 Timing of GST payment

The amount referred to in **clause 6.3** must be paid in addition to and at the same time and in the same manner (without any set-off or deduction) that the GST-exclusive consideration for the supply is payable or to be provided.

6.5 Tax invoice

The Supplier must deliver a tax invoice or an adjustment note to the recipient of a taxable supply before the Supplier is entitled to payment of an amount under **clause 6.3**.

6.6 Adjustment event

If an adjustment event arises in respect of a supply made by a Supplier under or in connection with this deed, any amount that is payable under **clause 6.3** will be calculated or recalculated to reflect the adjustment event and a payment will be made by the recipient to the Supplier or by the Supplier to the recipient as the case requires.

6.7 Reimbursements

- (a) Where a party is required under or in connection with this deed to pay for, reimburse or contribute to any expense, loss, liability or outgoing suffered or incurred by another party or indemnify another party in relation to such an expense, loss, liability or outgoing (**Reimbursable Expense**), the amount required to be paid, reimbursed or contributed by the first party will be reduced by the amount of any input tax credits to which the other party is entitled in respect of the Reimbursable Expense.
- (b) This clause 6.7 does not limit the application of clause 6.3, if appropriate, to the Reimbursable Expense as reduced in accordance with clause 6.7(a).

6.8 Calculations based on other amounts

If an amount of consideration payable or to be provided under or in connection with this deed is to be calculated by reference to:

- (a) any expense, loss, liability or outgoing suffered or incurred by another person (**Cost**), that reference will be to the amount of that Cost excluding the amount of any input tax credit entitlement of that person relating to the Cost suffered or incurred; and
- (b) any price, value, sales, proceeds, revenue or similar amount (**Revenue**), that reference will be to that Revenue determined by deducting from it an amount equal to the GST payable on the supply for which it is consideration.

6.9 No merger

This **clause 6** does not merge on the completion, rescission or other termination of this deed or on the transfer of any property supplied under this deed.

7 Stamp duty and costs

The New Developer will pay all stamp duty arising directly or indirectly from this deed.

8 Further acts

- (a) Each party will take all steps, execute all deeds and do everything reasonably required by any other party to give effect to any of the actions contemplated by this deed.
- (b) This deed binds each party which signs it even if other parties do not, or if the execution by other parties is defective, void or voidable.

9 Amendment

This deed may only be varied or replaced by a deed executed by the parties.

10 Governing law and jurisdiction

- (a) This deed and the transactions contemplated by this deed are governed by and are to be construed in accordance with the laws applicable in New South Wales.
- (b) Each party irrevocably and unconditionally submits to the non-exclusive jurisdiction of the courts exercising jurisdiction in New South Wales and any courts which have jurisdiction to hear appeals from any of those courts and waives any right to object to any proceedings being brought in those courts.

11 Counterparts

This deed may consist of a number of counterparts and the counterparts taken together constitute one and the same instrument.

12 General

12.1 Construction

Unless expressed to the contrary, in this deed:

- (a) words in the singular include the plural and vice versa;
- (b) any gender includes the other genders;
- (c) if a word or phrase is defined its other grammatical forms have corresponding meanings;
- (d) 'includes' means includes without limitation;
- (e) no rule of construction will apply to a clause to the disadvantage of a party merely because that party put forward the clause or would otherwise benefit from it;
- (f) a reference to:
 - a person includes a partnership, joint venture, unincorporated association, corporation and a government or statutory body or authority;
 - (ii) a person includes the person's legal personal representatives, successors, assigns and persons substituted by novation;
 - (iii) any legislation includes subordinate legislation under it and includes that legislation and subordinate legislation as modified or replaced;
 - (iv) an obligation includes a warranty or representation and a reference to a failure to comply with an obligation includes a breach of warranty or representation.

12.2 Headings

Headings do not affect the interpretation of this deed.

Execution

Executed as a deed.	
Signed, sealed and delivered by the) Minister for Planning and Public) Spaces in the presence of:))
Signature of Witness	Signature of Minister
Name of Witness (print)	Name of Minister
Executed by Dahua Group Sydney Project 3 Pty Ltd ACN 606 391 922 in accordance with s127(1) of the <i>Corporations Act 2001</i> (Cth):)))
Company Secretary / Director	Director
Name of Company Secretary / Director (print)	Name of Director (print)
Executed by [New Developer] ACN [X] in accordance with s127(1) of the <i>Corporations Act 2001</i> (Cth):) 5))
Company Secretary / Director	Director
Name of Company Secretary / Director (print)	Name of Director (print)

Annexure C

Greater Macarthur Growth Area



Brett Hulicol



3442-3062-0428v9 Planning Agreement ME_151446986_1 L\331496286.1